ALMA MATER STUDIORUM - UNIVERSITÀ DI BOLOGNA

ARCES - ADVANCED RESEARCH CENTER ON ELECTRONIC SYSTEMS FOR INFORMATION AND COMMUNICATION TECHNOLOGIES "E. DE CASTRO"

DOTTORATO DI RICERCA IN TECNOLOGIE DELL'INFORMAZIONE CICLO XXIII - ING-INF/01

Physical Models for Numerical Simulation of Si-Based Nanoscale FETs and Sensors

Tesi di Dottorato

Presentata da Luca Silvestri

Coordinatore del Corso di Dottorato: Chiar.mo Prof. Ing. CLAUDIO FIEGNA

Relatore:

Dott.ssa Ing. SUSANNA REGGIANI

Esame Finale: anno 2011

Alma Mater Studiorum - Università di Bologna FACOLTÀ DI INGEGNERIA

Dottorato di ricerca in Tecnologie dell'Informazione

Ciclo XXIII (Anni accademici 2008-2010)

Settore scientifico disciplinare ING-INF/01

Physical Models for Numerical Simulation of Si-Based Nanoscale FETs and Sensors

Tesi di Dottorato di: LUCA SILVESTRI

Coordinatore: Chiar.mo Prof. Ing. CLAUDIO FIEGNA

Relatore: Dott.ssa Ing. SUSANNA REGGIANI

Esame Finale: anno 2011

To my family

Contents

List of symbols and abbreviations	V
Abstract	ix
Sommario	xi
Introduction	xiii

Ι	MO	BILITY MODELS FOR NANOSCALE MOSFETs	1
1	Mol	oility Model for Differently-Oriented UTB SOI n-FETs.	3
	1.1	The Miller index notation	4
	1.2	Bulk silicon band structure	5
	1.3	Quantum confinement in UTB FETs	6
	1.4	Carrier mobility	11
	1.5	Generalized mobility model	13
	1.6	Phonon scattering	15
	1.7	Coulomb and surface roughness scattering	17
	1.8	Band structures and repopulation effects	19
		1.8.1 Anisotropy and non-parabolicity in the (110) orientation	20
	1.9	Scattering induced by interface states	23
	1.10	Model enhancements for $t_{\rm Si} < 5 \mathrm{nm}$	26
		1.10.1 (100) substrates \ldots	26
		1.10.2 (110) substrates \ldots	30
	1.11	Mobility in double-gate FETs	31
	1.12	Review of the complete mobility model	34
	1.13	Summary	35
2	Mol	oility Model for Differently-Oriented UTB SOI p-FETs	37
	2.1	Generalized mobility model	38

	2.2	Scattering contributions	42
	2.3	Band structures and repopulation effects	44
	2.4	Model enhancements for $t_{Si} < 5 \mathrm{nm} \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	46
		2.4.1 (100) substrates $\ldots \ldots \ldots$	46
		2.4.2 (110) substrates \ldots	47
	2.5	Review of the complete mobility model	50
	2.6	Summary	51
3	Effe	cts of High- κ Gate Stacks and Metal Gates	53
	3.1	Remote scattering effects with thin oxides and metal gates	55
	3.2	Remote scattering effects in high- κ stacks $\ldots \ldots \ldots \ldots \ldots \ldots$	57
	3.3	Summary	58
4	Effe	cts of Mechanical Stress for Planar and Vertical FETs	59
	4.1	Strain effects in biaxial conditions	60
	4.2	Uniaxial stress effects for (100) and (110) crystallographic orientation	ns 65
		4.2.1 Electron mobility	65
		4.2.2 Hole mobility	71
	4.3	Mobility model for non-planar FETs	75
	4.4	Summary	80
5	QD	D simulations	81
	5.1	1D quantum drift-diffusion model	81
	5.2	Analysis of SOI FETs with different surface orientations	85
	5.3	Analysis of DG-FETs with different silicon thicknesses	88
	5.4	Analysis of DG-FETs under $\langle 110\rangle$ uniaxial stress $\hfill \ldots \hfill \hfill \ldots \hfill \ldots \hfill \ldots \hfill \ldots \hfill$	89
	5.5	Summary	90
II	SIL	ICON NANOWIRES FOR GAS DETECTION	91
6	TC	AD analysis of gas sensors based on SiNWs	93
	6.1	Definition of the simulation decks	98
	6.2	Analysis of the $\rm Si/SiO_2$ trap states in top-down $\rm SiNWs$	100
	6.3	Analysis of the molecular layer interface before exposure	105
	6.4	Cooperative effect of molecules on surfaces upon gas exposure	106
	6.5	Effect of surface-to-volume ratio on sensitivity	108
	6.6	Summary	110

Appendices	115
A Mobility model coefficients	117
Bibliography	123
Curriculum Vitae	139

List of symbols and abbreviations

Symbols

$D_{i,v}$	Carrier diffusivity
$D_{it}(E)$	Trap states energy distribution
$\epsilon_{\rm Si}$	Silicon dielectric constant
ê	Strain tensor
E	Energy
$E(\mathbf{k})$	Energy dispersion relation
$E_{\rm C}$	Conduction band-edge
$E_{\rm eff}$	Effective electric field
$E_{\mathrm{Fn},i,v}$	Subband quasi-Fermi energy
E_g	Band gap
$E_{i,v}$	Subband energy
E_{\parallel}	Longitudinal electric field
E_{\perp}	Normal electric field
$E_{\rm V}$	Valence band-edge
$F_{1,j}^v$	Form factor
g_v	Valley degeneracy
ħ	Reduced Plank's constant
Η	Device height
Ι	Current
J	Current density vector
$\mathbf{J}_{i,v}$	Subband current density vector
k	Lattice wave vector

$k_{\parallel},k_{\perp 1},k_{\perp 2}$	Lattice wave vector components in the ellipsoid coordinate system
k_1, k_2, k_3	Lattice wave vector components in the device coordinate system
k_1', k_2', k_3'	Lattice wave vector components in the crystal coordinate system
k_B	Boltzmann constant
L	Gate length
n	Electron density
$N_{\rm depl}$	Depletion charge concentration
$N_{i,v}$	Subband electron concentration
$N_{\rm inv}$	Inversion-layer carrier concentration
\hat{m}	Effective mass tensor
m_0	Free electron mass
m_{d_v}	Valley density-of-state mass
m_l	Longitudinal effective mass
m_t	Transverse effective mass
m_x	Transport effective mass
m_y	Width effective mass
m_z	Quantization effective mass
$\hat{\mu}_{ ext{eff}}$	Effective mobility tensor
$\hat{\mu}_v$	Valley effective mobility tensor
μ_{top}	FinFET top-side mobility
μ_{lateral}	FinFET lateral-side mobility
N_A	Acceptor doping concentration
N_D	Donor doping concentration
p_v	Valley population
ϕ	Electrostatic potential
ϕ_s	Surface electrostatic potential
ψ	Wave function
q	Electron elementary charge
r	Position vector
$\Re_{B \leftarrow A}$	Rotation matrix from the coordinate system A to the coordinate system B
ρ	Charge density
$\hat{\sigma}$	Mechanical stress tensor
t_{IL}	Interfacial layer thickness
$t_{\rm ox}$	Gate oxide thickness
$t_{\rm Si}$	Silicon film thickness
$ au_{ m AC}$	Acoustic phonon average momentum relaxation time

$ au_{\mathrm{CS}}$	Coulomb average momentum relaxation time
$ au_{\delta t_{ m Si}}$	Thickness fluctuations average momentum relaxation time
$ au_{it}$	Interface states average momentum relaxation time
$ au_{\mathrm{OP}}$	Optical phonon average momentum relaxation time
$ au_{\mathrm{PS}}$	Phonon average momentum relaxation time
$ au_{ m RCS}$	Remote Coulomb average momentum relaxation time
$ au_{ m RPS}$	Remote phonon average momentum relaxation time
$ au_{ m RSR}$	Remote surface-roughness average momentum relaxation time
$ au_{ m SP}$	Surface phonon average momentum relaxation time
$ au_{ m SR}$	Surface-roughness average momentum relaxation time
\hat{T}	Kinetic energy operator
T	Lattice temperature
U	Potential energy
V	Voltage
W	Device width
W_v	Valley carrier distribution effective width
$\zeta_{i,v}$	Subband electron eigenfunction

Abbreviations

1D	One-dimensional
2D	Two-dimensional
2DEG	Two-dimensional electron gas
2DHG	Two-dimensional hole gas
3D	Three-dimensional
BOX	Buried oxide
CCS	Crystal coordinate system
CMOS	Complementary metal-oxide-semiconductor
DCS	Device coordinate system
DG	Double-gate
DIBL	Drain induced barrier lowering
DPCP	Diphenylchlorophosphate
ECS	Ellipsoid coordinate system
EOT	Equivalent oxide thicknesses
FET	Field Effect Transistor
GAA	Gate-All-Around

HH	Heavy hole
$I_{\rm ON}$	On-current
$I_{\rm OFF}$	Off-current
LH	Light hole
MOSFET	Metal-oxide-semiconductor field-effect transistor
MRT	Momentum relaxation time
MuG	Multi-gate
NR	Nanoribbon
NW	Nanowire
RCS	Remote Coulomb scattering
RPS	Remote phonon scattering
RSS	Remote surface-roughness scattering
OP	Organo-phosphorus compounds
SCE	Short-channel effect
SG	Single-gate
SiNR	Silicon nanoribbon
SiNW	Silicon nanowire
SOI	Silicon-on-insulator
\mathbf{SS}	Inverse sub-threshold slope
TABINOL	3-(4-ethynylbenzyl)-1, 5, 7-Trimethyl-3-AzaBIcyclo [3.3.1] Nonane- 7-methanOL
TCAD	Technology Computer-Aided-Design
UTB	Ultra-thin body

Abstract

To continuously improve the performance of metal-oxide-semiconductor fieldeffect transistors (MOSFETs), innovative device architectures, gate stack engineering and mobility enhancement techniques are under investigation. Performance will also derive from heterogeneity, coming from the increasing diversity of functions integrated on complementary metal-oxide-semiconductor (CMOS) platforms. For example, new architectural concepts and technological approaches are being studied in the field of nanosensing.

In this framework, new physics-based models for Technology Computer-Aided-Design (TCAD) simulation tools are needed to accurately predict the performance of upcoming nanoscale devices and to provide guidelines for their optimization. In this thesis, advanced physically-based mobility models for ultrathin body (UTB) devices with either planar or vertical architectures such as single-gate silicon-oninsulator (SG-SOI) field-effect transistors (FETs), double-gate FETs, FinFETs and silicon nanowire FETs, integrating strain technology and high- κ gate stacks are presented. The effective mobility of the two-dimensional electron/hole gas in a UTB FETs channel is calculated taking into account its tensorial nature and the quantization effects due to both the application of a transverse electric field and the structural confinement. All the scattering events relevant for thin silicon films and for high- κ dielectrics and metal gates have been addressed and modeled for UTB FETs on differently oriented substrates. The effects of mechanical stress on (100) and (110) silicon band structures have been modeled for a generic stress configuration. The resulting analytical models, based on physical insights and theoretical analyses, have been thoroughly calibrated on available experimental data. Finally, they have been combined to provide a unified compact mobility model for electrons and holes, which has shown to give a correct interpretation of the mobility features originating both in planar and non-planar post-CMOS devices. In addition, the effective mobility models have been incorporated into a quantum drift-diffusion (QDD) simulation tool, addressing the 2D coupled Schrödinger-Poisson equations on the device cross-sections normal to the transport direction and the 1D drift diffusion equation for the current flow. The QDD simulator has been extended to accommodate different surface orientations by means of a generalized effective-mass approach. Extensive simulation studies have been carried out to investigate different architectural and geometrical solutions for the future technology nodes. Finally, starting from the effective mobility models, local mobility models have been developed, in order to allow the easy inclusion in commercial TCAD tools.

Beyond-CMOS nanostructures are of interest not only to extend the FET scaling process, but also to develop innovative sensor applications. Benefiting from properties like large surface-to-volume ratio and extreme sensitivity to surface modifications, silicon-nanowire-based sensors are gaining special attention in research. In this thesis, emphasys is given to the silicon nanowire technology for biological and chemical detection. A comprehensive analysis of the physical effects playing a role in the detection of gas molecules is carried out by TCAD simulations combined with interface characterization techniques. The complex interaction of charge transport in silicon nanowires of different dimensions with interface trap states and remote charges is addressed to correctly reproduce experimental results of recently fabricated gas nanosensors.

Sommario

Al fine di migliorare costantemente le prestazioni elettriche dei transistori ad effetto di campo (*field effect transitor*, FET) di tipo metallo-ossido-semiconduttore (MOSFET), si stanno attualmente indagando nuove architetture, strutture di gate innovative e tecniche per migliorare la mobilità dei portatori di carica. L'aumento delle prestazioni proviene inoltre dalla eterogeneità, derivante dalla crescente diversità di funzioni integrate nelle piattaforme di tipo metallo-ossido-semiconduttore complementare (CMOS). Ad esempio, si stanno studiando nuove scelte architetturali e approcci tecnologici nel campo del nano-rilevamento.

In tale contesto, si necessita di nuovi modelli fisici per simulazioni TCAD (Technology Computer-Aided-Design), al fine di predire con accuratezza le prestazioni dei futuri dispositivi su scala nanometrica e di fornire linee guida per la loro ottimizzazione. In questa tesi si presentano modelli di mobilità su base fisica avanzati per dispositivi a film ultra-sottile (ultrathin body, UTB) con architettura sia planare che verticale, come il transistore a singolo gate realizzato in tecnologia SOI (*silicon-on-insulator*), il transistore a doppio gate, il transistore FinFET e il transistore basato su nanofilo di silicio (silicon nanowire), che integrano stress meccanico e strutture di gate ad alta costante dielettrica. La mobilità efficace del gas di elettroni/lacune bidimensionale nel canale di un transistore a film sottile è calcolata tenendo conto della sua natura tensoriale e degli effetti di quantizzazione dovuti sia all'applicazione di un campo elettrico nella direzione trasversale al dispositivo sia dal confinamento strutturale. I meccanismi di collisione rilevanti nei film sottili di silicio e in presenza di dielettrici ad alta permittività e gate metallici sono modellati per dispositivi UTB con differenti orientazioni del substrato. Gli effetti dello stress meccanico sulle strutture a bande del silicio (100) e (110) sono modellati per una generica configurazione di stress. I risultanti modelli analitici, che si basano su una visione fisica e su analisi teoriche, sono stati accuratamente calibrati sui dati sperimentali disponibili. Infine, sono stati combinati in un modello di mobilità compatto per elettroni e lacune, il quale ha dimostrato di interpretare correttamente le caratteristiche della mobilità in dispositivi post-CMOS sia di tipo planare che non-planare. In aggiunta, i modelli di mobilità efficaci sono stati incorporati in un simulatore del trasporto ohmicodiffusivo quantistico (QDD), che risolve le equazioni di Schrödinger e Poisson 2D accoppiate in una sezione del dispositivo trasversale alla direzione del trasporto e l'equazione ohmico-diffusiva per il flusso di corrente elettrica. Il simulatore QDD è stato adattato a differenti orientazioni cristallografiche attraverso un approccio generalizzato per il calcolo delle masse efficaci. Si sono eseguiti studi approfonditi attraverso simulazioni per investigare diverse soluzioni architetturali e geometriche per i futuri nodi tecnologici. Infine, a partire dai modelli di mobilità efficace, si sono sviluppati modelli di mobilità locale, allo scopo di consentirne una facile integrazione in simulatori TCAD commerciali.

Le nanostrutture post-CMOS sono di interesse non solo per estendere il processo di *scaling* dei dispositivi, ma anche per sviluppare nuove applicazioni per sensori. Beneficiando di proprietà come un elevato rapporto superficie-volume e un'estrema sensibilità alle variazioni superficiali, i sensori basati su nanofili di silicio stanno acquisendo notevole interesse in ricerca. In questa tesi, si da' enfasi alla tecnologia a nanofili di silicio per il rilevamento biologico e chimico. Un'analisi dettagliata degli effetti fisici che giocano un ruolo nel rilevamento di molecole di gas è effettuata mediante simulazioni TCAD combinate con tecniche di caratterizzazione dell'interfaccia. Viene affrontata la complessa interazione del trasporto di carica nei nanofili di silicio di diverse dimensioni con gli stati trappola all'interfaccia per riprodurre correttamente i risultati sperimentali di nanosensori fabbricati di recente.

Introduction

In the last fifty years, integrated circuits (ICs) have driven the information technology revolution. Complementary metal-oxide-semiconductor (CMOS) is the technology that makes it possible to integrate digital and analog circuitry in ever-smaller silicon chips that lie at the heart of all the electronic products, from personal computer to mobile phones and from electronic stability systems in cars to miniature medical devices. The main actor of this improvement has been the metal-oxide-semiconductor field-effect transistor (MOSFET), due to its ultrahigh scalability. However, the request for increasing performance and reduced area occupancy pushes the scaling process every day closer to its physical limits. Innovative device structures, gate stack materials, strain engineering and unconventional surface orientations have been therefore proposed and investigated. The purpose is to overcome the limitations that nowadays nanoscale MOSFETs experience, i.e., short channel effects and gate leakage.

In Fig. 1 the Overall Roadmap Technology Characteristics (ORTC) of the International Technology Roadmap for Semiconductors (ITRS) 2010, is reported, showing present and future trends in nanoelectronics. Ultrathin body (UTB) devices with either planar or vertical architectures such as single-gate fully depleted silicon-on-insulator (FDSOI) field-effect transistors (FETs) and multi-gate (MuG) FETs are the most promising candidates for the future technology nodes. In fact, UTB FETs enable better electrostatic control of the channel, hence a more aggressive scalability, reduced leakage currents, and enriched functionality. Nevertheless, in order to scale UTB FETs down to the ultimate technology nodes, the silicon body needs to be thinned down to a few nanometers to suppress short channel effects, thus raising a concern on device variability due to statistical process fluctuations and charge transport limitations. Carrier mobility is indeed a sensitive function of the silicon-body thickness.

In the last few years, gate oxide thicknesses have been reduced to about 1 nm as demanded by device scaling, leading to an exponential increase of the gate leakage. In addition, undesired detrimental effects on carrier mobility due to remote scattering effects have been observed. Therefore, the study of high- κ di-



Figure 1: Overall Roadmap Technology Characteristics (ORTC) of the International Technology Roadmap for Semiconductors (ITRS) 2010. Graphical trends, including overlay of 2009 industry logic "nodes" and ITRS trends for comparison, are shown [1].)

electrics to replace SiO₂ is motivated on one side by the necessity of reducing the gate leakage, and, on the other side by the additional carrier mobility degradation that has regularly been reported for high- κ -based gate stacks due to remote scattering events. Metal-gate electrodes are also investigated to overcome the polysilicon depletion effect and for compatibility with high- κ dielectrics. Again, effects on the channel effective mobility have been observed.

Most of the recently proposed architectural solutions and materials for continuing device scaling have been demonstrated to negatively affect the carrier mobility in the channel. On the other hand, carrier mobility has been proven to play an important role on device performance even for ultrashort gate lengths. Therefore, several techniques to optimize carrier mobility, such as strain engineering and suitable surface orientations of the channel material are being studied. Furthermore, structures such as rectangular SiNW-FETs or FinFETs usually exhibit vertical transport on unconventional crystallographic planes. Thus, a deep comprehension of the physical details related with the different crystallographic orientations is required.

In this variegated framework, new physical models for Technology Computer-

Aided-Design (TCAD) simulation tools are needed, to provide guidelines for the optimization of upcoming nanoscale devices. In the process of development of novel technologies and designs, the use of TCAD is of fundamental importance for reducing time and investments costs. Many are the possible levels of abstraction going down from compact models for circuit simulations to more physics based models for the analysis of transport in single devices.

In the first and main part of this thesis, the aim has been developing physicallybased models for TCAD simulation tools, necessary for the design and study of advanced nanoelectronic devices. In particular, since all the innovative technology solutions described above positively or negatively impact the low-field carrier mobility, and since its importance on device performances, accurate mobility models are needed. Although the mobility concept in short-channel FETs becomes questionable because of the strong influence of quasi-ballistic transport, and the fundamental relationship between low-field mobility and drive current in such devices is still not well understood, the device transport properties are traditionally related to the low-field carrier mobility measured in long-channel devices. Thus, physically-based mobility models are essential for interpreting experimental mobility data.

When developing a simulation tool, importance has to be given to the trade-off between the accuracy in the physical description and the required computational burden. Modeling complex physical effects, which are inherently affected by a wide variety of physical mechanisms by means of full physical calculations would be very expensive in terms of computational time and resources. As an alternative physics-based analytical models translate complex numerical calculations into simple analytical expressions, where a number of parameters are allowed to vary to best fit the experiments and/or theoretical predictions.

Moreover, the preference is always on analytical physics-based models rather than empirical one. The hope is that a good physics-based model can be made scalable to different gate lengths and widths without requiring table-model based interpolation. Likewise, the emerging of mobility anisotropy effects in UTB FETs due to the breaking of the 3D symmetry enforces the requirement of good electron and hole mobility models for TCAD simulations. The consideration of every single physical effect, such as subband splitting, valley repopulation, scattering mechanisms and the like, is important to keep track of the underlying physics and also helps in the extraction of empirical parameters, when these are needed to express complex effects in simplified form, as well as unknown physical parameters. So doing, it is not needed to extract parameters by a global fitting, which most likely would generate unphysical results.

Therefore, the efforts have been focused on the realization of physically-based analytical mobility models for electrons and holes in planar and vertical UTB FETs. The physical mechanisms affecting electron and hole mobility in ultrathinsilicon films with different crystallographic orientations, and in presence of biaxial/uniaxial strain and high- κ dielectrics have been addressed, modeled and calibrated on available experimental data. The resulting unified model has been shown to correctly predict the low-field mobility in planar and vertical multi-gate architectures, even with complex stress configurations. A numerical simulation tool based on the self-consistent solution of the Schrödinger and Poisson equation in UTB FETs with different crystallographic orientations has been developed and used to calibrate the analytical models. In addition, the mobility models are implemented in a quantum drift-diffusion simulator addressing the 2D coupled Schrödinger-Poisson equations on the device cross-sections normal to the transport direction and the 1D drift diffusion equation for the current flow, in order to investigate the extent of the current improvements related with mobility enhancement. In commercial TCAD simulators the effective-mobility dependence on integral (non-local) carrier concentration and electric field may lead to numerical problems. Therefore, local mobility models, which depends on the local normal electric field and carrier concentration are preferred. Thus, in order to make the new mobility models suitable for commercial TCAD tools, local models are also provided.

In Chapter 1, the development of an analytical low-field electron mobility model for UTB FETs with different crystallographic orientations is presented in detail and its validation against experiments is illustrated. The Coulomb, surfaceroughness and phonon scattering models for unconventional surface orientations and ultra-thin body are treated. An accurate analytical description of the energy subbands which includes non-parabolicity effects is shown to give a correct interpretation of experimental mobilities on different crystallographic orientations and channel directions, as well as of the repopulation effects in ultra-thin-silicon films. Additional effects for ultra-thin SOI, like silicon-thickness fluctuations, suppression of intervalley-phonon scattering and surface phonons are also treated. In Chapter 2, the corresponding model for low-field hole mobility is presented. It provides accurate predictions of hole mobility in SOI FETs with (100) and (110) crystallographic orientations and silicon thickness down to about 2 nm. Chapter 3 is focused on the remote scattering effects related to ultra-thin gate dielectrics and gate stacks composed of high- κ oxide and metal gates. In Chapter 4, the effects of strain are included in the mobility models for electrons and holes. Valley splitting, valley repopulation, band deformation, and scattering contributions under biaxial and uniaxial strain conditions are studied and modeled. The effects of mechanical stress on (110) planes are also addressed, which are essential for the study of mobility in vertical architectures with (110)-oriented sidewalls. The developed models are suitably combined to predict the mobility behavior in n- and p-FinFETs with complex stress configurations, showing a nice agreement with experimental data. In Chapter 5, the attention is on the mobility impact on the on-current. A quantum drift-diffusion simulation tool, which includes the new mobility models, is used to carry out extensive simulation study of singleand double-gate UTB FETs.

The research on new architectural solutions to fulfill the request for continuing the scaling process of nanoelectronics is also meeting the need of integrating new functionalities in ICs. In particular, special attention is devoted to nanostructures which could be integrated within CMOS platforms, enabling functions that allow the world of digital computing to interact with the real world, providing conversion of non-digital as well as non-electronic information, such as chemical, mechanical, thermal, acoustic, optical and biomedical functions to digital data and vice versa. Nowadays nanodevices for the development of innovative applications with increased performance in the field of nanosensing, energy harvesting, nanocooling and RF are being thoroughly investigated. Nanotechnology can offer powerful ways to bring added value, in terms of costs, reproducibility, sensitivity, automation and new functionalities in healthcare applications such-as in-vitro diagnosis, as well as in environment control (water, air, soil), agriculture and food, defence or homeland security. A wide range of sensors is studied, such as chemical sensors, sensors for liquid and gas spectroscopy. Miniaturization enables price reduction, functionality multiplication and integration with electronics which will reduce parasitic effects, as well as ultra-high-sensitivity detection for chemical and biological applications.

The second part of this thesis (Chapter 6) is focused on the study of silicon nanowires for gas sensor applications. In these devices the surface charges exert a significant effect on the electrical conductivity. Therefore, attention is given to the modeling of interface traps and remote charges, and how they affect the device electrostatics. A TCAD simulation study, combined with experimental characterization, allowed for a correct interpretation of the experimentally observed current modulation upon exposure to gas. Furthermore, the impact of nanowire surface-to-volume ratio on sensor sensitivity is investigated, demonstrating a significant performance enhancement when the device width is shrunk below 50 nm.

Part I

MOBILITY MODELS FOR NANOSCALE MOSFETs

Chapter 1

Mobility Model for Differently-Oriented UTB SOI n-FETs.

The scaling of CMOS devices is reaching intrinsic limitations and needs new technological solutions. Ultra-thin body (UTB) devices with either planar or vertical architectures such as single-gate silicon-on-insulator (SG-SOI) FETs, double gate (DG) FETs, FinFETs and silicon nanowires (SiNWs) are the most promising candidates for fabricating sub-50-nm devices [1]. The study of the device performance requires the development of predictive physical models for carrier transport. To this purpose both mobility models [2] and enhanced drift-diffusion models which account for quasi-ballistic transport have been proposed [3]. To further improve the device performance for the future technology nodes, attention is being paid to the carrier mobility, which has been proved to play an important role on device performance even for ultra-short gate lengths [4], [5].

Several techniques to optimize carrier mobility in UTB FETs, such as strain [6] and suitable surface orientations [7], are still under investigation. Furthermore, structures such as rectangular SiNWs, FinFETs or Tri-gate FETs usually exhibit sidewall transport on the (110) crystallographic planes [8] [9] [10] [11]. Thus, a deep comprehension of the physical details related with the different crystallographic orientations is required.

In order to scale UTB FETs down to the ultimate technology nodes, the silicon body needs to be thinned below 5 nm to suppress short-channel effects, thus raising a concern on transport limitations. It has been experimentally demonstrated that electron mobility is a sensitive function of the silicon-body thickness, especially when $t_{\rm Si}$ is below 5 nm [12] [13] [14] [15] [16] [17]. 4

The carrier mobility of single-gate (SG) and double-gate (DG) UTB MOS-FETs has been extensively investigated, and a TCAD model has been proposed [2]. However, a TCAD model for UTB MOSFETs with unconventional surface and current-flow orientations is missing, and only experimental investigations can be found in the literature. The aim of this work is to derive a physically-based mobility model for device simulation tools which accurately predicts the low-field electron mobility in SG and DG FETs with different surface and channel orientations and silicon thicknesses as small as 2.5 nm. The mobility model presented in [2] has been modified to account for different crystallographic orientations.

In the following, a complete description of the model is provided and its validation against experiments is illustrated. Starting from a mobility formulation for bulk MOSFETs as a function of the effective transverse field, doping density, surface and channel orientations [18], a number of improvements have been added to reproduce the experiments taken on (100), $(110)/\langle 100 \rangle$ and $(110)/\langle 110 \rangle$ SG and DG FETs with ultra-thin silicon body.

In sections 1.1, 1.2, 1.3 and 1.4 some concepts essential for the comprehension of this thesis are introduced. In particular, in section 1.3 a Schrödinger-Poisson solver for different crystal orientations is presented. In section 1.5 the generalized mobility model is discussed. The effective inversion-layer thickness model for unconventional surface orientations and ultra-thin body is described in section 1.6; Coulomb and surface-roughness scattering models for unconventional surface orientations are treated in section 1.7; an accurate analytical description of the energy subbands which includes non-parabolicity effects is reported in section 1.8. The scattering induced by interface states is modeled in section 1.9 to improve the fitting of the (110) mobility data. Additional effects for ultra-thin SOI are shown in section 1.10, more specifically, silicon-thickness fluctuations, the suppression of inter-valley phonon scattering due to the shift of the energy minima, and surface phonons. Finally, the introduction of a simple analytical formulation allows us to include the effect of volume inversion on DG UTB MOSFETs (section 1.11). A review of the complete model is reported in section 1.12.

1.1 The Miller index notation

The Miller indices are commonly used to specify directions and planes in a crystal [19]. The Miller indices of a plane are defined as follows. First of all, three lattice vectors have to be defined. For cubic crystal systems, the lattice vectors are chosen along the edges of the crystallographic unit cell. Any crystal plane intercepts the axes in certain points. The Miller indices are determined by taking the reciprocals of these numbers and reducing them to the smallest three integers having the same ratio. The result is a triplet of integer values in parentheses $(hk\ell)$. A Miller index 0 means that the plane is parallel to the respective axis.



Figure 1.1: Miller indices of some important planes.

Negative indices are indicated with a bar written over the number or with the minus before the number.

In the conventional notation, $[hk\ell]$ with square brackets instead of round brackets, denotes a direction in the basis of the lattice vectors. The notation $\{hk\ell\}$ denotes all planes that are equivalent to $(hk\ell)$ by the symmetry of the crystal. Similarly, the notation $\langle hk\ell \rangle$ denotes all directions that are equivalent to $[hk\ell]$ by symmetry. In cubic crystal systems the Miller indices of a plane are the same as those of the direction perpendicular to the plane. Fig. 1.1 shows the Miller indices of important planes in a cubic crystal.

1.2 Bulk silicon band structure

The band structure describes the variation of the energy E with the momentum **k**. Silicon is an indirect band gap material with a band gap (E_g) of approximately 1.12 eV [20]. In the effective mass approximation, the band structure close to the conduction band edge can be approximated by ellipsoidal energy surfaces and a parabolic energy dispersion $E(\mathbf{k})$. In bulk silicon the conduction band minima are located at six equivalent points near the zone boundary X along the Δ symmetry lines. Choosing the momentum coordinate system aligned with the principal axes of a generic valley, the energy dispersion reads

$$E(\mathbf{k}) = \frac{\hbar^2 k_{\parallel}^2}{2m_l} + \frac{\hbar^2 (k_{\perp 1}^2 + k_{\perp 2}^2)}{2m_t} , \qquad (1.1)$$

where $\mathbf{k} = (k_{\parallel}, k_{\perp 1}, k_{\perp 2})$. $m_l = 0.916 \ m_0$ and $m_t = 0.19 \ m_0$ are the longitudinal and the transverse mass, respectively, m_0 is the free electron mass and \hbar the reduced Planck constant. The constant energy surfaces of all six valleys are ellipsoids of revolution grouped into three pairs placed along the principal axes $\langle 100 \rangle$, as shown in Fig. 1.2. In the following, the term "valley" is often adopted referring to a valley pair. 6



Figure 1.2: Constant energy surfaces for the conduction band minima of silicon. Equivalent valley pairs are labeled with the same number. The silicon crystal coordinate system (CCS) and the single-ellipsoid coordinate system (ECS) are indicated.

The valence band edges are located at the Γ point. The transport properties for the valence band are mainly affected by the properties of the heavy hole (HH) and light hole (LH) valleys. The constant-energy surfaces of HH and LH bands have complex warped shape [20].

1.3 Quantum confinement in UTB FETs

In Fig. 1.3-(A) a 3D sketch of the silicon-on-insulator (SOI) single-gate (SG) FET is shown. In the SG FET the silicon channel is sandwiched between the gate oxide (GOX) and a buried oxide (BOX), which provides vertical isolation limiting the detrimental effect of parasitic capacitances and leakage currents of the pn junctions. The short-channel effects (SCEs) consist in the loss of control of the gate on the potential profile along the channel, due to the aggressive reduction of the gate length L. The SOI technology allows the integration of multiple gates to ensure a better electrostatic control reducing SCEs. The 2D scheme of a planar DG FETs is illustrated in Fig. 1.3-(B), where two gate/oxide stacks are processed independently. Multi-gate (MuG) devices with a non-planar structure have been also proposed, such as FinFETs, where the conducting channel is formed in a thin silicon body covered by the gate stack on three sides, or gate-all-around silicon nanowire (SiNW) FETs. Although several experimental realizations have been reported during the last years for MuG-FETs, there are still many challenges associated with integration of these new structures [1]. In this chapter, the focus is on planar architectures while vertical structures will be treated later in this thesis.

Charge carriers confined in a narrow potential well does not behave classically. Increasing $V_{\rm GS}$ beyond the threshold voltage increases the inversion charge density



Figure 1.3: (A) 3D schematic of the SG-SOI device structure. The different device regions are depicted with different colors. The silicon layer between the buried oxide (BOX) and the back-gate contact is not shown. The device coordinate system (DCS) is also indicated. (B) 2D cross-section of a DG-SOI FET along the transport direction.

in the channel, which also increases the electric field perpendicular to the channel. The electric field in the inversion layer of a FET is strong enough to produce a potential well whose width in the direction perpendicular to the surface is small compared to the carrier wavelength [21]. Thus, quantization of the motion in discrete levels is expected along this direction. The energy levels of the carriers are called subbands, each of which corresponds to a quantized level of motion in the direction vertical to the interface. In addition, when considering ultra-thin body (UTB) FETs with silicon-film thickness below 10 nm, as requested by device scaling rules [1], the electron gas is structurally confined in the vertical direction, with a continuum for motion in the plane parallel to the surface. Therefore, in the thin silicon layer a quasi two-dimensional electron gas (2DEG) or hole gas (2DHG) is formed.

The calculations of the energy levels of carriers confined in a quantum well are commonly based on the following approximations [21]: i) the Hartree approximation, stating that each electron moves in the average potential produced by all other electrons, thus neglecting many-body effects; ii) in the semiconductor, the effective mass approximation is applied; iii) for the calculation of the energy levels it can also be assumed that the barrier between the insulator and the semiconductor is large enough that the envelope wave functions vanish at the semiconductor-insulator interface, which is reasonable for the Si/SiO₂ interface.

The physical model for the description of the device behavior comprises the Schrödinger and Poisson equations, namely

$$\nabla_{\mathbf{r}} \cdot (\epsilon_{\mathrm{Si}} \nabla_{\mathbf{r}} \phi) = \rho \tag{1.2}$$

CHAPTER 1. MOBILITY MODEL FOR DIFFERENTLY-ORIENTED UTB SOI N-FETS.

$$\left(\hat{T} + U(\mathbf{r})\right)\psi = E\psi$$
, (1.3)

which are needed to realistically compute the device electrostatics. In (1.2), $\phi(\mathbf{r})$ is the electric potential, $\epsilon_{\rm Si}$ is the silicon permittivity and ρ is the charge density, given by the sum of the contributions due to free carriers, doping concentration and other fixed charge centers, as for example charges trapped at the Si/SiO₂ interface. In (1.3), E is the total electron energy, \hat{T} is the operator for the electron kinetic energy, and $U(\mathbf{r})$ is the electron potential energy.

In the following, a reference frame with the z axis normal to the Si/SiO₂ interface and current transport along the x-direction is assumed, as shown in Fig. 1.3; the y-axis is along the device width. For the sake of simplicity, it is further assumed that the FET has a translational symmetry along the width, so that 1.2 and 1.3 can be solved in the (x, z) plane only. Thus, the position vector **r** and the (x, z) coordinates are used interchangeably. The general expression for $E(\mathbf{k})$ in the device coordinate system (DCS) (k_1, k_2, k_3) reads out

$$E(k_1, k_2, k_3) = \sum_{i,j=1}^{3} \frac{\hbar^2 k_i k_j}{2m_{ij}} , \qquad (1.4)$$

where m_{ij} are the components of the effective mass tensor.

If the potential is assumed to be a function of z only, i.e. U(x, z) = U(z), it is possible to the separate the trial solution of (1.3) into a z-dependent factor $\xi(z)$, and a plane wave factor representing free motion in the xy plane [22]

$$\psi(x, y, z) = \xi(z) \exp(jk_1 x + jk_2 y) .$$
(1.5)

The kinetic energy operator \hat{T} is obtained by replacing k_z with the quantummechanical operator $-j(\partial/\partial z)$. Therefore, by substituting (1.5) in (1.3), it is found that the functions ξ must satisfy the equation

$$\frac{\hbar^2}{2m_{33}}\frac{\mathrm{d}^2\xi}{\mathrm{d}z^2} + j\hbar^2 \left(\frac{k_1}{m_{13}} + \frac{k_2}{m_{23}}\right)\frac{\mathrm{d}\xi}{\mathrm{d}z} + (q\phi(z) + E')\xi(z) = 0 , \qquad (1.6)$$

where

8

$$E' = E - \frac{\hbar^2}{2} \left(\frac{k_1^2}{m_{11}} + \frac{2k_1k_2}{m_{12}} + \frac{k_2^2}{m_{22}} \right) .$$
 (1.7)

Following Stern and Howard [22], the first derivative in the above equation can be eliminated taking

$$\xi(z) = \zeta(z) \exp\left(-j \, z \, m_{33} \left(\frac{k_1}{m_{13}} + \frac{k_2}{m_{23}}\right)\right) \,. \tag{1.8}$$

The differential equation for $\zeta(z)$ takes the form

$$\left(-\frac{\hbar^2}{2m_z}\frac{d^2}{dz^2} + U(z)\right)\zeta_i(z) = E_i\zeta_i(z) , \quad \text{with} \quad m_z = m_{33} , \qquad (1.9)$$

where the eigenfunctions and the eigenvalues are labeled by a subscript i. The energy spectrum is given by

$$E(k_1, k_2) = E_i + \frac{\hbar^2}{2} \left(\left(\frac{1}{m_{11}} - \frac{m_{33}}{m_{13}^2} \right) k_1^2 + 2 \left(\frac{1}{m_{12}} - \frac{m_{33}}{m_{13}m_{23}} \right) k_1 k_2 + \left(\frac{1}{m_{22}} - \frac{m_{33}}{m_{23}^2} \right) k_2^2 \right) \quad (1.10)$$

and represents constant-energy ellipses above the minimum energy E_i . The energy levels E_i for a given value of m_z generate a set of subband minima called ladder. Since the value of the quantization mass depends on the substrate orientation, so do the number and the degeneracy of the subband ladders. Obviously, if conduction band valleys have the same orientation with respect to the surface, these valleys belong to the same ladder. It is worth noting that the valleys with the largest quantization mass m_z have the lowest energy. Following a widely used convention, the subbands belonging to the ladder lowest in energy are labeled $0, 1, 2, \ldots$, those of the second ladder $0', 1', 2', \ldots$, the third ladder $0'', 1'', 2'', \ldots$, and so on [21].

In order to determine the device electrostatics, the Schrödinger equation and the Poisson equation have to be solved self-consistently by numerical methods. The electrostatic potential determined by solving (1.2), enters in (1.3) through the determination of the potential energy $(U(\mathbf{r}) = E_g/2 - q\phi(\mathbf{r}))$, with q the elementary charge). On the other hand, the electron density depends on the solution of the Schrödinger equation through the square modulus of the electron wave function. Thus, a coupling between (1.2) and (1.3) is determined and an iterative self-consistent scheme has to be applied for the determination of the solution.

Eq. (1.2) is solved in the 2D domain (x, z). The boundary conditions for the Poisson equation are calculated by assuming charge neutrality and equilibrium at the contacts. Neumann boundary conditions for the potential are applied to the bottom edge of the buried oxide in SG-SOI FETs. In order to compute the quantum-mechanical charge density ρ , the 1D Schrödinger-like equation (1.9) is solved at each mesh point \overline{x} along the channel, where $0 < \overline{x} < L$, and for each valley. The eigenfunctions are zero at the Si/SiO₂ interfaces. Due to the constant density of states for a 2DEG, the electron concentration for each subband belonging to the v^{th} valley is proportional to the Fermi integral of order zero, which is expressed below

$$N_{i,v} = \frac{g_v m_{d_v} k_B T}{\pi \hbar^2} \ln(1 + \exp(-(E_{i,v} - E_{\mathrm{Fn},i,v}/k_B T))) |\zeta_{i,v}|^2 .$$
(1.11)

where $m_{d_v} = \sqrt{m_{x_v} m_{y_v}}$ is the density-of-states effective mass of the 2DEG in the v^{th} valley, with m_{x_v} and m_{y_v} the effective masses along the transport and the width direction, respectively. g_v is the valley degeneracy ($g_v = 2$ for the silicon

conduction band), k_B is the Boltzmann constant, T the lattice temperature. $E_{i,v}$, $E_{\text{Fn},i,v}$ and $\zeta_{i,v}$ are the energy, the quasi-Fermi level and the eigenfunction of the i^{th} subband belonging to the v^{th} valley, respectively.

The Schrödinger-Poisson solver in [23] has been generalized to SG- and DG-SOI MOSFETs with different crystal orientations. To this purpose, the effective masses along the quantization (m_z) , the transport (m_x) and the width (m_y) directions have been computed for each valley in an arbitrary-oriented device. Three orthogonal coordinate systems are considered: the device coordinate system (DCS) (k_1, k_2, k_3) , with k_1 , k_2 and k_3 along the source-to-drain (i.e., transport), the device width and the channel thickness (i.e., quantum confinement) direction, respectively (see Fig. 1.3); the crystal coordinate system (CCS) (k'_1, k'_2, k'_3) , with k'_1 along the [100] direction, k'_2 along [010] and k'_3 along [001] (see Fig. 1.2); the single-ellipsoid coordinate system (ECS) $(k_{\parallel}, k_{\perp 1}, k_{\perp 2})$, along the principal axes of each constant energy ellipsoid (see Fig. 1.2).

Eq. (1.1) can be written as

$$E = \frac{\hbar^2}{2} \vec{k}_E^T (M_E^{-1}) \vec{k}_E , \qquad (1.12)$$

where $\vec{k}_E = (k_{\parallel}k_{\perp 1}k_{\perp 2})^T$, and (M_E^{-1}) is the 3×3 diagonal matrix with m_l^{-1} , m_t^{-1} and m_t^{-1} along the diagonal.

Defining $\Re_{E \leftarrow C}$ the rotation matrix which transforms the components of a vector in the CCS $\vec{k}_C = (k'_1 k'_2 k'_3)^T$, to its components in the ECS,

$$\vec{k}_E = \Re_{E \leftarrow C} \vec{k}_C \,. \tag{1.13}$$

A second rotation matrix transforms the wave vector $\vec{k}_D = (k_1 k_2 k_3)^T$ in the DCS in the corresponding one in the CCS

$$\vec{k}_C = \Re_{C \leftarrow D} \vec{k}_D \,. \tag{1.14}$$

By combining (1.13) with (1.14) it is found

$$\vec{k}_E = \Re_{E \leftarrow D} \vec{k}_D \,, \tag{1.15}$$

where

10

$$\Re_{E \leftarrow D} = \Re_{E \leftarrow C} \Re_{C \leftarrow D} \,. \tag{1.16}$$

Substituting (1.15) in (1.12) it is obtained

$$E = \frac{\hbar^2}{2} \vec{k}_D^T (M_D^{-1}) \vec{k}_D , \qquad (1.17)$$

where the inverse mass tensor in the DCS (M_D^{-1}) is

$$(M_D^{-1}) = \Re_{E \leftarrow D}^T (M_E^{-1}) \Re_{E \leftarrow D} .$$
 (1.18)

The quantization mass $m_z = m_{33}$ is directly obtained from (M_D^{-1}) . Following the generalized effective mass approach by Rahman [11], each conduction band ellipsoid can be mapped into an equivalent regular ellipsoid whose principal axes are oriented along the device coordinate axes x, y and z with corresponding effective masses $m_x = m'_1$, $m_y = m''_2$ and m_z , respectively, where

$$\frac{1}{m_1'} = \left(\frac{1}{m_{11}} - \frac{m_{33}}{m_{31}^2}\right) \tag{1.19}$$

and

$$\frac{1}{m_2''} = \left(\frac{1}{m_2'} - \frac{m_1'}{m_{12}'^2}\right), \qquad (1.20)$$

$$\frac{1}{m_2'} = \left(\frac{1}{m_{22}} - \frac{m_{33}}{m_{23}^2}\right), \quad \frac{1}{m_{12}'} = \left(\frac{1}{m_{12}} - \frac{m_{33}}{m_{31}m_{23}}\right). \tag{1.21}$$

The resulting m_x and m_y enter in Eq. (1.11), m_z in Eq. (1.9).

The Schrödinger-Poisson solver for differently-oriented devices will be extensively utilized in the development of the carrier mobility models.

1.4 Carrier mobility

The carrier mobility characterizes the ability of charge carriers to move in a semiconductor or a metal in presence of an electric field, and is defined as the ratio between the carrier drift velocity and the electric field itself. In a three-dimensional silicon lattice the drift velocity is a linear function of the electric field for low values of the field itself, then saturates for fields approaching 10^4 V/cm. Thus, the low-field mobility is a constant equal to about $1400 \text{ cm}^2/\text{Vs}$ for electrons and $500 \text{ cm}^2/\text{Vs}$ for holes in undoped silicon at room temperature.

Inside the inversion layer of a MOSFET the effective low-field mobility μ_{eff} is much lower than in a three-dimensional lattice, and it has been found to be universal function of the effective vertical field, as experimentally demonstrated by Takagi [24]. E_{eff} is defined as

$$E_{\rm eff} = (q/\epsilon_{\rm Si})(\eta N_{\rm inv} + N_{\rm depl}) \quad , \tag{1.22}$$

with $N_{\rm inv}$ the carrier concentration in the inversion layer, $N_{\rm depl}$ the depletion charge per unit area and η is a constant equal to 1/2 for electrons and 1/3 for holes. For unconventional (110) and (111) substrate orientations it was found that a similar universal behavior is achieved when the value of η is properly adapted [24].

From a microscopic point of view it is possible to demonstrate that the mobility is proportional to the average time between two perturbations of the carrier motion. These perturbations, called scattering events, can be caused by several factors. In conventional bulk MOSFETs the most relevant are the interactions 12



Figure 1.4: Schematic of the universal mobility curves for bulk MOSFETs. μ_{CS} , μ_{PS} and μ_{SR} are the Coulomb-, phonon- and surface roughness-limited mobility, respectively.

with i) lattice vibrations (phonon scattering), ii) ionized atoms of the dopant or fixed charges (Coulomb scattering), and iii) the microscopic roughness of the Si/SiO₂ interface (surface roughness scattering). The impact of the different scattering mechanisms is described in Fig. 1.4, which shows the typical dependence of the channel mobility on the effective field.

However, with the advent of the UTB SOI technology and the introduction of high- κ gate oxides and metal gates the universal mobility behavior is no longer respected, due to the detrimental effect of additional scattering mechanisms. In order to compensate the mobility loss, strained channels and suitable surface orientations are being investigated. These aspects will be widely addressed in this thesis.

Approaching the ultimate technology nodes, the device channel length is pushed down in the deca-nanometric regime. The mobility concept in ultra-short channel FETs becomes questionable because of the strong influence of quasiballistic transport, and the fundamental relationship between low-field mobility and conduction current in such devices is still not clear. Apart from the importance of the these aspects, the analysis of the role played by the effective mobility in short-channel FETs is beyond the scope of this thesis. The mobility concept implies a uniform device subject to a vanishing electric field in the current-flow direction. In fact, an additional scattering mechanism, optical phonon scattering, which is unimportant at low-field conditions, becomes dominant at higher fields leading to the carrier velocity saturation [25]. The conditions of device uniformity and low-field in the current-flow direction are best met with long-channel FETs, and this is why experimental data are usually obtained from such devices. Thus, the low-field mobility models presented in this thesis turn out to be essential for the interpretation of the experimental results providing guidelines for the device optimization.


Figure 1.5: Top: orientation of the reference axes. Bottom: in-plane minima projections for (100), (110) and (111) surface orientations. The equivalent minima are labeled with the same number. The most relevant in-plane current flow directions are also indicated.

The effective low-field mobility μ_{eff} can be experimentally determined at low drain biases as

$$\mu_{\rm eff} = \frac{L^2 I_{\rm D}}{V_{\rm DS} \int_{-\infty}^{V_{\rm GS}} C_{\rm GC} \, dV_{\rm GS}} \,, \tag{1.23}$$

where L is the MOSFET length, $V_{\rm GS}$ and $V_{\rm DS}$ are the gate and drain voltages, respectively. $I_{\rm D}$ and $C_{\rm GC}$ are the measured drain current and gate-to-channel capacitance.

1.5 Generalized mobility model

Consider a silicon film on a substrate with one of the three crystallographic orientations indicated in Fig. 1.5, bottom. The z axis is set parallel to the structural confinement direction, while carrier transport occurs along the x axis. The inversion-layer quantization, due to the combined effect of structural confinement and application of a transverse electric field, causes the formation of energy ladders. More specifically, (100)-oriented FETs exhibit two energy ladders: the lower one, referred to as "unprimed", is two-fold degenerate and originates from the valleys labeled "3", while the upper one, referred to as "primed", is four-fold degenerate and is related with the valleys "1" and "2". In (110) substrates, the "unprimed" ladder is related to the valleys "1" and "2", while the primed one is related to the "3" valleys. Finally, a single six-fold degenerate ladder is formed along the (111) orientation.

In order to calculate the effective masses along the quantization (m_z) , the transport (m_x) and the width (m_y) directions for each valley in an arbitraryoriented device, the generalized effective mass approach by [11] described in section 1.3 has been implemented. Two different channel directions are analyzed for 14

(110)-oriented samples, namely the $\langle 100 \rangle$ and $\langle 110 \rangle$ (see Fig. 1.5). The effective masses corresponding to the analyzed cases are reported in table 1.1.

The quantization leads to unequal relative populations of the different valley pairs. Therefore, in-plane transport is generally described by a 2D tensorial effective mobility which retains the anisotropy of the single-valley effective mobilities:

$$\hat{\mu}_{\text{eff}} = \sum_{v=1}^{3} p_v \hat{\mu}_v \,. \tag{1.24}$$

where $\hat{\mu}_v$ and p_v are the mobility tensor and relative population of the v^{th} valley, respectively. A similar formulation is already available in 3D drift-diffusion transport simulation tools which handle bulk piezoresistivity and, in general, material anysotropy (see, e.g., [26] and [27]). Such tools can be directly used also in this case given that the out-of-plane (normal to the interface) mobility component plays no role and provided that the in-plane effective mobility model (1.24) is implemented. Unfortunately, this approach is unpopular in commercial TCAD tools because the effective-mobility dependence on integral (non-local) electron concentration and electric field may lead to numerical problems. As an alternative, a local mobility tensor $\hat{\mu}$ which depends on the local normal electric field $E_{\perp}(z)$ and carrier concentration n(z) can be defined, satisfying the following equation:

$$\hat{\mu}_{\text{eff}} = \frac{\int_{o}^{t_{\text{Si}}} (n(z) - n_0(z))\hat{\mu}(n, E_{\perp}) dz}{\int_{o}^{t_{\text{Si}}} (n(z) - n_0(z)) dz} \quad , \tag{1.25}$$

where $n(z) - n_0(z)$ is the excess electron concentration in the inversion layer. It should be noted that experiments measure only the xx component of the mobility tensor.

A two-step procedure is followed in the model development. In the first step, an analytical model for the effective mobility (1.24) is defined as a function of the effective electric field E_{eff} and the inversion-charge concentration per unit area, calibrating the parameters on experiments. E_{eff} is calculated as:

$$E_{\text{eff}} = \frac{\int_{o}^{t_{\text{Si}}} (n(z) - n_0(z)) E_{\perp}(z) dz}{\int_{o}^{t_{\text{Si}}} (n(z) - n_0(z)) dz} \quad .$$
(1.26)

In the second step, a local mobility model, which depends on $E_{\perp}(z)$ and n(z), is provided as described in section 1.12.

The relative populations are calculated by using the Boltzmann statistics as

$$p_v = \frac{m_{d_v} \exp\left(-E_{C_v}/k_B T\right)}{\sum_{v=1}^3 m_{d_v} \exp\left(-E_{C_v}/k_B T\right)},$$
(1.27)

where $m_{d_v} = \sqrt{m_{x_v} m_{y_v}}$ is the density-of-states effective mass of the 2DEG in the v^{th} valley, E_{C_v} is the valley bottom energy, k_B is the Boltzmann constant

(Wafer)/ <channel></channel>	m_x	m_y	m_z	valleys (see Fig. 1.5)	
$(100)/\langle 100 \rangle$	m_l	m_t	m_t	1	
	m_t	m_l	m_t	2	
	m_t	m_t	m_l	3	
$(110)/\langle 100\rangle$	m_t	$\frac{m_l+m_t}{2}$	$\frac{2m_lm_t}{m_l+m_t}$	1,2	
	m_l	m_t	m_t	3	
$(110)/\langle 110\rangle$	$\frac{m_l+m_t}{2}$	m_t	$\frac{2m_lm_t}{m_l+m_t}$	1,2	
	m_t	m_l	m_t	3	
$(111)/\langle 112\rangle$	$\frac{2m_t(2m_l+m_t)}{3(m_l+m_t)}$	$\frac{m_l+m_t}{2}$	$\frac{3m_lm_t}{2m_l+m_t}$	1,2	
	$\frac{2m_l+m_t}{3}$	m_t	$\frac{3m_lm_t}{2m_l+m_t}$	3	

Table 1.1: Principal effective masses for a 2DEG in (100), (110) and (111) oriented samples ($m_l = 0.916 \ m_0$ and $m_t = 0.19 \ m_0$)

and T the lattice temperature. As will be shown in section 1.8, the subband bottom energies are calculated analytically, accurately reproducing the solution of the Schrödinger-Poisson problem in the cross section normal to the transport direction. Following [28], the single-valley mobility tensor is modeled as

$$\hat{\mu}_v = \mu_v \hat{m}_v^{-1}, \quad \hat{m}_v^{-1} = \begin{pmatrix} m_0/m_{x_v} & 0\\ 0 & m_0/m_{y_v} \end{pmatrix},$$
 (1.28)

where \hat{m}_v^{-1} is the inverse scaled mass tensor of a 2DEG, defined for each valley v to account for the anisotropy effects induced by different in-plane crystal directions. This is expecially needed when considering the unprimed 4-fold valleys in (110) samples. Finally, μ_v is calculated accounting for the different scattering mechanisms combined via Matthiessen's rule, which assumes uncorrelated scattering mechanisms:

$$\mu_v = \frac{q}{m_0 \sum_j \tau_{v_j}^{-1}}.$$
(1.29)

In (1.29), τ_{v_j} represents the average momentum relaxation time (MRT) due to the *j*-th scattering mechanism for the v^{th} valley, as discussed in the next sections.

1.6 Phonon scattering

The average MRT $\tau_{\text{PS},v}$ is obtained by averaging the mean relaxation times $\langle \tau_{\text{PS},i,v} \rangle$ of the *i*th subbands in the *v*th ladder, each weighted with the population $p_{i,v}$ of the same subband

$$\tau_{\mathrm{PS},v} = \frac{\sum_{i} p_{i,v} \langle \tau_{\mathrm{PS},i,v} \rangle}{\sum_{i} p_{i,v}} \quad . \tag{1.30}$$

The mean $\langle \tau_{\text{PS},i,v} \rangle$ can in turn be calculated according to the study in [29] considering the effect of intravalley acoustic phonon scattering dominant at low fields.

16



Figure 1.6: Effective widths for the unprimed and primed subband ladders as a function of the effective field at $t_{Si} = 10 \text{ nm}$ (left) and $t_{Si} = 6 \text{ nm}$ (right). Symbols: numerical computations. Solid lines: Eq. (1.35).

Intervalley scattering is not considered in view of the low-field regime, which is close to equilibrium. First, the microscopic scattering rate in the i^{th} subband of the v^{th} ladder can be written as

$$\frac{1}{\tau_{\text{PS},i,v}(E)} = C_v m_{d_v} \sum_j F_{i,j}^v U(E - E_{j,v}) , \qquad (1.31)$$

where C_v is a constant related to the physical parameters of the acoustic-phonon scattering mechanism. The summation extends over all subbands of the same ladder, U being the unit-step function, $E_{j,v}$ the bottom energy of the j^{th} subband, and $F_{i,j}^v$ the form factor relative to the (i, j) subbands couple

$$F_{i,j}^{v} = \int_{0}^{t_{\rm Si}} |\zeta_{i,v}(z)|^{2} |\zeta_{j,v}(z)|^{2} \mathrm{d}z , \qquad (1.32)$$

where $\zeta_{i,v}(z)$ are the electron eigenfunctions. Then, by considering that most of the electron population of each subband occupies the subband's bottom, it is assumed $\langle \tau_{\text{PS},i,v} \rangle = \tau_{\text{PS},i,v}(E_{i,v})$. With this approximation, (1.30) becomes

$$\frac{1}{\tau_{\rm PS,v}} = \frac{C_v m_{d_v}}{W_v} \quad , \tag{1.33}$$

where the effective width W_v of the electron distribution relative to the v^{th} ladder has been introduced

$$W_{v} = \frac{\sum_{i} p_{i,v} \left(\sum_{j \le i} F_{i,j}^{v}\right)^{-1}}{\sum_{i} p_{i,v}}$$
(1.34)

and C_v are parameters reported in Appendix A.

The numerical computation of the effective widths for (100),(110) and (111) surface orientations has been carried out in equilibrium conditions ($V_{\text{D}S} \approx 0$) by

means of the Schrödinger-Poisson solver generalized to single-gate and doublegate SOI MOSFETs with different crystal orientations described in section 1.3. The calculated W_v are reported in Fig. 1.6 vs. the effective field E_{eff} for different surface orientations in SG-SOI FETs with $t_{\text{Si}} = 10$ and 6 nm, respectively. The average effective width is modeled as:

$$W_v = \frac{W_{\mathrm{T}v}}{\left[1 + (W_{\mathrm{T}v}/W_{\mathrm{E}v})^4\right]^{1/4}} , \qquad (1.35)$$

where

$$W_{\mathrm{T}v} = \frac{2}{3} t_{\mathrm{Si}} + W_{\mathrm{T}0v} \left(\frac{t_{\mathrm{Si}}}{t_{\mathrm{Si0}}}\right)^4 \left(\frac{E_{\mathrm{eff}}}{E_{\mathrm{eff0}}}\right)$$
(1.36)

and

$$W_{\rm Ev} = W_{\rm E0v} \left(\frac{E_{\rm eff}}{E_{\rm eff0}}\right)^{-\gamma} . \tag{1.37}$$

 $W_{\text{T}v}$ represents the effective width of the confined electron gas at low effective fields and small silicon thicknesses; $W_{\text{E}v}$ is the effective width at large normal fields and thick SOI films. In (1.36) and (1.37) $t_{\text{Si0}} = 10^{-7}$ cm, $E_{\text{eff0}} = 10^{6}$ V/cm; $W_{\text{T}0v}$ and $W_{\text{E}0v}$ are fitting parameters calibrated on numerical results (see Fig. 1.6, solid lines) and are reported in Appendix A. By comparing the average widths of the primed and unprimed subband ladders, a slighter dependence on the electric field is observed, due to the higher energy levels which are less sensitive to it. Eq. (1.37) is the generalization to the v^{th} valley of the theoretical formulation by Ando et al. [21], relative to the effective width for a single subband (single valley) case:

$$W = \frac{8}{3} \sqrt[3]{\frac{2}{3}} \sqrt[3]{\frac{\hbar^2}{qm_z}} E_{\text{eff}}^{-1/3} .$$
 (1.38)

 γ in (1.37), whose theoretically predicted value is 1/3, as shown in Eq. (1.38), is found to be equal to 0.29 for the (111) orientation and for the unprimed ladders of (100) and (110), while the values of 0.17 and 0.2 have been found for the primed ladders of (100) and (110), respectively (see Appendix A). As illustrated in Fig. 1.6, the difference between the unprimed and primed effective widths is more pronounced in (100) than in (110) samples. This effect can be due to the smaller difference between the quantization masses of the unprimed and primed valleys in the (110) case, leading to closer primed and unprimed energy levels.

1.7 Coulomb and surface roughness scattering

The Coulomb scattering term is modeled as

$$\frac{1}{\tau_{\rm CS,v}} = C_{\rm CS_0} \left(\frac{N_{\rm inv0}}{N_{\rm inv}}\right) \left(\frac{N_A}{N_{A0}}\right)^{\sigma} \quad , \tag{1.39}$$



Figure 1.7: Electron mobility in (100), (110) and (111) bulk MOSFETs vs. effective field for various substrate doping concentrations. Symbols: experiments from [24]. Solid lines: this model.

where N_A is the substrate doping density, $N_{A0} = 10^{17} \text{ cm}^{-3}$, $N_{\text{inv0}} = 10^{13} \text{ cm}^{-2}$, C_{CS_0} and σ are fitting parameters (see Appendix A), which are extracted by comparing the analytical model with experiments for bulk MOSFETs [24] with different N_A , as shown in Fig. 1.7. The inverse-linear dependence with the inversion-layer electron density N_{inv} is mainly due to the enhanced screening effect exerted by the electron charge [30]. N_{inv} for the bulk MOSFETs with uniform N_A measured in [24] can be calculated by inverting the usual expression

$$E_{\rm eff} = (q/\epsilon_{\rm Si})(\eta N_{\rm inv} + N_{\rm depl}) \quad , \tag{1.40}$$

with η is a constant equal to 1/2 for (100) and 1/3 for (110) and (111) substrates, as suggested in [24]. In (1.40), N_{depl} is calculated as

$$N_{\rm depl} = \sqrt{4\epsilon_{\rm Si}\Phi_B N_A/q} \quad , \tag{1.41}$$

with $\Phi_B = k_B T \log(N_A/N_i)$ the Fermi potential and $N_i = 8.765 \times 10^9 \text{ cm}^{-3}$ the intrinsic carrier concentration at 300 K. It is worth noting that, when implement-

ing the mobility model in a TCAD tool like, e.g., the 1D quantum drift-diffusion solver for SOI MOSFETs described in Chapter 5, E_{eff} and N_{inv} are directly obtained from the numerical results.

The theoretical formulation of surface roughness scattering under the assumption of single subband occupation reads

$$\frac{1}{\tau_{\mathrm{SR},v}} = C_{\mathrm{SR0}} m_{d_v} \left(\frac{E_{\mathrm{eff}}}{E_{\mathrm{eff0}}}\right)^{\delta} \tag{1.42}$$

where C_{SR0} is a constant and $\delta = 2$ [24]. Here the same expression is used, with C_{SR0} a fitting parameter (see Appendix A) and $\delta = 2.7$ for (100), 1.5 for (110) and 1 for (111) orientations, as found from experiments in Fig. 1.7. The different impact on mobility of Coulomb and surface-roughness scattering for different orientations was predicted by Monte Carlo simulations [31], and was ascribed to the different energy quantization and interface properties. This result is not surprising since the density of surface atoms and available bonds strongly depend on the crystal orientation [32], [20].

1.8 Band structures and repopulation effects

In order to compute the relative populations of the unprimed and primed ladders in (100) and (110) oriented samples and their dependence on the silicon-film thickness and the effective field, an analytical formulation based on physical considerations is developed. In (111) samples a single six-fold degenerate ladder is present; hence no repopulation occurs. For zero normal electric field (quantum well) the analytical solution of the Schrödinger equation provides the expression for the energy levels. The relative distance between the primed and unprimed subband edges reads:

$$\Delta E_{\rm CT} = E'_{\rm CT} - E_{\rm CT} = \frac{(\hbar\pi)^2}{2t_{\rm Si}^2} \left(\frac{1}{m_z^{pr}} - \frac{1}{m_z^{unpr}}\right) \quad . \tag{1.43}$$

From (1.43), the separation between the energy minima of the two ladders increases with the reduction of $t_{\rm Si}$, and electrons mostly populate the unprimed ladder. On the other hand, at large electric fields the energy minima can be theoretically calculated assuming a triangular potential well [22] as:

$$E_{CE_{v}} = \left(\sqrt[3]{\frac{9}{32}} + \sqrt[3]{\frac{9}{4}}\right) \sqrt[3]{\frac{\hbar^{2}q^{2}}{m_{z_{v}}}} E_{eff}^{2/3}$$

= $E_{CE0_{v}} \sqrt[3]{\frac{m_{0}}{m_{z_{v}}}} \left(\frac{E_{eff}}{E_{eff0}}\right)^{\chi_{v}}$. (1.44)

The last expression is adopted and E_{CE0_v} and χ_v are used as fitting parameters (see Appendix A) to accurately reproduce the energy minima of bulk MOSFETs



Figure 1.8: Difference between the conduction subband edges of the 2-fold and 4-fold valleys in (100) samples (left) and (110) samples (right) vs. the effective normal field for SG-SOI FETs with different silicon thicknesses. Symbols: numerical values; solid lines: Eq. (1.45).

on (100) and (110) substrates, as computed by the Schrödinger-Poisson solver. Indicating with $\Delta E_{\rm CE}$ the difference between the ladder minima at high electric fields, the behavior of $\Delta E_{\rm C}$ can be expressed as

$$\Delta E_{\rm C} = \Delta E_{\rm CT} \left[1 + (\Delta E_{\rm CE} / \Delta E_{\rm CT})^{\beta} \right]^{1/\beta} , \qquad (1.45)$$

where $\beta = 3.5$. Fig. 1.8 compare Eq. (1.45) with numerical simulation results provided by the Schrödinger-Poisson solver for different silicon thicknesses, effective fields and substrate orientations. The above model for the energy difference between the subband edges of the primed and unprimed ladders allows us to compute the relative valley populations from Eq. (1.27), which are reported in Fig. 1.9 vs. $t_{\rm Si}$ for (100)-oriented samples. According to previous works [33], [17], a clear repopulation effect occurs at about $t_{\rm Si} = 7$ nm. The unprimed ladder, which exhibits the lower transport effective mass (see table 1.1), turns out to be fully populated for $t_{\rm Si} \leq 4$ nm, with a beneficial effect on mobility.

1.8.1 Anisotropy and non-parabolicity in the (110) orientation

Further band-structure analyses are required for (110) substrates. As already pointed out, the effective mobility in (110) samples exhibits a strong in-plane anisotropy. Consider for example a $\langle 100 \rangle$ -oriented FET channel on a (110) substrate. As shown in Fig. 1.5, bottom, the unprimed ladder is the most populated one and exhibits a higher mobility due to the low transport effective mass (see table 1.1). On the contrary, if the FET channel is $\langle 110 \rangle$ oriented, the transport effective mass of the four-fold ladder is higher. This clarifies the experimental mobility reduction with respect to that of the $\langle 100 \rangle$ oriented FET channel [34].



Figure 1.9: Relative occupancy of the unprimed and primed subband ladders in (100) oriented SG-SOI samples vs. the silicon thickness for two different effective fields from (1.27). Symbols: numerically calculated relative populations; lines: this model. Inset: energy shift between primed and unprimed ladders from (1.45).

An additional effect has been outlined in the experimental analysis carried out by Uchida in Ref. [34]. The strong nonparabolicity of the conduction band in the [110] crystalline direction, i.e. the quantization direction for the (110) orientation, leads to a smaller subband-edge difference than predicted by Eq. (1.45) between the unprimed and primed ladders for energies exceeding 100 meV. In order to account for this effect, Eqs. (1.43) and (1.45) have been modified. The nonparabolicity effects are assumed to play a relevant role in the 2-fold primed ladder, mainly because of the high energy minimum. At zero field, the lowest energy of the primed ladder in the parabolic-band approximation is

$$E'_{\rm CT} = \frac{(\hbar\pi)^2}{2t_{\rm Si}^2 m_{\rm Z}^{pr}} \quad . \tag{1.46}$$

According to [35], a non-parabolic energy dispersion relationship for the conduction band minimum can be expressed as

$$E_{\rm CT}^{\prime \rm NP}(1 + \alpha E_{\rm CT}^{\prime \rm NP}) = \frac{(\hbar k_z)^2}{2m_z^{pr}} \quad , \tag{1.47}$$

where α is the non-parabolicity factor and k_z is the wave vector component in the quantization direction. When $k_z = \pi/t_{\rm Si}$ and using (1.46), Eq. (1.47) yealds

$$E_{\rm CT}^{\prime \rm NP} = \frac{-1 + \sqrt{1 + 4\alpha E_{\rm CT}^{\prime}}}{2\alpha} \quad . \tag{1.48}$$

The α value has been fitted on Uchida's nonparabolic energy dispersion relationship [34] shown in the inset of Fig. 1.10, and has been fixed at 3.4 eV⁻¹.



Figure 1.10: Difference of conduction-subband edges between the 2-fold and 4-fold valleys of (110) oriented substrates vs. the inversion charge concentration for SG-SOI FETs with different silicon thicknesses after nonparabolic corrections (1.48) and (1.49). The parabolic and nonparabolic bands of a bulk MOSFET calculated in [34] are also shown. Inset: E(k) relationship of energy ellipsoids "3" in Fig. 1.5 along [110] crystal orientation calculated in [34] (symbols) and from (1.48) (solid line).

A second correction is made in order to account for nonparabolicity effects at high electric fields. From Eq. (1.44) the energy minima have an increasing trend with E_{eff} due to the parabolic-band approximation. In order to avoid an overestimation of ΔE_{C} , Eq. (1.45) is modified as follows

$$\Delta E_{\rm C}^{\rm NP} = \Delta E_{\rm C} \left[1 - \log \left(1 + \frac{N_{\rm inv}}{9 \times 10^{12}} \right) \right] \quad , \tag{1.49}$$

and reproduce the difference of the subband edges of the primed and unprimed ladders calculated in [34], as shown in Fig. 1.10. At low $N_{\rm inv} \Delta E_{\rm C}^{\rm NP}$ increases by decreasing $t_{\rm Si}$ up to about 4 nm, decreases for $t_{\rm Si}$ between 4 and 2.5 nm where it becomes zero and unprimed and primed ladders cross each other (see Fig. 1.10). The effects of anisotropy and nonparabolicity on experimental mobilities for $(110)/\langle 100 \rangle$ and $(110)/\langle 110 \rangle$ devices are shown in Fig. 1.11. The enhancement of the 4-fold ladder population, which is favourable for the $\langle 100 \rangle$ and unfavourable for the $\langle 110 \rangle$ channel directions, has a strong impact on mobility if a parabolic band model is used (dashed lines). The nonparabolic corrections to $\Delta E_{\rm C}$ (solid lines) make the subband repopulation less effective.

As illustrated in Fig. 1.11, the application of the nonparabolic corrections to $\Delta E_{\rm C}$ is required to reproduce the experimental results. It is worth noting that the effective width in (1.34) is still based on a parabolic-band approximation, but in the low-medium $E_{\rm eff}$ range and for $t_{\rm Si} > 5$ nm, the non-parabolicity effects on carrier mobility are negligible.



Figure 1.11: Electron mobility in bulk and SG-SOI MOSFETs on (110) substrates vs. the effective field for $\langle 100 \rangle$ and $\langle 110 \rangle$ channel directions. Symbols: experiments from [34] and [16]. Solid lines: this model with non-parabolic corrections. Dashed lines: this model with parabolic bands. Arrows highlight the different impact on mobility of the two non-parabolic corrections in (1.48) and (1.49) for small t_{Si} and high electric field, respectively.

The physical effects considered so far allow us to reproduce the experimental mobilities in [13] and [16] with $t_{\rm Si}$ as small as about 5 nm, as illustrated in Fig. 1.12. However, to further improve the fitting of the (110) experiments, the scattering induced by interface states in SG-SOI FETs has to be included, as explained in the next section. Unfortunately, no experimental investigations are available in the literature to validate the mobility model in the (111) SOI case.

1.9 Scattering induced by interface states

As experimentally observed in Fig. 1.12, a mobility degradation appears at very low effective fields for (110)-oriented FETs with $t_{\rm Si} = 9$ and 5.2 nm, even if the devices are practically undoped. In [17], the interface-state limited mobility μ_{it} was calculated for (100) samples in the frame of a momentum-relaxation time (MRT) approximation accounting for intersubband transitions and for the silicon-film thickness (Fig. 6 in [17]). A negligible degradation of μ_{it} has been demonstrated when the silicon film is reduced from 30 down to about 10 nm, but becomes important below 10 nm. As detailed below, when $t_{\rm Si}$ is shrunk below 10 nm in SOI FETs it is appropriate to model an additional scattering contribution which accounts for the mobility reduction induced by interface states

$$\frac{1}{\tau_{it,v}} = C_{\rm CS_0} \left(\frac{N_{it}}{N_{it0}}\right) \left(\frac{N_{\rm inv0}}{N_{\rm inv}}\right)^{0.7} \left(\omega \frac{t_{\rm Si0}}{t_{\rm Si}}\right)^{\varphi} \quad , \tag{1.50}$$



Figure 1.12: Electron mobility in (100), (110)/(100) and (110)/(110) SG-SOI FETs vs. effective field for various silicon thicknesses. Symbols: experiments in [16] and [13]. μ vs. $N_{\rm inv}$ curves in [16] have been converted in μ vs. $E_{\rm eff}$ curves assuming $N_A = 5 \times 10^{15}$ cm⁻³. Dashed lines: mobility model without the scattering induced by interface states. Solid lines: mobility model with the scattering induced by interface states as given by Eq. (1.50).

where $N_{it0} = 5 \times 10^{10} \text{ cm}^{-2}$, φ and ω have been extracted from the $\mu_{it}(t_{\rm Si})$ curve in Fig. 6 of Ref. [17] and were found to be 1.27 (see inset in Fig. 1.14) and 112, respectively. Also, the $N_{\rm inv}^{-0.7}$ dependence in (1.50) has been extracted by the fitting in the inset of Fig. 1.14 [13], [36].

According to a recent experimental analysis on biaxially strained n-MOSFETs [37] and on highly-doped MOSFETs, [38], the $N_{\rm inv}$ dependence of μ_{it} increases when the mean distance between electrons and the Si/SiO₂ interface $\langle Z_{\rm inv} \rangle$ decreases, even in the weak inversion region. This is exactly what happens when the silicon-body thickness is strongly reduced. The result extracted from [17] of $N_{\rm inv}^{-0.7}$ is confirmed by the experimental investigation in [13] at low $t_{\rm Si}$, thus validating our approach.

In addition, it is shown in [37] that μ_{it} is degraded in strained channels since the unprimed ladder, whose electron wave functions are nearer to the interface than those in the primed ladder in view of their larger quantization mass, becomes more populated (see Fig. 1.13). When a (110) substrate is considered, the quantization masses of the unprimed and primed ladders are very close and the first eigenfunctions of the two ladders are approximately at the same distance from the interface, as illustrated in Fig. 1.13. Thus, the impact of repopulation on μ_{it} in (110) structures may be neglected and the interface state scattering can



Figure 1.13: First unprimed and primed eigenfunctions for (100) and (110) SG-SOI structures as a function of position along the z axis at $E_{\text{eff}} \simeq 8 \times 10^4$ V/cm calculated by our Schrödinger-Poisson solver.

be modulated only by $\langle Z_{\rm inv} \rangle$. However, the $\mu_{it}(t_{\rm Si})$ curve in Fig. 6 of [17] continues to decrease exactly as $t_{\rm Si}^{1.27}$ also when the unprimed valleys are completely populated (between 2 and 3 nm), i.e. when $\langle Z_{\rm inv} \rangle$ is simply reduced by structural confinement. Thus, the same $t_{\rm Si}^{1.27}$ dependence of μ_{it} can be used for (110) structures. The value of φ for (110) devices has been verified by a comparison with experimental mobilities from [15] at low temperature, as illustrated in Fig. 1.14. Considering that the phonon-limited mobility depends on temperature as $T^{-1.75}$ as suggested by [24], the scattering rate in Eq. (1.33) has been multiplied by 0.15 for all valleys to reproduce the experiments at 100 K. In (1.50) only ω has been adjusted with respect to the 100 K fitting ($\omega(100 K) = 480$).

In (1.50), N_{it} represents the density of states per unit area at the interface. Recently, a number of publications reported the amount of interface states in devices with different orientations (e.g. [7], [39]) and indicated that (110)oriented devices have an interface state density about three times larger than (100)-oriented devices. As already mentioned in section 1.7, a higher interface state density is indeed expected due to a larger atomic surface density, which increases the probability of dangling bonds or coordination defects at the interfaces. Therefore, considering that in a (100) oriented device the interface-state density is typically of the order of $5 \times 10^{10} \text{ cm}^{-2}$, $N_{it} = 1.5 \times 10^{11} \text{ cm}^{-2}$ has been used for (110) structures. Thus, this scattering contribution is found to be effective only in the (110) case as shown in Fig. 1.12 and does not remarkably influence conventionally-oriented mobilities. This is confirmed by the investigation in [17], showing that an unrealistically large amount of interface states should be needed to observe a mobility degradation at low effective field in conventionally-oriented



Figure 1.14: Electron mobility vs. effective field at 100 K for (110) SG-SOI FETs. Measurements from [15] (symbols), our model with 15% of phonon scattering (solid line). Inset: μ_{it} vs. t_{Si} from simulations in [17] (symbols) and from our model in (1.50).

devices.

26

For $t_{\rm Si} < 5 \,\mathrm{nm}$, further physical effects need to be addressed and modeled, as explained in section 1.10. Due to the lack of experimental data on (111) substrates, the mobility analysis $t_{\rm Si} < 5 \,\mathrm{nm}$ is limited to the (100) and (110) orientations.

1.10 Model enhancements for $t_{\rm Si} < 5 \, \rm nm$

1.10.1 (100) substrates

1) Scattering induced by silicon-thickness fluctuations

The measured (100) mobility for $t_{\rm Si} < 5$ nm exhibits a strong degradation which was ascribed to SOI-thickness fluctuation scattering [2], [17], [40]. In [2], it was assumed that this effect influences mobility only at low effective gate fields. Here the previous model [2] is improved by recognizing that the structural confinement induced by $t_{\rm Si}$ is important at high electric fields as well. Thus, the scattering contribution

$$\frac{1}{\tau_{\delta t_{\rm Si}h,v}} = \frac{1}{\tau_{\delta t_{\rm Si}h} \left(\frac{t_{\rm Si}}{t_{\rm Si0}}\right)^{\eta_1} \left(\frac{E_{\rm eff0}}{E_{\rm eff}}\right)} \quad , \tag{1.51}$$

is added to the formulation in [2], which reads

$$\frac{1}{\tau_{\delta t_{\rm Si}l,v}} = \frac{1}{\tau_{\delta t_{\rm Si}l} \left(\frac{t_{\rm Si}}{t_{\rm Si0}}\right)^{\eta_1} \left(C_{\delta t_{\rm Si}} \left(\frac{E_{\rm eff}}{E_{\rm eff0}}\right)^{\eta_2} + 1\right)} \quad , \tag{1.52}$$



Figure 1.15: Electron mobility vs. effective field at 25 K for SG-SOI FETs. Symbols: measurements from [12]. Solid lines: surface-roughness + siliconthickness-fluctuation limited mobility.

to reproduce the measurements at 25 K [12] shown in Fig. 1.15. In (1.51) and (1.52) $\tau_{\delta t_{\rm Si}l}$, $\tau_{\delta t_{\rm Si}h}$ and $C_{\delta t_{\rm Si}}$ are fitting parameters; η_1 is found to be equal to 6 as theoretically predicted [40], and η_2 is equal to 1. At 300 K only $\tau_{\delta t_{\rm Si}l}$ and $\tau_{\delta t_{\rm Si}h}$ are adjusted with respect to the 25 K fitting (see Appendix A).

2) Suppression of inter-valley phonon scattering

As highlighted in [33], another important physical effect must be considered when the silicon-film thickness is smaller than 5 nm, i.e. the suppression of inter-valley phonon scattering. In the absorption process of inter-valley f-type phonons, electrons in the unprimed ladder with an energy E reach states in the primed ladder with energy $E + E_{fp}$, where E_{fp} is the energy of the inter-valley f-type phonon. Thus, if the energy difference between the minima of the primed and unprimed ladders is higher than E_{fp} , the scattering process is suppressed. The simplest way to introduce the suppression of inter-valley phonon scattering is to model a reduction of the inverse momentum relaxation time in the expression (1.33) relative to the unprimed valleys:

$$\frac{1}{\tau_{\rm PS}^{\prime unpr}} = \frac{1}{\tau_{\rm PS}^{unpr}} \left(1 - f(\Delta E_{\rm C}) \right) \quad . \tag{1.53}$$

Exploiting the subband structure model presented in the previous section, $f(\Delta E_{\rm C})$ in (1.53) is modeled with a combination of analytical functions:

$$f(\Delta E_{\rm C}) = \frac{f_l}{\left[1 + (f_l/f_h)^{\alpha_f}\right]^{1/\alpha_f}} , \qquad (1.54)$$



Figure 1.16: Electron mobility vs. silicon thickness at 300 K for (100) SG-SOI devices. Symbols: measurements from [12]. Lines: phonon + surface-roughness limited mobility (dashed), phonon + surface-roughness + thickness fluctuation limited mobility (dashed-dotted), phonon + surface-roughness + thickness-fluctuation + surface optical-phonon limited mobility (long-dashed), total mobility (solid). Inset: function f as extracted from experiments compared with the analytical expression (1.54) vs. the difference of the energy band edges of the primed and unprimed valleys.

with

$$f_{l,h} = \frac{f_0}{1 + \exp\left[-\left(\Delta E_{\rm C} - \Delta E_{\rm C0_{l,h}}\right)/C_{l,h}\right]} , \qquad (1.55)$$

where f_0 , ΔE_{C0_l} , ΔE_{C0_h} , C_l and C_h are fitting parameters (see Appendix A) calibrated on experiments from [12] at $E_{\rm eff}=0.3$ MV/cm, as depicted in Fig. 1.16, and $\alpha_f = 15$. Thus, the mobility enhancement observed in the experimental data from [12] cannot be entirely attributed to repopulation effects, as supposed in our previous work [2]. $f(\Delta E_{\rm C})$ is reported in the inset of Fig. 1.16. It may be noticed that it becomes different from zero for $\Delta E_{\rm C} > 60 \,\mathrm{meV}$, which coincides with the value of E_{fp} reported in [33], and starts saturating when $\Delta E_{\rm C}$ is between 250 and 300 meV. This energy range is the same over which the experimentally-observed mobility enhancement in biaxially-strained (100) FETs tends to saturate. Also in this case, the same physical effects come into play, namely: the enhancement of the unprimed-valley population and the suppression of inter-valley phonon scattering. In several works ([41], [42] and [43]) the saturation of mobility enhancement occurs when the percentage of Ge in the SiGe layer is about 30%. Calculating the corresponding energy shift with the commonly-used deformation potential model [44] and assuming again 30% Ge, $\Delta E_{\rm C} = 209 \,\mathrm{meV}$ is found, which has to be added to the quantization-induced shift (see Fig. 1.8), thus



Figure 1.17: Electron mobility in (100) SG-SOI FETs versus effective field for small silicon thicknesses. Measurements in [16] (symbols), phonon + surface roughness + interface states limited mobility (dotted-dashed), phonon + surfaceroughness + interface-states + thickness-fluctuation + suppression of inter-valley phonon scattering term (dashed), total mobility (solid).

obtaining a value between 250 and 300 meV.

Another important observation is that, according to (1.53)-(1.55), the mobility enhancement in (100) samples is expected to increase as E_{eff} increases, because of the ΔE_{C} growth with E_{eff} (see Fig. 1.8). However, when E_{eff} exceeds 0.3 MV/cm in very thin-film FETs (below 5 nm), additional scattering mechanisms limit the carrier mobility, i.e. the scattering induced by silicon-thickness fluctuations presented in the previous section and the scattering with surface optical phonons, that will be described in the next section.

3) Surface optical phonons

In order to reproduce the mobilities of devices thinner than 2.7 nm at 300 K, the surface-mode optical-phonon scattering contribution is introduced [2], namely

$$\frac{1}{\tau_{\rm SP,v}} = C_{\rm SP0} \exp\left(-\gamma \frac{t_{\rm Si}}{t_{\rm Si0}}\right) \quad , \tag{1.56}$$

where $C_{\rm SP0}$ and γ are fitting parameters calibrated on experiments, and are reported in Appendix A. The exponential form is somewhat arbitrary, even if it reflects the fundamental physical dependencies [17]. A confirmation of the nature of this contribution, i.e. the electron-phonon interaction, is given by the good agreement of the model with experiments for $t_{\rm Si} = 2.48$ nm at 25 K, shown in Fig. 1.15. The inclusion effects of the above scattering terms are highlighted in Fig. 1.17.



Figure 1.18: Electron mobility vs. silicon thickness at 300 K and $E_{eff} = 3.38 \times 10^5 \text{ V/cm}$ for $(110)/\langle 100 \rangle$ SG-SOI devices. Symbols: measurements from [15]. Lines: phonon + surface-roughness limited mobility (dash-dotted); phonon + surface-roughness + thickness-fluctuation limited mobility (dashed); total mobility (solid).

1.10.2 (110) substrates

1) Scattering induced by silicon-thickness fluctuations

Unfortunately, no experimental investigations are available in the literature for (110) SOI at low temperature and for $t_{\rm Si} < 5 \,\mathrm{nm}$ to calibrate the scattering induced by silicon-thickness fluctuations. Thus, the same expressions (1.51) and (1.52) adopted for the conventionally-oriented SOI are used. The model parameters are calibrated directly on experiments at 300 K (see Appendix A). Specifically, η_1 is found to be equal to 7.5 and η_2 to 1.5. The different behaviors of $\tau_{\delta t_{\rm Si}}$ observed for (100) and (110) samples can be attributed to two factors: first, (100) and (110) FETs show different quantization effective masses and subband structures; next, the fabrication processes for (100) and (110) orientations generate different interface characteristics, as already mentioned in section 1.7. These considerations are supported by the fact that a single set of parameters allows us to reproduce the effective mobilities of (110) devices with different current flow directions, i.e. $\langle 100 \rangle$ and $\langle 110 \rangle$.

2) Suppression of inter-valley phonon scattering

The suppression of inter-valley phonon scattering should not take place in (110) FETs, because, as described above, the non-parabolicity effect limits the energy difference between the two ladders below E_{fp} , thus forbidding the suppression of inter-valley scattering and reducing carrier repopulation. This is confirmed



Figure 1.19: Electron mobility in $(110)/\langle 100 \rangle$ and $(110)/\langle 110 \rangle$ SG-SOI FETs versus effective field for small silicon thicknesses. Measurements in [16] (symbols), coulomb + phonon + surface roughness + interface states limited mobility (dotted-dashed), coulomb + phonon + surface roughness + interface states + thickness fluctuation limited mobility (dashed), total mobility (solid).

by the experimental mobility in (110) substrates shown in Fig. 1.18, where the curves monotonically decrease by shrinking the silicon-body thickness.

3) Surface optical phonons

The same expression (1.56) adopted for the conventional orientation is also used in this case. The model parameters are calibrated directly on experiments at 300 K and are reported in Appendix A. The inclusion effects of the above scattering terms are highlighted in Fig. 1.19.

1.11 Mobility in double-gate FETs

As already pointed out in [2], no substantial differences are experimentally observed when (100) double gate instead of single gate devices are considered to justify a modification of the mobility model. The comparison between the model and experiments is illustrated in Fig. 1.20, which shows a satisfactory agreement with experiments. In (110) substrates, a slight mobility enhancement at medium-high electric field is experimentally demonstrated for $(110)/\langle 100 \rangle$ and $(110)/\langle 110 \rangle$ double gate devices [15]. The reasons of this (110) mobility behavior can be attributed to the volume inversion which suppresses surface-roughness



Figure 1.20: Electron mobility in DG MOSFETs vs. effective field for various silicon thicknesses. Solid lines: this model; symbols: measurements from [13] (top) and [14] (bottom).

scattering, because the carriers flow occurs at the center of the SOI layer, far from the Si/SiO₂ interface [15] [14] [45]. Thus, C_{SR_0} in Eq. (1.42) is corrected for a double-gate FET as

$$C_{\rm SR_0}^{\rm DG} = C_{\rm SR_0} - \left(\frac{C'_{vi}}{\exp((t_{\rm Si}/t_{\rm Si0} - C''_{vi})/C'''_{vi}) + 1}\right) \quad , \tag{1.57}$$

where C'_{vi} , C''_{vi} and C'''_{vi} are fitting parameters calibrated directly on experiments from [15] and reported in Appendix A. This fitting is shown in Fig. 1.21 for $(110)/\langle 100 \rangle$ devices at $E_{\text{eff}} = 0.338 \times 10^5 \,\text{V/cm}$, which corresponds to $N_{\text{inv}} = 6 \times 10^{12} \,\text{cm}^{-2}$ for $N_A = 5 \times 10^{15} \,\text{cm}^{-3}$. The same result has been obtained also for $(110)/\langle 110 \rangle$ devices (not shown).

Fig. 1.21 exhibits a mobility enhancement at $t_{\rm Si} = 5 \,\mathrm{nm}$ for the double-gate device architecture, because the volume inversion is more effective in thinner silicon films [15]. However, when $t_{\rm Si}$ is shrunk below about 3.7 nm, silicon-thickness fluctuations and surface optical-phonons prevail on volume inversion, thus masking its effect. The resulting model is shown in Fig. 1.22, together with the experimental mobilities in $(110)/\langle 100 \rangle$ and $(110)/\langle 110 \rangle$ samples, showing a nice agreement.

As discussed above, no volume inversion is visible in (100) experimental data. An extensive investigation based on Monte Carlo simulations has been carried out in [45] on the different mobility behaviors in (100) and (110) DG structures. Due to the different subband structures of (100) and (110) substrates, the second subband of the unprimed valley in (100) samples is close to the energy of the first (lowest) primed subband while, in (110) samples, the second unprimed subband



Figure 1.21: Electron mobility vs. silicon thickness at 300 K for $(110)/\langle 100 \rangle$ SG and DG devices. Symbols: measurements from [15] and [16]. Dashed line: model without the correction for DG devices in (1.57). Solid line: model with the correction in (1.57).



Figure 1.22: Electron mobility vs. silicon thickness at 300 K for $(110)/\langle 100 \rangle$ and $(110)/\langle 110 \rangle$ DG devices. Symbols: measurements from [15]. Dashed line: model without the correction for DG devices in (1.57). Solid line: model with the correction in (1.57).

lies significantly higher than the lowest primed subband. When the electric field increases, the population of the second unprimed subband in (100) FETs grows in DG structures, while remaining quite the same in SG ones. This is verified by Schrödinger-Poisson simulations shown in Fig. 1.23-(B). On the contrary, the higher unprimed subbands in (110) samples remain practically empty. Thus, intersubband scattering in (100) DG-FETs between the first primed and second

34



Figure 1.23: (A) Electron concentration as a function of position along the z axis in SG and DG FETs on (100) and (110)-oriented substrates. (B) Relative population of the second unprimed subband in (100) structures with $t_{\rm Si} = 5$ nm as a function of the effective field in SG and DG devices.

unprimed subbands completely masks the beneficial effect of volume inversion due to the high carrier density in the second unprimed subband.

Moreover, by exploiting our Schrödinger-Poisson solver, the electron concentration profiles have been investigated in the transverse direction in (100) and (110) single- and double-gate FETs. As shown in Fig. 1.23-(A), due to the different band structures, the electron distribution in (100) samples is quite close to the interface both in single- and double-gate FETs at medium-high transverse electric fields ($E_{\rm eff} = 3.5 \times 10^5 \,\mathrm{V/cm}$ in the figure). Instead, it is more spread around the symmetry plane for both device types in (110) samples. These effects can qualitatively justify the stronger impact of volume inversion on (110) structures.

1.12 Review of the complete mobility model

All the scattering terms described so far need to be accounted for to calculate the effective mobility, i.e.

$$\sum_{j} \frac{1}{\tau_{v_{j}}} = \frac{1}{\tau_{\text{PS},v}} + \frac{1}{\tau_{\text{CS},v}} + \frac{1}{\tau_{\text{SR},v}} + \frac{1}{\tau_{\delta t_{\text{Si}}l,v}} + \frac{1}{\tau_{\delta t_{\text{Si}}h,v}} + \frac{1}{\tau_{\text{SP},v}} + \frac{1}{\tau_{it,v}} \quad .$$
(1.58)

The modification in (1.53) has to be applied to the phonon scattering term relative the the unprimed valleys. When considering (110) double gate devices the



Figure 1.24: Electron mobility in (100) SG-SOI FETs versus effective field for a wide range of silicon thicknesses at 300 K. Solid lines: effective mobility model; dashed lines: simulation results obtained by using the local mobility model; symbols: measurements from [12].

parameter of the surface roughness scattering C_{SR_0} changes as in (1.57).

For implementation reasons, drift-diffusion commercial device simulators require a mobility model dependent on the local electron concentration n and transverse electric field E_{\perp} , rather than the nonlocal inversion-layer electron density and effective field. To this purpose, N_{inv} and E_{eff} are simply replaced by n and E_{\perp} in our model, and the fitting parameters are re-extracted by comparison with experiments and are reported in Appendix A-table A.2. Figs. 1.24, 1.25 and 1.26 show the comparison between measurements and local model (dashed lines) for the complete set of measurements data in [12] and [16] for SG-SOI FETs as a function of the effective field and silicon thickness. The effective mobility is also shown with solid lines. The performance of the local mobility model is good, the maximum relative error being 15%. This partly confirms the qualitative results obtained in [46].

1.13 Summary

In this chapter a low-field electron mobility model suitable for device-simulation tools in (100) and (110) UTB-SOI MOSFETs has been presented. The model accounts for the main physical effects related to the quantum-mechanical structural confinement, such as valley repopulation, in-plane anisotropy and nonparabolicity, and transposes them into simple analytical formulations. The correct description of the band structure (energy levels and effective masses) is shown to be fundamental to predict mobility in UTB devices for any arbitrary in-plane current-flow direction. The inclusion of silicon-thickness fluctuations and surface-



Figure 1.25: Electron mobility in $(110)/\langle 100 \rangle$ SG-SOI FETs versus effective field for a wide range of silicon thicknesses at 300 K. Solid lines: effective mobility model; dashed lines: simulation results obtained by using the local mobility model; symbols: measurements from [16].



Figure 1.26: Electron mobility in $(110)/\langle 110 \rangle$ SG-SOI FETs versus effective field for a wide range of silicon thicknesses at 300 K. Solid lines: effective mobility model; dashed lines: simulation results obtained by using the local mobility model; symbols: measurements from [16].

phonon scattering extends the validity of the model to very small silicon thicknesses. Moreover, accounting for the scattering induced by interface states and the impact of volume inversion on surface-roughness scattering, the model is able to reproduce (110) mobilities in SG and DG structures.

Experimental data on (100), (110)/ $\langle 100 \rangle$ and (110)/ $\langle 110 \rangle$ SG and DG MOS-FETs are reproduced with a maximum error of about 15%.

Chapter 2

Mobility Model for Differently-Oriented UTB SOI p-FETs

Recently, it has been experimentally demonstrated that the hole mobility, similarly to the electron mobility, is a sensitive function of the silicon-body thickness, especially when $t_{\rm Si}$ is below 5 nm [47]. A number of experimental investigations can be found on UTB single-gate silicon-on-insulator (SG-SOI) pMOSFETs along with theoretical microscopic analyses accounting for full-band structure and the most relevant scattering mechanisms [48] [49] [50], but a physically-based analytical model for the hole low-field mobility is still missing.

The aim is to derive a TCAD mobility model, suitable for device simulation tools, which accurately predicts the low-field hole mobility in bulk and UTB FETs with different surface and channel orientations and silicon thicknesses from bulk-like to values as small as 2.3 nm.

In the following, a complete description of the model is provided and its validation against experiments is illustrated. The mobility formulation is given in 2.1. The modeling of the scattering contributions, namely, acoustic and optical phonon, Coulomb, surface-roughness and interface-state scattering, is described in section 2.2. An accurate analytical description of the energy subbands is reported in section 2.3. The additional effects needed for ultra-thin SOI are shown in section 2.4. A review of the complete model is reported in section 2.5.



Figure 2.1: Top: orientation of the reference axes. Bottom: in-plane minima projections for (100), (110) and (111) surface orientations. The most relevant in-plane current flow directions are also indicated.

2.1 Generalized mobility model

Consider a silicon film on a substrate with one of the three crystallographic orientations indicated in Fig. 2.1, bottom. The z axis is set parallel to the structural confinement direction, while carrier transport occurs along the x axis. The inversion-layer quantization, due to the combined effects of the structural confinement and the application of a transverse electric field, causes the formation of energy subbands splitted in three different groups: heavy hole (HH), light hole (LH) and split-off subbands. The latter ones have not been considered in this work because of their lower energy, which makes them practically unpopulated [48]. The complex shape of the valence-band valleys makes the analytical calculation of the principal effective masses quite problematic. Therefore a simplified approach has been followed. The masses along the quantization direction (m_z) for the (100), (110) and (111) oriented wafers have been extracted by comparing the valley edges calculated by means of a six-band $\mathbf{k} \cdot \mathbf{p}$ approach reported in [48] with the well-known analytical expression determined for a triangular well by Stern and Howard [22], which reads:

$$E_{V_v} = \left(\sqrt[3]{\frac{9}{32}} + \sqrt[3]{\frac{9}{4}}\right) \sqrt[3]{\frac{\hbar^2 q^2}{m_{z_v}}} E_{\text{eff}}^{2/3} , \qquad (2.1)$$

where \hbar is the reduced Planck constant, q is the elementary charge, E_{eff} is the transverse effective field and m_{z_v} is the quantization mass relative to the v^{th} valley. In the following v = 1 and v = 2 indicate the LH and HH valleys, respectively. The comparison of Eq. (2.1) with the $\mathbf{k} \cdot \mathbf{p}$ results is illustrated in Fig. 2.2, while the extracted quantization masses are reported in table 2.1. In order to



Figure 2.2: Energy of the valley edges as a function of the effective field for (100), (110) and (111) wafers. Symbols: numerical computations from [48]. Solid lines: Eq. (2.1) with a shift of -44 meV.

calculate the 2D density-of-state effective masses, different approaches have been used for the three considered crystallographic orientations. More specifically, in the (100) and (111) orientations circular parabolic in-plane bands for the HH and LH valleys have been assumed. Within such approximation the HH and LH density-of-state effective masses have been extracted by comparing the analytical calculations of the relative valley populations as functions of the effective field with the numerical data reported in [48]. For the analytical calculations, the Boltzmann statistics has been assumed:

$$p_v = \frac{m_{d_v} \exp\left(-E_{V_v}/k_B T\right)}{\sum_{v'=1}^2 m_{d_{v'}} \exp\left(-E_{V_{v'}}/k_B T\right)}.$$
(2.2)

The comparison of the analytical p_v with numerical results is reported in Fig. 2.3.

Differently from the cases described above, in the (110) case a clear in-plane anysotropic energy distribution can be observed (Fig. 2.1, bottom). In this case, elliptical parabolic in-plane bands for the HH and LH valleys have been used, and the effective masses along the transport (m_x) and the device width (m_y) directions have been extracted by comparison with the band calculations reported in [51] as shown in Fig. 2.4. The extracted effective masses are reported in table 2.1. Finally, the density-of-state effective masses in the (110) case have been calculated as $m_{d_v} = \sqrt{m_{x_v} m_{y_v}}$, and their values have been validated by comparing the analytical p_v with the corresponding numerical results shown in [48] (see Fig. 2.3).

Similarly to the electron mobility model in Chapter 1, the in-plane transport is generally described by a 2D tensorial effective mobility which retains the



Figure 2.3: Relative valley populations as a function of the effective field for (100), (110) and (111) wafers. Symbols: numerical calculations by [48]. Solid lines: Eq. (2.2).

anisotropy of the single-valley effective mobilities:

40

$$\hat{\mu}_{\text{eff}} = \sum_{\nu=1}^{2} p_{\nu} \hat{\mu}_{\nu} \,. \tag{2.3}$$

A similar formulation is already available in 3D drift-diffusion transport simulation tools which handle bulk piezoresistivity and, in general, material anisotropy (see, e.g., [27]). Such tools can be directly used also in this case given that the outof-plane (normal to the interface) mobility component plays no role and provided that the in-plane effective mobility model (2.3) is implemented. Unfortunately, this approach is unpopular in commercial tools because the effective mobility dependence on integral (non-local) carrier concentration and electric field may lead to numerical problems. As an alternative, a local mobility tensor $\hat{\mu}$ which depends on the local normal electric field $E_{\perp}(z)$ and the hole concentration n(z)can be defined, satisfying the following equation:

$$\hat{\mu}_{\text{eff}} = \frac{\int_0^{t_{\text{Si}}} (n(z) - n_0(z))\hat{\mu}(n, E_{\perp}(z))dz}{\int_0^{t_{\text{Si}}} (n(z) - n_0(z))dz},$$
(2.4)

where $(n(z) - n_0(z))$ is the excess carrier concentration in the inversion layer. It should be noted that the experiments measure only the xx component of the mobility tensor.

As for the electron mobility described in Chapter 1, a two-step procedure has been followed in the model development. In the first step, an analytical model has been defined for the effective mobility (2.3) as a function of E_{eff} and the inversion-charge concentration per unit area, calibrating the parameters on



Figure 2.4: Energy dispersion relations in (110) wafers obtained by using the $\mathbf{k} \cdot \mathbf{p}$ approach [51] (symbols) and the parabolic band approximation (lines). The $\langle 110 \rangle$ and $\langle 100 \rangle$ in-plane crystal directions have been used to separately extract m_x and m_y .

experiments. E_{eff} is calculated as:

$$E_{\rm eff} = \frac{\int_0^{t_{\rm Si}} (n(z) - n_0(z)) E_{\perp}(z) dz}{\int_0^{t_{\rm Si}} (n(z) - n_0(z)) dz} \,.$$
(2.5)

In the second step, a local mobility model, which depends on $E_{\perp}(z)$ and n(z), is provided as described in section 2.5.

As far as the relative valley populations are concerned, Eq. (2.2) is used. In Eq. (2.2), the subband edges E_{V_v} can be calculated as given by Eq. (2.1) when a bulk MOSFET structure is considered. A different approach needs to be followed when a thin silicon film is addressed, as will be shown in section 2.3. The single-valley mobility tensor is modeled as

$$\hat{\mu}_v = \mu_v \hat{m}_v^{-1}, \quad \hat{m}_v^{-1} = \begin{pmatrix} m_0/m_{xv} & 0\\ 0 & m_0/m_{yv} \end{pmatrix},$$
(2.6)

where \hat{m}_v^{-1} is the inverse normalized mass tensor of a 2DHG, defined for each valley v to account for the anisotropy effects induced by different in-plane crystal directions. Note that when (100) or (111) samples are considered $m_{xv} = m_{yv} = m_{dv}$. Finally, μ_v is calculated as usual by accounting for the different scattering mechanisms combined via Matthiessen's rule:

$$\mu_v = \frac{q}{m_0 \sum_j \tau_{v_j}^{-1}} \,. \tag{2.7}$$

In (2.7), τ_{v_j} represents the average momentum relaxation time (MRT) due to the *j*-th scattering mechanism for the v^{th} valley. The average MRTs are illustrated in the next sections.

(Wafer)/ <channel></channel>	m_z	m_x	m_y	m_d	valleys
(100)	0.25	0.245	0.245	0.245	1 (LH)
	0.29	0.43	0.43	0.43	2 (HH)
$(110)/\langle 110 \rangle$	0.23	0.274	0.348	$\sqrt{m_x m_y} = 0.309$	1 (LH)
	1.8	0.122	0.215	$\sqrt{m_x m_y} = 0.162$	2 (HH)
$(110)/\langle 100 \rangle$	0.23	0.348	0.274	$\sqrt{m_x m_y} = 0.309$	1 (LH)
	1.8	0.215	0.122	$\sqrt{m_x m_y} = 0.162$	2 (HH)
(111)	0.41	0.244	0.244	0.244	1 (LH)
	0.67	0.454	0.454	0.454	2 (HH)

Table 2.1: Effective masses for a 2DHG in (100), (110) and (111) oriented samples in units of m_0 .

2.2 Scattering contributions

42

Following the approach adopted for the electron mobility in Chapter 1, the acoustic phonon-limited inverse MRT relative to the v^{th} valley is calculated as

$$\frac{1}{\tau_{\rm AC,v}} = \frac{C_v m_{d_v}}{W_v} \quad , \tag{2.8}$$

where W_v is the effective width of the hole distribution in the v^{th} valley and C_v is a constant related to the intravalley acoustic-phonon scattering parameters (see Appendix A). Intervalley scattering is not considered in view of the low-field regime (vanishing longitudinal electric field), which is close to equilibrium. The average effective widths W_v are modeled as for the electron mobility (see Eqs. (1.35), (1.36) and (1.37)) and their calibration has been carried out against the numerical predictions of the Schrödinger-Poisson solver, generalized to SG-SOI p-FETs with different crystal orientations, as shown in Fig. 2.5. The extracted parameters are reported in Appendix A. The eigenfunctions are zero at the Si/SiO₂ interfaces. Neumann boundary conditions for the potential are applied to the bottom edge of the buried oxide in SG-SOI FETs. γ , whose theoretically predicted value is 1/3 [21], is found to be equal to 0.29, 0.33 and 0.29 for the HH valleys, while the values of 0.29, 0.17 and 0.24 have been found for the LH valleys of (100), (110) and (111), respectively.

The overall (110) phonon-limited mobility, both from numerical calculations [48] and experiments [47], showed a trend in $E_{\rm eff}$ which differs significantly from the usual $E_{\rm eff}^{-1/3}$ (see Fig. 2.6). At low temperatures such a trend is indeed no longer observed. This may suggest that optical phonons can limit mobility at low $E_{\rm eff}$ and high temperatures, when the two lowest HH bands and the first LH band are close enough to allow intersubband transitions assisted by absorption of optical phonons to play a role. On the contrary, when the confinement induced by $E_{\rm eff}$ becomes stronger, this process is weakened by the distance of the energy



Figure 2.5: Effective widths for the unprimed and primed subband ladders as a function of the effective field at $t_{Si} = 10 \text{ nm}$ (left) and $t_{Si} = 6 \text{ nm}$ (right). Symbols: numerical computations. Solid lines: Eq. (1.35).

levels, and the phonon-limited mobility increases until a usual decreasing trend with E_{eff} is recovered again [48]. In order to model this effect the optical phononlimited inverse MRT relative to the HH band in (110) wafers has been accounted for as

$$\frac{1}{\tau_{\rm OP,2}} = C_{\rm OP} \, m_{d_2} \left(\frac{E_{\rm eff0}}{E_{\rm eff}}\right)^{\zeta} \quad , \tag{2.9}$$

where $\zeta = 0.31$ has been extracted from calculations in [48] (see inset in Fig. 2.6) and $C_{\rm OP}$ is a fitting parameter, whose value is reported in Appendix A. The phonon-limited mobility relative to the HH valley has been calculated by combining the scattering term in (2.9) with (2.8) via Eq. (2.7). In Fig. 2.6 the calculated total phonon-limited mobility curves are compared with those extracted from experiments in [47] with different $t_{\rm Si}$. The details on the modeling of the optical phonon scattering as a function of $t_{\rm Si}$ are discussed in section 2.4.

The Coulomb and surface roughness scattering terms are modeled as in Eqs. (1.39) and (1.42), and the fitting parameters, reported in Appendix A, have been extracted by comparing the analytical model with a large set of experiments for bulk MOSFETs in the three considered crystallographic orientations (see Fig. 2.7). Finally, a mobility degradation with respect to the universal bulk-mobility curve is experimentally observed at low-medium E_{eff} for (110)-oriented SG-SOI FETs with $t_{\text{Si}} = 32$ nm, even if the device is essentially undoped (see Fig. 2.13). Recently, a number of publications reported the amount of interface states in devices with different orientations (e.g. [7], [39]) and indicated that (110)-oriented devices have an interface state density about three times larger than



Figure 2.6: Phonon-limited hole mobility vs. effective field for $(110)/\langle 110 \rangle$ surface extracted from experiments in [47] (symbols) and from our model. Inset: μ_{PS} vs N_{inv} from simulations in [48].

(100)-oriented devices. Moreover, it is known that the buried-oxide interface can influence mobility in UTB SOI-FETs [55]. Thus, when considering SOI-FETs it is appropriate to model an additional contribution which accounts for the mobility reduction induced by interface states

$$\frac{1}{\tau_{it,v}} = C_{it_0} \left(\frac{N_{it}}{N_{it0}}\right) \left(\frac{N_{inv0}}{N_{inv}}\right)^{\zeta} \quad . \tag{2.10}$$

where $N_{it0} = 5 \times 10^{10} \text{ cm}^{-2}$, $\zeta = 0.5$ as reported in the literature and C_{it_0} is a constant extracted from comparison with experiments (see Appendix A). A value of $1.5 \times 10^{11} \text{ cm}^{-2}$ for N_{it} (similar to what experimentally shown in [32]) has been used to reproduce the experiments by Tsutsui et al. [47]. Considering that in a (100)-oriented device N_{it} is typically of the order of $5 \times 10^{10} \text{ cm}^{-2}$ [32], the above scattering contribution is found to be effective only in the (110) case. N_{it} should be considered an effective interface state density accounting also for the back interface defects.

Differently from electrons (see Chapter 1), the hole mobility curves at low temperature reported in [47] are independent of $t_{\rm Si}$ down to 9 nm. Thus, no dependence on $t_{\rm Si}$ has been modeled in (2.10). For $t_{\rm Si} < 9$ nm, additional physical effects become relevant and contribute to degrade mobility, as explained in the following.

2.3 Band structures and repopulation effects

In order to calculate the relative populations of the LH and HH valleys in samples with different t_{Si} and orientations, an analytical function has been developed



Figure 2.7: Hole mobility in (100), (110) and (111) bulk MOSFETs vs. effective field for various substrate doping concentrations. Symbols: (100) experiments from [24], (110) from [52] and (111) from [53] and [54]. Solid lines: our model.

based on physical considerations. Due to the lack of experimental data on (111) substrates, the mobility analysis is limited to the (100) and (110) ones.

For zero normal electric field (quantum well) the analytical solution of the Schrödinger equation provides the expression for the energy levels. The relative distance between the LH and HH valley edges reads:

$$\Delta E_{\rm VT} = E'_{\rm VT} - E_{VT} = \frac{(\hbar\pi)^2}{2t_{\rm Si}^2} \left(\frac{1}{m_{z,1}} - \frac{1}{m_{z,2}}\right).$$
(2.11)

The separation between the energy minima of the two valleys increases with the reduction of $t_{\rm Si}$, and holes mostly populate the HH valley.

At large normal electric fields the energy edges can be theoretically calculated assuming a triangular potential well as in Eq. (2.1) and their difference is indicated with $\Delta E_{\rm VE}$ in the following. Finally, adopting the same formulation used for the conduction band edges in Eq. (1.45), the difference between the valley edges at high normal fields $\Delta E_{\rm V}$ can be expressed as

$$\Delta E_{\rm V} = \Delta E_{\rm VT} \left[1 + (\Delta E_{\rm VE} / \Delta E_{\rm VT})^{\beta} \right]^{1/\beta} , \qquad (2.12)$$

where $\beta = 3.5$. In order to validate Eq. (2.12), the relative valley populations for (100) FETs have been calculated by using Eq. (2.2) as a function of $t_{\rm Si}$ and compared with those reported in [48] (see Fig. 2.8). A clear repopulation



Figure 2.8: Relative occupancy of the HH and LH valleys in (100) oriented samples vs. the silicon thickness at $E_{\text{eff}} = 4 \times 10^5$ V/cm from (2.2). Symbols: numerically calculated relative populations in [48]; lines: this model.

effect is evident at about $t_{\rm Si} = 7$ nm. The HH valley, i.e. the unprimed one, which exhibits the higher transport effective mass (see table 2.1), turns out to be repopulated for $t_{\rm Si} < 7$ nm, with a negative effect on mobility. This effect contributes to the monotonic mobility degradation with decreasing $t_{\rm Si}$ shown in section 2.4. As far as the (110) orientation is concerned, the same formulation in Eq. (2.12) has been used. However, in this case the repopulation effect does not remarkably influence the hole mobility, since the HH valley is almost entirely populated even at very thick silicon films and bulk MOSFETs (Fig. 2.3).

2.4 Model enhancements for $t_{Si} < 5 \text{ nm}$

2.4.1 (100) substrates

1) Scattering induced by silicon-thickness fluctuations

The measured (100) mobility for $t_{\rm Si} < 5$ nm exhibits a strong degradation which was ascribed to the scattering induced by $t_{\rm Si}$ fluctuations [17]. Eqs. (1.51) and (1.52) are used here to reproduce the experiments available at 25 K [12], as shown in Fig. 2.9 (top). The extracted parameters are reported in Appendix A.

2) Suppression of inter-subband phonon scattering

This effect should not take place in (100) FETs, because the energy difference between the HH and LH valleys is lower than the energy of the f-type phonons even at high E_{eff} (see Fig. 2.2), thus forbidding the suppression of the intersubband phonon scattering. This is confirmed by the experimental mobility in



Figure 2.9: Hole mobility vs. effective field at low temperature for (100) (top) and (110)/(110) oriented FETs (bottom). Symbols: measurements from [12] and [47]. Solid lines: Coulomb + surface roughness + silicon thickness-fluctuation limited mobility. The phonon scattering contribution depends on temperature as $T^{1.75}$ [48] thus its influence on mobility at low temperature is quite negligible (1% and 3% of phonon scattering at room temperature at 25 K and 40 K respectively).

(100) substrates reported in Fig. 2.10, bottom, where the curve monotonically decreases by shrinking $t_{\rm Si}$.

3) Surface optical phonons

In order to reproduce the mobilities of devices thinner than 3 nm at 300 K, the surface-mode optical-phonon scattering contribution has been introduced [2]. The effects of the above scattering terms are highlighted in Fig. 2.10, top. The values of the fitting parameters are reported in Appendix A.

2.4.2 (110) substrates

1) Scattering induced by silicon-thickness fluctuations

The same model has been used for the (110) case as well, with parameters extracted by comparison with experiments carried out at 40 K by Tsutsui et al. [47] (see Fig. 2.9, bottom).

As explained in section 2.2, when (110) p-FET is considered, the impact of optical-phonon scattering becomes nontrivial and its dependence on $t_{\rm Si}$ should be correctly modeled. In [47], the phonon scattering limited mobility has been extracted by means of the Matthiessen rule and by assuming that only the phonon

48



Figure 2.10: Hole mobility vs. effective field (top) and silicon thickness (bottom) at 300 K for (100) devices. Symbols: measurements from [12]. Lines: Coulomb + phonon + surface-roughness limited mobility (dashed), Coulomb + phonon + surface-roughness + thickness fluctuation limited mobility (longdashed), Coulomb + phonon + surface-roughness + thickness-fluctuation + surface optical-phonon limited mobility (solid).

contribution would change remarkably with temperature. However, this is incorrect because the scattering induced by thickness fluctuations varies with temperature as well. Indeed, the phonon limited mobility extracted in [47] for $t_{\rm Si} < 5$ nm results to be negative for $N_{inv} < 10^{12}$ cm⁻². Here, the temperature dependence of the MRTs relative to $t_{\rm Si}$ fluctuations has been assumed to be equal in (100) and (110) FETs. Thus, in order to correctly model the phonon-limited mobility in ultra-thin (110) SOI, the latter has been re-extracted from the experiments and used as a reference for the model validation (see open symbols in Fig. 2.11, top).

2) Suppression of inter-valley phonon scattering

As anticipated in section 2.2, the absorbtion of optical phonons between the lowest HH and the lowest LH subbands plays a role at low fields, while it is suppressed at high E_{eff} . The simplest way to introduce the suppression of inter-valley phonon scattering is to model a reduction of the inverse momentum relaxation time (2.9) as a function of ΔE_{V} :

$$\frac{1}{\tau'_{\rm OP,2}} = \frac{1}{\tau_{\rm OP,2}} \left(1 - f(\Delta E_{\rm V}) \right) , \qquad (2.13)$$


Figure 2.11: Phonon-limited mobility in (110)/(110) FETs versus effective field (top) and versus silicon thickness (bottom) for small silicon thicknesses. The measurements are from [47] (symbols). The impact of the suppression on intervalley optical phonons and of the surface-optical phonons are highlighted.

By exploiting the subband structure model presented in the previous section, $f(\Delta E_V)$ in (2.13) is modeled as:

$$f(\Delta E_{\rm V}) = \frac{f_0}{1 + \exp\left[-\left(\Delta E_{\rm V} - \Delta E_{\rm V0}\right)/C_{\rm iv}\right]} \quad , \tag{2.14}$$

where f_0 , $\Delta E_{\rm V0}$ and $C_{\rm iv}$ are fitting parameters (see Appendix A) calibrated on experiments by [47] at $N_{\rm inv} = 3 \times 10^{12} \text{ cm}^{-2}$, as depicted in Fig. 2.11 (bottom). Thus, the mobility enhancement observed in [47] can be nicely reproduced by the above approach. $f(\Delta E_{\rm V})$ becomes significantly different from zero for $\Delta E_{\rm V} >$ 60 meV.

Another important observation is that, according to (2.13)-(2.14), the mobility enhancement in (110) samples is expected to increase as E_{eff} increases, because of the ΔE_{V} growth with E_{eff} (see Fig. 2.2). However, in very thin-film FETs (below 5 nm), additional scattering mechanisms limit the carrier mobility, i.e., the scattering induced by t_{Si} fluctuations (presented in the previous section) and the scattering with surface optical phonons (described below).

3) Surface optical phonons

The same expression for the conventional orientation is also used in this case. The model parameters are calibrated directly on experiments at 300 K. The effects of this scattering term is shown in Figs. 2.11 (top) and 2.13.



Figure 2.12: Hole mobility in (100) FETs versus effective field for a wide range of silicon thicknesses at 300 K. Solid lines: effective mobility model; dashed lines: simulation results obtained by using the local mobility model; symbols: measurements from [12] and [56].



Figure 2.13: Hole mobility in (110)/(110) FETs versus effective field for a wide range of silicon thicknesses at 300 K. Solid lines: effective mobility model; dashed lines: simulation results obtained by using the local mobility model; symbols: measurements from [47].

2.5 Review of the complete mobility model

All the scattering terms described above need to be accounted for in Eq. (2.7) to calculate the effective mobility. When bulk MOSFETs are considered $1/\tau_{it,v}$

should be neglected. Figs. 2.12 and 2.13 compare the predictions of the complete mobility model with the complete set of measurements data. For implementation reasons, commercial drift-diffusion device simulators require a mobility model dependent on the local carrier concentration n and transverse electric field E_{\perp} , rather than the non-local inversion layer carrier density and effective field. To this purpose, $N_{\rm inv}$ and $E_{\rm eff}$ are simply replaced by n and E_{\perp} in our model, and the fitting parameters are re-extracted by comparison with experiments and are reported in Appendix A-table A.4. Figs. 2.12 and 2.13 show the comparison between measurements and local model (dashed lines) for the complete set of measurements data in [12], [56] and [47] for SG-SOI FETs as a function of $E_{\rm eff}$ and $t_{\rm Si}$. The effective mobility is also shown with solid lines. The performance of the local mobility model is good, the maximum relative error being 13%.

2.6 Summary

In this chapter a low-field hole mobility model suitable for device-simulation tools in (100) and (110) UTB-SOI MOSFETs has been presented. The model accounts for the main physical effects related to the quantum-mechanical structural confinement and transposes them into simple analytical formulations. The inclusion of silicon-thickness fluctuations and surface-phonon scattering extends the validity of the model to very small silicon thicknesses. Moreover, accounting for the scattering induced by interface states and the impact of the suppression of intervalley optical phonon scattering, the model is able to reproduce (100) and (110) mobilities. Experimental data on (100), and (110) are reproduced in the model with a maximum error of about 10-15%.

Chapter 3

Effects of High- κ Gate Stacks and Metal Gates

The further miniaturization of CMOS technology requires the use of high- κ gate dielectric, as an alternative of conventional SiO₂, to reduce gate leakage current while keeping unvaried the gate capacitance [57]. Metal-gate electrodes have also attracted attention to overcome the poly-depletion effect that takes place under gate inversion for poly-Si gates resulting in a capacitance loss, and for compatibility with high- κ materials [58]. One of the most suited and very promising candidate is hafnium oxide (HfO_2) combined with titanium nitride (TiN) metal gate. However, apart from the specific materials and the related issues coming from their integration, a degradation of the inversion channel mobility has been regularly reported for devices with high- κ gate stacks. Extensive investigation has been carried out to discover the possible causes of the mobility loss. It has been suggested that the main physical mechanisms playing a role are the coulombic interactions of carriers in the inversion layer with remote charges or dipoles far from the Si/oxide interface (remote Coulomb scattering) [59], scattering events related to the potential modification of channel electrons due to the roughness of the oxide/gate interface (remote surface roughness) [60], and remote soft-phonon scattering [61]. Mobility lowering due to remote scattering mechanisms is observed in general for equivalent oxide thicknesses (EOT) less than about 2 nm, and results more effective for high- κ -based devices. The scattering with low energy optical phonons in high- κ dielectrics is due to the large polarizzability of their bonds and is therefore an intrinsic property [61].

However, even if there is no consensus yet on the relative importance of the different remote scattering mechanisms and on their physical origin, the experimentally observed mobility loss remains a crucial issue. A comprehensive mobility model for numerical simulation of high- κ and metal gate stacks has been lacking



Figure 3.1: Electron mobility as a function of the effective field for devices with SiO_2 /poly-Si gate stack on (100)-oriented Si, and with different oxide thicknesses and substrate doping concentration. Symbols: experiments from [24], [62] and [64]. Lines: this model.

so far, which makes device simulation unreliable, if quantitatively accurate predictions are sought. As shown in Chapters 1 and 2, the use of different silicon crystal orientations can be a way to compensate for the mobility degradation in high- κ MOSFETs, since carrier mobility can be improved by a suitable selection of surface and channel orientations. On the other hand, novel structures, e.g. FinFETs and Tri-gate FETs, widely utilize unconventional surface and channel orientations.

In the following, simple analytical formulations for remote scattering terms have been added to the electron and hole mobility models for differently-oriented FETs presented in Chapters 1 and 2, respectively. Recent experimental results for devices with gate stacks composed of different materials, and with varying oxide thickness, channel doping, crystalline surface orientation and channel direction [54] [62] [63], have been exploited to separately model the scattering contributions in the total mobility.

Aim of this investigation is that of providing an analytical low-field mobility model for MOSFETs with either poly-Si or TiN gate, ultrathin SiO₂/HfO₂ gate stacks and different substrate and channel orientations. Physical insights, theoretical analyses and experimental investigations are used to develop and accurately calibrate the model. In section 3.1, remote scattering terms are modeled and calibrated for devices with ultrathin SiO₂/poly-Si and SiO₂/TiN gate stacks. Remote scattering terms for high- κ dielectrics are then addresses in section 3.2 for (100), (110) and (111) silicon crystal orientations.



Figure 3.2: Electron mobility as a function of the effective field for devices with SiO_2/TiN gate stack on (100)-oriented Si, and with different oxide thicknesses. Symbols: experiments from [62]. Lines: this model.

3.1 Remote scattering effects with thin oxides and metal gates

From recent experimental analyses [62] [64] a mobility lowering with reduced oxide thickness t_{ox} has been observed mainly for electrons for both polysilicon and metal gates (Figs. 3.1, 3.2 and 3.4). Such effect can be attributed to remote Coulomb scattering (RCS), remote soft-phonon scattering (RPS) and remote surface-roughness scattering (RSS) mechanisms: they have been modeled as additional terms in the Matthiessen formulations (1.58) and (2.7) for electrons and holes, respectively, each with an exponential dependence on the oxide thickness (t_{ox}), calibrated on experiments. The semi-empirical formulation of the remote-phonon inverse relaxation time relative to the v^{th} valley reads

$$\frac{1}{\tau_{\text{RPS},v}} = \exp\left(-k_{\text{RPS}} t_{\text{ox}}\right) \frac{C_{\text{RPS}} m_{d_v}}{W_v} \quad , \tag{3.1}$$

with C_{RPS} and k_{RPS} fitting parameters, whose values are reported in Appendix A-table A.5. W_v is given in Eq. (1.35) and $m_{d_v} = \sqrt{m_{x_v} m_{y_v}}$ as explained in Chapter 1.

The remote-surface-roughness-limited inverse relaxation time reads

$$\frac{1}{\tau_{\text{RSR},v}} = C_{\text{RSR}} \exp\left(-k_{\text{RSR}} t_{\text{ox}}\right) m_{d_v} \left(\frac{E_{\text{eff}}}{E_{\text{eff0}}}\right)^{\delta}$$
(3.2)

with C_{RSR} and k_{RSR} fitting parameters (see Appendix A-table A.5). $E_{\text{eff0}} = 10^6 \text{ V/cm}$ and δ is given in Eq. (1.42).



Figure 3.3: Electron mobility as a function of the effective field for devices with $SiO_2/HfO_2/TiN$ gate stack on (100)-oriented Si, and with different oxide thicknesses. Symbols: experiments from [62] and [54]. Lines: this model.

Finally, the inverse relaxation time limited by the remote Coulomb scattering is modeled as

$$\frac{1}{\tau_{\text{RCS},v}} = C_{\text{RCS}} \exp\left(-k_{\text{RC}} t_{\text{ox}}\right) \left(\frac{N_{\text{inv0}}}{N_{\text{inv}}}\right) \left(\frac{N_{A/D}}{N_{A0}}\right)^{\beta_{\text{RCS}}} , \qquad (3.3)$$

where C_{RCS} and k_{RCS} are fitting parameters (see Appendix A-table A.5). $\beta_{\text{RCS}} = 1.96$, $N_{\text{inv0}} = 10^{13} \text{ cm}^{-2}$ and $N_{A0} = 10^{17} \text{ cm}^{-3}$, while $N_{A/D}$ is the acceptor/donor doping concentration when considering n- or p-FETs, respectively.

The calibrated model is shown in Fig. 3.1 for electrons. The hole mobility shows negligible effects related to the oxide thinning as shown in Fig. 3.4. Thus no additional contribution is added to the model.

In [62], an experimental investigation has been carried out on SiO₂/TiN devices in order to address the impact of TiN itself on mobility and to quantitatively determine how it influences the overall degradation, often fully attributed to the high- κ dielectrics. Devices with metal gates (SiO₂/TiN) show a clear mobility reduction at low electric fields, attributed to a higher Coulomb scattering contribution (Fig. 3.2). At higher fields, experiments show a weaker $E_{\rm eff}$ dependence, giving rise to a slight mobility enhancement. As such effects originate from chemical reactions related to the adopted deposition process [62], different values of $C_{\rm CS_0}$ ($C_{\rm CS_0} = 9.13 \times 10^{-4} \text{ s}^{-1}$ for electrons and $8.4 \times 10^{-5} \text{ s}^{-1}$ for holes) and $C_{\rm SR0}$ ($C_{\rm SR0} = 4.98 \times 10^{13} \text{ cm}^2 \text{eV}^{-1} \text{s}^{-3}$ for electrons and $6.9 \times 10^{13} \text{ cm}^2 \text{eV}^{-1} \text{s}^{-3}$ for holes) in (1.39) and (1.42), respectively, are used in the presence of TiN gates. δ , that is equal to 2.7 for (100) oriented n-type devices with poly-Si gate, is found to be 1.8 for TiN metal gates (see Fig. 3.2), in agreement with [65].



Figure 3.4: Hole mobility as a function of the effective field for devices with both SiO_2 /poly-Si and SiO_2 /HfO₂/TiN gate stacks on (100)-oriented Si, and with different oxide thicknesses. Symbols: experiments from [24] and [62]. Lines: this model.

3.2 Remote scattering effects in high- κ stacks

Here the main goal is to develop a mobility model for gate stacks realized with an SiO₂ interfacial layer (with given thickness $t_{\rm IL}$), which either spontaneously interposes or is intentionally grown, and a high- κ dielectric on top (with given thickness $t_{\rm HfO_2}$). To this purpose, the experiments carried out by [62] on devices with metal gates (TiN) and for different $t_{\rm IL}$ and different $t_{\rm HfO_2}$ have been used to calibrate the remote scattering terms. On the other hand, the remote scattering terms related to the presence of the high- κ dielectric are assumed independent of the gate material (polySi or TiN). As no significant dependence on the high- κ layer thickness is experimentally observed, the assumption in [62] that the remote scattering terms are mainly related to the surface at the SiO₂/HfO₂ interface has been followed. Thus, $t_{\rm ox}$ in the scattering terms (3.1), (3.2) and (3.3) has been replaced by the interfacial-layer thickness $t_{\rm IL}$. In Figs. 3.3 and 3.4 the model parameters are calibrated on experiments with $t_{\rm IL}$ down to about 1 nm for electrons and holes, respectively. The extracted fitting coefficients are reported in Appendix A-table A.5.

Finally, the model has been applied to different surface and channel orientations without any change in the previously fitted data. A nice agreement is shown in Fig. 3.5 between the model for electron mobility in $SiO_2/HfO_2/poly-Si$ gate stack and the corresponding experimental results.



Figure 3.5: Electron mobility as a function of inversion electron concentration for devices with SiO_2/HfO_2 gate stacks with poly-Si gate and different surface and channel orientations. Symbols are experiments in [54] and [63]. The thickness of the interfacial layer is 1 nm. Lines: this model.

3.3 Summary

In this chapter, the effects of high- κ and metal gate stacks and surface and channel orientations have been included in the mobility models for electron and holes presented in the previous chapters. Simple analytical formulations are adopted to account for the main physical effects affecting carrier mobility in high- κ based devices and to perform fast and accurate numerical simulations.

Chapter 4

Effects of Mechanical Stress for Planar and Vertical FETs

Among all the new technologies proposed and investigated to continuously improve device performance, strain engineering during the past decade has been the dominant one. It allows to enhance device performance while providing a lowcost and low-risk technique by maintaining the traditional MOSFET fabrication process. In particular, strain has a large effect on the conduction current, because carrier mobility can be enhanced by appropriate stress configurations. The origin of strained-silicon technology can be traced back to silicon layers grown on relaxed silicon-germanium (SiGe) substrates in the 1980s [66]. The thin silicon layer tends to keep the larger lattice constant of the SiGe thus inducing a biaxial tensile stress. In the 1990s, two other strained-silicon techniques have been studied based on process-induced strain to introduce uniaxial tensile and compressive strain into the channel: high-stress capping layers deposited on MOSFETs [67] and SiGe in the source and drain area [68]. Even though the primary focus of the industry in the 1980s and 1990s was on biaxial stress, recently, growing attention is being paid to uniaxial process-induced stress for several reasons. First, uniaxial versus biaxial stress has been demonstrated to provide larger mobility enhancement and drive current improvement for nanoscale short-channel devices due to band warping and reduced conduction effective mass. Second, uniaxial process-induced stress has the advantage of a smaller shift in the threshold voltage [69]. However, because many process flow parameters need to be changed when fabricating strained MOSFETs, there is some uncertainty on whether strain alone is responsible for the performance enhancement (for example, reduced external resistance can itself play some role). Therefore, in order to check the role of strain itself and to investigate the best mechanical stress configuration, the wafer-bending apparatus has been widely utilized by researchers to apply well defined external mechanical stress by simply bending the silicon wafer and characterizing the on-chip devices under stress conditions. The amount of applied stress can be easily extracted from the wafer curvature as explained in [6].

In addition to strained conventional MOSFETs ultra-thin body (UTB) devices with either planar or vertical architectures such as silicon-on-insulator (SOI) FETs, double gate (DG) FETs, FinFETs, Trigate FETs and silicon nanowires (SiNWs) are the most promising candidates for the future technology nodes. A recent research issue is that of investigating the role played by strain in such architectures and eventually combining these new architectures with uniaxial strain engineering to enable additional performance enhancement.

In the previous chapters mobility models for planar UTB SOI structures with unconventional orientations and high- κ oxides and metal gates have been proposed. Further attention needs to be paid to the carrier mobility behavior in non-planar FETs, due not only to the different crystallographic orientations of the top and lateral channels, but also to the complex stress configurations originating from the process steps (see, e.g., [70]). In addition, the experimental studies in narrow FinFETs [71] showed mobility behaviors which still are not clearly explained.

Even if several theoretical works have been performed to analyze straininduced carrier mobility enhancement in both n- and p-MOSFETs [72], [73], a lack of universally accepted mobility models for planar and non-planar structures with both externally-applied or process-induced mechanical stress makes it difficult to perform quantitatively accurate simulations of such devices.

In this chapter, the aim is to provide a unified compact mobility model for electrons and holes comprehensive of all the relevant effects in FinFETs, based on the combination of the top and lateral channels. The model is physically based and extensively calibrated on experiments. It is shown to give a correct interpretation of the mobility features originating in non-planar devices.

In section 4.1 the model for biaxial stress is presented, while in section 4.2 the effects of uniaxial strain for (100) and (110) crystallographic planes are addressed and modeled. Afterwards, the resulting mobility models for electrons and holes are combined to compute the effective mobility of non-planar n- and p-FinFETs by averaging the top- and sidewall-channel mobilities (see section 4.3).

4.1 Strain effects in biaxial conditions

The models described in Chapter 1 and 2 for electron and hole mobilities, respectively, have been extended to include the effects of strain on valley splitting, valley repopulation, band deformation and scattering contributions. The stress tensor $\hat{\sigma}$ or, alternatively, the strain tensor $\hat{\varepsilon}$ in the crystal coordinate system (CCS) is the input of the model. As a first step, the 3×3 stress matrix $\hat{\sigma}_D$, con-



Figure 4.1: Orientation of the device coordinate system (DCS) and the crystal coordinate system (CCS).

ventionally given in the device coordinate system (DCS), have to be opportunely rotated in the CCS (see Fig. 4.1):

$$\hat{\boldsymbol{\sigma}}_{\boldsymbol{C}} = \Re_{D \leftarrow C}^{T} \, \hat{\boldsymbol{\sigma}}_{\boldsymbol{D}} \, \Re_{D \leftarrow C} \,, \tag{4.1}$$

where $\Re_{D \leftarrow C}$ is the rotation matrix from the CCS to the DCS. The six-component vector notation for $\hat{\varepsilon}$ and $\hat{\sigma}_C$ allows to write the strain in the CCS as

$$\hat{\boldsymbol{\varepsilon}} = \mathbf{S}\,\hat{\boldsymbol{\sigma}}_{\boldsymbol{C}} \,\,, \tag{4.2}$$

where **S** is the six-by-six stiffness compliance matrix. The cubic symmetry of the silicon crystal reduces the independent components of **S** to only three, $S_{11} = 7.681 \times 10^{-6} \text{ MPa}^{-1}$, $S_{12} = -2.138 \times 10^{-6} \text{ MPa}^{-1}$ and $S_{44} = 12.56 \times 10^{-6} \text{ MPa}^{-1}$ [27].

The energy shifts of the conduction and of the HH and LH valleys are calculated for biaxial strain following [48]. For the conduction band, the edges of the valleys whose minima are located along the $v^{th} \langle 100 \rangle$ axis is shifted by

$$\Delta E_{C_v}^s = \Xi_d(\varepsilon_{xx} + \varepsilon_{yy} + \varepsilon_{zz}) + \Xi_u \varepsilon_{vv} , \qquad (4.3)$$

where the index v = x corresponds to valleys "1", v = y to valleys "2", and v = z to valleys "3", respectively (see Fig. 4.1). $\Xi_d = 1.1$ eV and $\Xi_u = 9.29$ eV denote the dilatation and the shear deformation potentials of silicon, respectively. The valley repopulation is calculated by adding the strain-induced shifts to the quantization eigenvalues in Eq. (1.27).

For the valence band, the energy shifts for stress along $\langle 100 \rangle$ can be written as [74]

$$\Delta E_{\mathrm{V},1}^{s} = -\frac{\Delta_{0}}{6} + \frac{\delta E_{001}}{4} + \frac{1}{2} \left[\Delta_{0}^{2} + \Delta_{0} \ \delta E_{001} + \frac{9}{4} (\delta E_{001})^{2} \right]^{1/2} , \qquad (4.4)$$

$$\Delta E_{\rm V,2}^s = \frac{\Delta_0}{3} - \frac{\delta E_{001}}{3} , \qquad (4.5)$$

where index 1 indicates the LH band and 2 the HH band, respectively. $\Delta_0 = 44$ meV and $\delta E_{001} = 2b(\varepsilon_{zz} - \varepsilon_{xx})$, where b = -2.33 eV is the shear deformation potential for a strain of tetragonal symmetry, and z is the quantization direction (perpendicular to the conduction plane). The energy shifts induced by the strain are added to the corresponding energy eigenvalues in Eq. (2.2) to the purpose of calculating the relative populations with the Boltzmann statistics approximation.

For holes, a complication arises because the valence band is strongly deformed by strain. The quantization masses m_{z_v} for holes extracted from the subband energy calculations in the unstressed case nicely reproduce the subband energy calculations also in the stressed case as reported in [48], therefore no m_{z_v} change is needed when stress is applied. On the contrary, the in-plane effective masses are strongly modified by strain, as shown by [48] and [75]. The in-plane inverse effective masses are defined as polynomial functions of ε_{xx}

$$\frac{1}{m_{d_v}^{\text{str}}} = \frac{1}{m_{d_v}^{\text{unstr}}} \left(1 + \sum_{j=1,3} b_{v,j} \varepsilon_{xx}^j \right) , \qquad (4.6)$$

where $m_{d_v}^{\text{unstr}}$ are given in table 2.1, and the fitting parameters $b_{v,j}$ are reported in Appendix A-table A.6. The density-of-state masses $m_{d_v}^{\text{str}}(\varepsilon_{xx})$ for strained silicon grown on $\text{Si}_{(1-x)}\text{Ge}_x$ substrates with different Ge content x are reported in Fig. 4.2, along with the mobility enhancement due to band changes only, compared with theoretical calculations by Wang et al. [73]. The empirical relationship between Ge content x and strain in Si reads:

$$\varepsilon_{xx} = \frac{(a_{\rm SiGe} - a_{\rm Si})}{a_{\rm Si}}, \quad a_{\rm SiGe} = a_{\rm Si} + a_0 x (1 - x) + (a_{\rm Ge} - a_{\rm Si}) x^2, \quad (4.7)$$

where $a_{\rm Si} = 5.4331 \times 10^{-8}$ cm, $a_{\rm Ge} = 5.646 \times 10^{-8}$ cm and $a_{\rm SiGe}$ are the lattice constants of pure Si, pure Ge and SiGe, respectively, and $a_0 = 0.200326 \times 10^{-10}$ cm.

For both electrons and holes the valley repopulation and mass deformation alone do not explain the mobility enhancement shown by experiments. In fact, when strain is applied, the interband optical phonon scattering is reduced due to the band splitting, and the mobility is enhanced (see section 1.10.1). Moreover,



Figure 4.2: Modeled density-of-states effective masses $m_{d_1} = m_{d_{LH}}$ and $m_{d_2} = m_{d_{HH}}$ (lines) compared with data extracted from subband repopulation [48] and $\mathbf{k} \cdot \mathbf{p}$ band calculations [75] (symbols). The mobility gain contribution due to band changes (lines) nicely follows the data based on the $\mathbf{k} \cdot \mathbf{p}$ band structure [73] up to x = 0.8 (triangles).

the assumption of a beneficial effect of strain on the interface as indicated by the experiments is followed. Thus, the inverse momentum relaxation times (MRTs) are modeled as a function of strain. For electrons, the same polynomial function has been applied to all the relaxation time contributions:

$$\sum_{j} \frac{1}{\tau_{v_j}^{\text{str}}} = \sum_{j} \frac{1}{\tau_{v_j}^{\text{unstr}} (1 + \eta_1 \varepsilon_{xx} + \eta_2 \varepsilon_{xx}^2 + \eta_3 (\varepsilon_{xx} - \varepsilon_{yy}))} , \qquad (4.8)$$

where $\sum_{j} \frac{1}{\tau_{v_j}^{\text{str}}}$ is given by Eq. (1.58) and $\eta_1 = 109.95$, $\eta_2 = -616.76$ and $\eta_3 = -30.00$ are fitting parameters independent of the valley index. Calibration of η_1 and η_2 is shown in Fig. 4.3 with experiments by Driussi et al. [76], Nayfeh et al. [77] and Rim et al. [78].

For holes, the valence band is strongly warped and non-parabolic effects arise when considering the band deformed by strain. As far as the relaxation times are concerned, only the phonon-limited contribution explicitly shows a dependence on strain. The strain effect was already shown in the literature by Fischetti et al. [48] and Lee et al. [25]. Here, the expression of the effective width given by Eq. (1.37) has been extended to account for the effects of strain and to reproduce theory and experiments as shown in Figs. 4.4 and 4.5. The strain-dependent effective width reads:

$$W_{\mathrm{E}_{v}}^{\mathrm{str}} = W_{\mathrm{E}0_{v}}^{\mathrm{unstr}} \sqrt[3]{\frac{\hbar^{2}}{qm_{z_{v}}}} E_{\mathrm{eff0}}^{-\gamma^{\mathrm{unstr}}} \left(\frac{E_{\mathrm{eff}}}{E_{\mathrm{eff0}}}\right)^{-\gamma^{\mathrm{str}}} , \qquad (4.9)$$



Figure 4.3: Effective mobility for electrons as a function of the effective field in strained-Si grown on $Si_{(1-x)}$ Ge_x subtrates with different Ge content. Symbols: experiments from [76], [77] and [78]. Lines: this model.



Figure 4.4: The exponent of the phonon-limited mobility given by Eq. (4.10) and calibrated on experiments as a function of strain, compared with both theoretical predictions [48] and experimental extractions [25].

with $W_{\rm E0_v}^{\rm unstr}$, $\gamma^{\rm unstr}$ and $E_{\rm eff0}$ the same introduced in section 2.2,

$$\gamma^{\text{str}} = \gamma^{\text{unstr}} + \frac{p_0 \varepsilon_{xx} + p_1 \varepsilon_{xx}^2}{1 + p_2 \varepsilon_{xx}^2} , \qquad (4.10)$$

with $p_0 = -18.32$, $p_1 = 2.10 \times 10^4$ and $p_2 = 2.74 \times 10^4$. Finally, the phononlimited relaxation time extended to account for the strain effects reads:

$$\frac{1}{\tau_{AC,v}^{\text{str}}} = \frac{1}{\tau_{AC,v}^{\text{unstr}}} \left(f_0 + \frac{f_1}{1 + \exp\left[(\varepsilon_{xx} - \varepsilon_0)/\varepsilon_1\right]} \right)$$
(4.11)



Figure 4.5: Effective mobility model for holes as a function of the effective field in strained-Si grown on $Si_{(1-x)}$ Ge_x substrates with different Ge content. Symbols: experiments from [78], [79] and [80]. Lines: this model.

with $\tau_{AC,v}^{\text{unstr}}$ given by Eq. (2.8), $f_0 = 0.33$, $f_1 = 0.7$, $\varepsilon_0 = 0.008$ and $\varepsilon_1 = 0.00248$. The calibration of the overall hole mobility model is shown in Fig. 4.5 with experiments by Rim et al. [78], Mizuno et al. [79] and Leitz et al. [80].

4.2 Uniaxial stress effects for (100) and (110) crystallographic orientations

The equienergy surfaces of the six conduction band minima and of the HH and LH valleys projected on the (100) and (110) planes are reported in Fig. 4.6, as well as the relative effective masses along with the relevant directions.

4.2.1 Electron mobility

The experimental and theoretical analyses carried out by [81] and [72] have been considered in order to accurately model uniaxial-strain effects for (100)- and (110)-oriented nFETs. In particular, whenever the strain tensor in the principal crystal system contains nonvanishing shear elements (e.g., $\varepsilon_{xy} \neq 0$ as a result of uniaxial stress along (110) crystal direction), the band-edge energy of the valley pair "3" moves down with respect to the valleys "1" and "2" of a quantity ΔE^{shear} defined as:

$$\Delta E^{\text{shear}} = -\frac{\Theta}{4k^2} \varepsilon_{xy}^2, \quad |\varepsilon_{xy}| < k \tag{4.12}$$

where $\Theta = 0.53$ eV is the band separation between the two lowest conduction bands at the conduction-band edge of the unstrained lattice and k = 0.0189[72]. The valley repopulation is calculated by adding the shear-strain-induced



Figure 4.6: Isoenergetic surfaces projected on the (100) and (110) planes for electrons and holes, along with the relative principal effective masses.

shift to the quantization eigenvalues in Eq. (1.27). Moreover, valley pair "3" results deformed (see inset of Fig. 4.7) under $\langle 110 \rangle$ tensile stress. The in-plane effective mass m_t of valleys "3" is changed differently in the parallel $\langle 110 \rangle$ and perpendicular $\langle -110 \rangle$ direction with respect to the stress as:

$$m_{t\langle 110\rangle}^{\rm str} = m_t^{\rm unstr} (1 + \gamma^{\rm str} \varepsilon_{xy}) \tag{4.13}$$

$$m_{t\langle -110\rangle}^{\rm str} = m_t^{\rm unstr} (1 - \delta^{\rm str} \varepsilon_{xy}) \tag{4.14}$$

where $m_t^{\text{unstr}} = m_t$, $\gamma = 3.089$ and $\delta = 5.296$. The effective masses modulated by strain are reported in Fig. 4.7 compared with numerical calculations reported in [81]. On the other hand no band warping is observed when uniaxial $\langle 100 \rangle$ stress is applied [81]. From the above expressions the corresponding 2DEG mass tensor and the density-of-state effective mass can be easily obtained for the mobility computation (see section 1.5).

$(100)/\langle 110 \rangle$ nFETs

In (4.13) and (4.14), $m_{t\langle 110\rangle}$ and $m_{t\langle -110\rangle}$ correspond to m_{x_3} and m_{y_3} in the model, respectively, for conventional (100)/(110) FETs (see section 1.5). The inverse MRTs change with strain is accounted for as in (4.8). The model for uniaxiallystressed (100)-nFETs is shown in Fig. 4.8 for both compressive and tensile stress and for stress directed both parallel and perpendicular to the transport direction.

Uniaxial tensile $\langle 110 \rangle$ stress on (100) silicon advantageously contributes to repopulate the unprimed two-fold valley "3", shifting downwards the corresponding valley edge with respect to the one relative to the primed ladder "1"-"2" (Eq. (4.3)). However, since the (100) primed and unprimed valley edges are

4.2. UNIAXIAL STRESS EFFECTS FOR (100) AND (110) CRYSTALLOGRAPHIC ORIENTATIONS 67



Figure 4.7: The two-fold valleys "3" are warped by the $\langle 110 \rangle$ strain: the in-plane effective mass m_t is differently changed in the parallel and perpendicular direction. The m_t models (lines) are compared with numerical data by [81] (symbols).



Figure 4.8: (Left) Effective electron mobility under uniaxial $\langle 110 \rangle$ tensile and compressive strain for $(100)/\langle 110 \rangle$ nFETs. (Right) Electron mobility enhancement under tensile uniaxial $\langle 110 \rangle$ strain parallel and perpendicular to the transport direction. Experiments: [82]. Lines: this model.

quite separated due to large difference between the quantization masses (see Fig. 4.6), the repopulation effect is weak and saturates at small stress levels, while the interband scattering is remarkably reduced by strain-induced valley splitting. In addition, the decrease of the transport effective mass m_{x_3} shown in Fig. 4.7 when tensile stress is applied in the transport direction is also beneficial (Eq. 1.28), especially under high mechanical stress.

A final validation of the mobility model under uniaxial $\langle 110 \rangle$ stress is reported for electrons in Fig. 4.9, where the mobility enhancement as a function of strain



Figure 4.9: Effective electron mobility enhancement as a function of strain. Experiments: [76], [77], [41], [42], [6] [81].

is compared with experimental results obtained using high strain wafer bending. The biaxial case is also reported for comparison, showing that the enhancement for uniaxial $\langle 110 \rangle$ stress is higher than for the biaxial case due to the effective mass deformation with shear strain. A negligible dependence on effective field is observed on a range from 0.4 to 1 MV/cm (not shown).

$(100)/\langle 100 \rangle \ \mathrm{nFETs}$

For completeness, the model has been generalized to account for uniaxial stress along $\langle 100 \rangle$ in-plane crystal directions. It is worth noting that the last term in the denominator of (4.8) is different from zero only when uniaxial stress along $\langle 100 \rangle$ crystal directions is applied ($\varepsilon_{xx} \neq \varepsilon_{yy}$). The parameter η_3 in (4.8) has been therefore calibrated on experiments for (100)/ $\langle 100 \rangle$ nFETs, as shown in Fig. 4.10 for both compressive and tensile stress and for stress directed both parallel and perpendicular to the transport direction.

$(110)/\langle 110 \rangle \text{ nFETs}$

For (110)/ $\langle 110 \rangle$ nFETs, $m_{t\langle 110 \rangle}$ and $m_{t\langle -110 \rangle}$ in (4.13) correspond to m_{x_3} and m_{z_3} , respectively (see section 1.5). The variation of m_{z_3} is neglected in our model for (110) crystal orientation, since it would make the energy splitting between the unprimed four-fold ladder "1"-"2" and the primed two-fold ladder "3" to be overestimated. The energy splitting calculated with full-band approach in [34] when uniaxial stress is applied along $\langle 110 \rangle$ is thus nicely reproduced by the analytical model in (4.3) with the additional shear term in (4.12) (see Fig. 4.11).



Figure 4.10: (Left) Effective electron mobility under uniaxial $\langle 100 \rangle$ tensile and compressive strain for $(100)/\langle 100 \rangle$ nFETs. (Right) Electron mobility change under tensile uniaxial $\langle 100 \rangle$ strain parallel and perpendicular to the transport direction. Experiments: [82]. Lines: this model.



Figure 4.11: Energy difference between 2-fold valley and 4-fold valley minima of (110) crystallographic plane as a function of strain. The model (solid lines), accounting for additional shift due to shear strain (4.12), is compared with calculations in [34]. The model in (4.3) is also reported for comparison (dashed lines).

Differently from the (100) orientation, where the most relevant effect influencing electron mobility is the variation of scattering contributions with strain, in (110)-oriented planes valley repopulation and effective mass change are dominant in the mobility enhancement. For (110)/ $\langle 110 \rangle$ nFETs the applied stress is advantageous when the primed two-fold valley "3" is repopulated, because of its lower effective mass along the transport direction (see Fig. 4.6). This holds when the primed and unprimed valley edges are moved closer to each other by



Figure 4.12: Electron mobility enhancement for (110)-nFETs as a function of strain for different configurations. Symbols: numerical calculations from [34]. Lines: this model.

applying uniaxial tensile stress along the transport direction or compressive along the width direction.

$(110)/\langle 100 \rangle \text{ nFETs}$

On the other hand, in $(110)/\langle 100 \rangle$ nFETs the stress produces a mobility enhancement when the unprimed four-fold ladder "1"-"2" is repopulated, because of its lower effective mass along the transport direction (see Fig. 4.6). This condition is accomplished again if tensile stress along the transport direction or compressive along the width direction is applied.

In Fig. 4.12, all the favorable conditions above are examined for high strain levels, where the analytical model is compared with numerical calculations from [34]. No change in the scattering contributions with stress is considered. It is worth noting that $(110)/\langle 110 \rangle$ nFETs shows higher mobility enhancement than $(110)/\langle 100 \rangle$ nFETs. Moreover, $(110)/\langle 110 \rangle$ nFETs under uniaxial $\langle 110 \rangle$ stress shows up to be the best configuration, since the mobility enhancement does not saturate even at high strain levels, thanks to the reduction of the conduction effective mass. On the other hand, the mobility enhancement by uniaxial $\langle 100 \rangle$ compressive stress saturates at high stress, since the change of the electron population saturates at large energy splittings. The model for uniaxially-stressed (110)-nFETs is validated on experiments for different stress configurations in Fig. 4.13 for low strain level and in Fig. 4.14 with numerical calculations from [34] for high strain.

4.2. UNIAXIAL STRESS EFFECTS FOR (100) AND (110) CRYSTALLOGRAPHIC ORIENTATIONS 71



Figure 4.13: Electron mobility variation versus inversion charge concentration in $(110)/\langle 110 \rangle$ and $(110)/\langle 100 \rangle$ nFETs. Symbols: experiments from [34]. Lines: this model.



Figure 4.14: Electron mobility versus inversion charge concentration in (110)/(110) and (110)/(100) nFETs. Symbols: calculations from [34]. Lines: this model. The (100) universal curve is also shown for comparison.

4.2.2 Hole mobility

$(100)/\langle 110 \rangle \ \mathrm{pFET}$

As far as the valence band deformation under uniaxial stress conditions is concerned, starting from the circular parabolic band model, the change in the parallel $\langle 110 \rangle$ and perpendicular $\langle -110 \rangle$ directions with respect to the stress has been considered as depicted in Fig. 4.15.



Figure 4.15: The HH band is modeled as a single ellipsoid with stress dependent transverse $(m_{(HH)} < -110 > = m_{y_2})$ and longitudinal $(m_{(HH)} < 110 > = m_{x_2})$ masses (lines). Symbols: data by [83] and [75].

The model reads:

$$\frac{1}{m_{x_2}^{\text{str}}} = \frac{1}{m_{x_2}^{\text{unstr}}} \left(1 + \sum_{j=1,2} c_{2,j} \, \varepsilon_{xy}^j \right) \tag{4.15}$$

$$\frac{1}{m_{y_2}^{\text{str}}} = \frac{1}{m_{y_2}^{\text{unstr}}} \left(1 + \sum_{j=1,2} d_{2,j} \, \varepsilon_{xy}^j \right) \tag{4.16}$$

where $m_{x_2}^{\text{unstr}} = m_{y_2}^{\text{unstr}} = 0.43m_0$ (see Fig. 4.6) and the fitting parameters $c_{2,j}$ and $d_{2,j}$ are reported in Appendix A-table A.6. The phonon inverse MRT variation with strain is accounted for as in (4.11). From the above expressions the corresponding 2DHG mass tensor and the density-of-states effective mass can be easily obtained for the hole mobility computation (see section 2.1). The model for uniaxially-stressed (100)-pFETs is shown in Fig. 4.16 for both compressive and tensile stress.

A final validation of the mobility model under uniaxial $\langle 110 \rangle$ stress is reported for holes in Fig. 4.17, where the mobility enhancement as a function of strain is compared with experiments on uniaxial $\langle 110 \rangle$ compressive strain (using wafer bending [6] or extracted from devices with SiGe source/drain regions or compressive contact etch-stop layers [84]). The biaxial results are also reported for comparison. The hole mobility presents the highest enhancement with uniaxial compressive stress (validated with experiments up to a factor 3), mainly due to the HH transport effective mass (m_{x_2}) reduction in Eq. (4.15). A relevant dependence on electric field is observed for both biaxial and uniaxial stress cases, 4.2. UNIAXIAL STRESS EFFECTS FOR (100) AND (110) CRYSTALLOGRAPHIC ORIENTATIONS 73



Figure 4.16: Effective hole mobility under uniaxial $\langle 110 \rangle$ tensile and compressive strain for (100)-oriented substrates. Experiments: [82]. Lines: this model.



Figure 4.17: Effective hole mobility enhancement as a function of strain. Experiments: [80], [41], [42], [6] [84] [82]. Lines: this model.

showing that the hole mobility enhancement with strain is partly lost at higher transverse effective electric fields.

$(110)/\langle 110 \rangle \text{ pFETs}$

As fas as the (110) valence band is considered, HH valley presents ellipsoidal shape in the (110) plane (see Fig. 4.6). In order to take into account the HH band deformation under $\langle 110 \rangle$ uniaxial stress, the density-of-states effective mass $m_{d_2} = \sqrt{m_{x_2}m_{y_2}}$ has been modeled as a function of shear strain following [75] as



Figure 4.18: Hole mobility variation in (110)/(110) pFETs under uniaxial stress along the transport direction versus stress. Symbols: experiments [53], [85]; lines: this model. Inset: density-of-state effective mass variation as a function of stress. Symbols: model from [75]



Figure 4.19: Hole mobility variation in pFinFET under uniaxial stress along the fin-height direction versus effective electric field. Symbols: experiments [86]; lines: model.

shown in the inset of Fig. 4.18. The model reads:

$$m_{x_2}^{\text{str}} = m_{x_2}^{\text{unstr}} \left(f_0 + \frac{f_1}{1 + \exp\left(\varepsilon_{xy}/\varepsilon_0\right)} \right) \,, \tag{4.17}$$

with $m_{x_2}^{\text{unstr}} = 0.122 \, m_0$ (see Fig. 4.6), $f_0 = 0.6$, $f_1 = 0.8$ and $\varepsilon_0 = 1.884 \times 10^{-3}$. m_{y_2} is left unchanged. The increase (decrease) of m_{d_2} in the presence of



Figure 4.20: 3D schematic view of the SiNW FET. Also shown the crystallographic orientations of the top, bottom and lateral planes along with the crystallographic directions of the transport, the device width and the device height directions.

uniaxial $\langle 110 \rangle$ tensile (compressive) strain makes the effective mobility to decrease (increase), as shown in Fig. 4.18.

$(110)/\langle 100 \rangle \text{ pFETs}$

On the other hand, the uniaxial $\langle 100 \rangle$ strain does not affect m_d and only a repopulation effect occurs, thus being less effective on mobility as shown in Fig. 4.19, where the model is compared with the slight experimental mobility enhancement observed when tensile uniaxial stress is applied along the fin height direction of a narrow FinFET (i.e., where the transport occurs manly on (110) sidewalls, as will be explained in detail in the next section).

4.3 Mobility model for non-planar FETs

As already mentioned before in this thesis, multi-gate (MuG) devices with a nonplanar structure are being studied for the next technology nodes. Several experimental demonstrations have been reported in the last years, moving from partially gate-covered channels, as vertical double-gate FETs or FinFETs (where the gate stack is on three sides), to gate-all-around devices, such as silicon nanowires (SiNW) FETs.

In Fig. 4.20 a 3D sketch of the rectangular SiNW FETs is shown, where the semiconductor is completely surrounded by the gate stack, which ensures the best electrostatic control on the conducting channel. The two most common orientations are shown, where the sidewalls are (100)- or (110)-oriented, depending on the crystallographic direction of the transport.

Experimental measurements have been recently carried out on long-channel rectangular SiNW-FETs and FinFETs with different widths and orientations,



Figure 4.21: Recently extracted experimental mobilities from [87] and [88] for vertical FETs (i. e., rectangular SiNW FETs with all the interfaces oriented along (100) crystallographic orientation and narrow FinFETs with sidewalls along (110) crystallographic planes) are compared with mobilities for planar FETs on (100) and (110) crystallographic orientations.

showing that the effective mobilities in non-planar devices with a dominant transport contribution along the $(100)/[1\bar{1}0]$ or $(110)/[1\bar{1}0]$ directions tend to the corresponding universal curves, with the exclusion of the values at low N_{inv}, which are affected by a stronger impact of the interface states, as shown in Fig. 4.21. The above results suggest that a unified mobility model for (100)- and (110)- oriented UTB planar devices could be applied to non-planar device mobilities by correctly combining the top- and sidewall-channel contributions.

A 3D schematic of the FinFET is reported in Fig 4.22. The typical orientation with sidewall channels on a (110) crystallographic plane and the transport along the [110] direction is illustrated. The equienergy surfaces of the six conduction band minima and of the HH and LH valleys projected on the (110) plane are also reported, as well as the relative effective masses along with the relevant directions. The top channel lies on a conventional (100) plane with [110] transport direction. In the following, a FinFET structure is considered, but the model can be easily extended to SiNW FETs. As proposed in [71], the effective mobility of a FinFET of W > 75 nm can be written as the average of the effective mobilities in the top and sidewall channels:

$$\mu_{\rm eff} = \frac{\mu_{\rm top}W + \mu_{\rm lateral}2H}{2H + W} \,. \tag{4.18}$$

Eq. (4.18) holds under the assumption that the charge is equally distributed in the top and sidewall channels. Otherwise, the inverse charge contributions have to be taken into account. By using Eq. (4.18) and the models described so far, the



Figure 4.22: 3D schematic view of a FinFET. Also shown the isoenergetic surfaces projected on the (110) plane for electrons and holes, along with the relative conduction and quantization effective masses.

effective mobility for both electrons and holes is nicely reproduced for FinFETs with $SiO_2/HfO_2/TiN$ gate stack and with W down to 75 nm (see Fig. 4.23 for hole mobility with no stress). For W < 75 nm, the rounding of the fin corners must be taken into account. Following [89], a (111) crystallographic orientation can be assumed in the corners. Hence, the FinFET mobility can be written as:

$$\mu_{\rm eff} = \frac{\mu_{\rm top} W' + \mu_{\rm lateral} 2H' + \mu_{\rm c} L_{\rm c}}{2H' + W' + L_{\rm c}}, \qquad (4.19)$$

where μ_c is the corner mobility, and W', H' and L_c are the effective width, height and corner extensions, respectively, which need to be extracted from TEM images as shown in the inset in Fig. 4.23. By using Eq. (4.19), the carrier mobility is nicely predicted for FinFETs with W down to 25 nm (Fig. 4.23).

For width narrower than 25 nm, the presence of intrinsic stress related to the fabrication process can not be neglected, especially when metal gates are used [70]. In such narrow structures, the fin height is usually about 50-60 nm and the transport occurs mainly on (110)-oriented sidewalls, thus the above modeled strain effects for (110) FETs are essential to correctly evaluate the carrier mobility. In Figs. 4.24 and 4.25, the model is nicely compared with experiments on narrow FinFETs with different stress configurations. The corresponding (110) mobilities for planar devices are also reported for comparison. The slight differences between the unstrained mobilities (filled symbols) are due to stronger impact of interface charges in complex 3D structures ($N_{it} = 9 \times 10^{11}$ cm⁻² has been used in Eq. (1.50) to fit the unstressed FinFET mobility from [88]). Moreover, remote scatterings are considered for thin oxides (FinFETs measured in [88] have $t_{\rm ox} = 2.2$ nm) and for SiO₂/HfO₂/TiN gate stack (FinFETs measured in



Figure 4.23: Hole mobility of FinFETs with W ranging from 2875 to 25 nm. Symbols: experiments [71]. Solid lines: this model using Eq. (4.18). Dashed line: this model using Eq. (4.19). Inset: TEM image of the narrow FinFET (W = 25 nm) [71], with the extracted geometrical extensions.

[70] have $t_{\rm IL} \approx 1$ nm). For electrons, strained FinFETs can reach a relevant mobility enhancement when a longitudinal tensile stress is applied, with a resulting mobility comparable to the bulk (110) case under similar stress. Vertical compressive strain in FinFET channel, due to both intrinsic stress in TiN metal gate and to differences in thermal expansion coefficients following plastic relaxation at high temperature [70], also contributes to enhance electron mobility. Finally, into such narrow fins almost fully elastic strain relaxation is known to take place along the fin width direction [88], [90], thus the effects of the corresponding strain component are not studied in this thesis.

On the contrary, the hole mobility shows a slight enhancement at low inversion charge concentrations and degradation at higher inversion charge concentrations with respect to the (110) planar case, respectively, when a high compressive vertical strain is present in the channel. A strong tensile stress along the transport direction seriously degrades hole mobility. This suggests that, assuming a symmetric variation of the HH conduction mass with strain, longitudinal compressive stress would strongly enhance hole mobility. Unfortunately, to the author knowledge, no experimental measurements with high compressive longitudinal stress are available in literature to date to validate the model. The vertical compressive stress induced by 3 GPa tensile TiSiN metal gate (MG), with thickness 7 nm in [91], is estimated by the formula:

$$\sigma_{\rm Si} \approx \frac{\sigma_{\rm MG}}{\left(W_{\rm fin}/t_{\rm MG} + Y_{\rm MG}/Y_{\rm Si}\right)},\tag{4.20}$$

with σ_{MG} and t_{MG} the intrinsic stress in the metal electrode and its thickness, respectively, and Y_{MG} (for TiSiN [92]) and Y_{Si} the elastic moduli of the metal



Figure 4.24: Electron mobility of $(110)/[1\bar{1}0]$ planar and vertical devices under different stress conditions. Symbols: experiments [88], [70] and numerical calculations [34]. Lines: this model. A compressive vertical strain of 0.45% in assumed for the data in [70], as measured for a 20 nm-wide FinFET.



Figure 4.25: Hole mobility of $(110)/[1\overline{10}]$ planar and vertical devices under different stress conditions. Symbols: experiments [93], [70], [91]. Lines: this model. A compressive vertical stress of about 500 MPa is estimated to be induced by 3 GPa tensile metal electrode in [91].

gate and silicon underneath, respectively. With $Y_{\rm MG} \approx 500$ GPa (for TiSiN [92]), $Y_{\rm Si} \approx 130$ GPa and $W_{fin} = 11$ nm, a value of about 500 MPa is estimated and used in the simulations.

4.4 Summary

In this chapter, a new analytical model for the electron and hole low-field mobility in ultra-thin body structures with differently-oriented channels and accounting for the effects of mechanical stress has been developed and calibrated on planar FETs. The new model is used to compute the effective mobility of non-planar FinFETs averaging the top- and sidewall-channel mobilities. Experimental data for a wide range of planar and non-planar devices are nicely reproduced demonstrating a good predicting capability in FinFETs. The model can be useful to extend the standard TCAD commercial tools to the performance prediction of post-CMOS technology nodes.

Chapter 5

QDD simulations

In the first part of this thesis, unified models for the low-field effective electron and hole mobilities in MOSFETs, which feature ultrathin SiO_2/HfO_2 gate stacks, different substrate and channel orientations, quantum-confinement effects for single-gate (SG) and double-gate (DG) SOI FETs, and general stress conditions, are presented. In the previous chapter the models for planar FETs have been combined and applied to the study of the effective mobility in vertical architectures. In this chapter the mobility models are included in a 1D quantum drift-diffusion (QDD) solver in order to investigate the *I-V* characteristics improvements related with mobility enhancement.

The QDD tool features are illustrated in section 5.1. Then the simulation studies of nMOSFETs aimed at investigating the impact of different surface orientations, silicon-film thicknesses (for SOI and DG devices) and uniaxial stresses are decribed in sections 5.2, 5.3 and 5.4, respectively.

5.1 1D quantum drift-diffusion model

For relatively-long devices, i.e. for channel lengths greater than the average distance that the charge carrier travels in the semiconductor before being scattered (carrier mean-free path), estimated to be in the 10-20 nm range (depending on the channel length and the transverse electric field [94]), an accurate simulation of carrier transport is possible using a technique which combines the solution of the coupled SchrödingerPoisson equations within the device cross sections normal to the transport direction with the classical drift-diffusion model. This approach is referred to as quantum drift-diffusion (QDD). The theoretical foundations of such approach, given in [95], are reported in the following.



Figure 5.1: Schematic of the ultrathin-body device structure. The device coordinate system consists of orthogonal axes x, y, and z along the transport, width, and thickness directions, respectively.

Semiconductor devices where the electron gas is structurally confined in onedimension, such as UTB SG- or DG-FETs are considered here. However, the treatment holds for non-planar devices, such as FinFETs and SiNW FETs, as well with minor changes. As already pointed out in section 1.3, the physical description of the device electrostatics can be derived by solving the Poisson and Schrödinger equations, namely

$$\nabla_{\mathbf{r}} \cdot (\epsilon_{\mathrm{Si}} \nabla_{\mathbf{r}} \phi) = \rho \tag{5.1}$$

$$\left(-\frac{\hbar^2}{2}\left(\nabla_{\mathbf{r}}\cdot\frac{\nabla_{\mathbf{r}}}{\hat{m}}\right) + U(\mathbf{r})\right)\psi = E\psi \quad , \tag{5.2}$$

where the kinetic energy operator has been written in compact form, with $1/\hat{m}$ the inverse effective-mass tensor. The device coordinate system is oriented as shown in Fig. 5.1, where the z axis is normal to the Si/SiO₂ interface, the current transport is assumed along the x-direction and the y-axis is along the device width. The UTB FET has a translational symmetry along the width, so that we can solve 5.1 and 5.2 in the (x, z) plane. Thus, the position vector **r** refers to (x, z)coordinates. Finally, it is assumed that the wave function can be decomposed into the product of two separate functions of x and z, i.e. $\psi(x, z) = \varsigma(x)\zeta(z)$, with $\zeta(z)$ a real function. If the channel length is larger than the carrier meanfree path, electrons in the channel are expected to suffer several collisions. In this case, electron transport in the longitudinal direction can be recounted by a modified drift-diffusion model which, however, must be harmonized with the Schrödinger equation. The key to ensure this compatibility is Bohm's theory, according to which the time-dependent Schrödinger equation can be decomposed into two coupled equations with unknowns $R = |\psi|$ and $S = \hbar \arg[\psi]$ [96]. The first one is a norm conservation equation; the second one can be interpreted as the Hamilton–Jacobi equation, provided a quantum potential $Q(\mathbf{r})$ is added to the classical potential $U(\mathbf{r})$. This means that a classical description of electron transport is possible even for decananometer-size devices without resorting to the Ehrenfest theorem. Bohm's quantum potential for a conduction electron in a semiconductor crystal is

$$Q(\mathbf{r}) = -\frac{\hbar^2}{2R} \left(\nabla_{\mathbf{r}} \cdot \frac{\nabla_{\mathbf{r}} R}{\hat{m}} \right) \,. \tag{5.3}$$

If the potential is independent of x, i.e. U(x, z) = U(z), we may assume that $\zeta(x) = \exp(jk_1x)$. Under this simplifying assumption $R(x, z) = |\psi(x, z)| = \zeta(z)$. Eq. (5.3) turns out to be quite similar to the usual Schrödinger-like equation for $\zeta(z)$

$$\left(-\frac{\hbar^2}{2m_{z_v}}\nabla_z^2 + U(z)\right)\zeta_{i,v}(z) = E_{i,v}\zeta_{i,v}(z) , \qquad (5.4)$$

and becomes identical to it if it is assumed

$$Q(z)_{i,v}(z) = E_{i,v} - U(z) .$$
(5.5)

Thus, according to the previous identification, Bohm's potential as well will assume the discrete set of determinations $Q_{i,v}$ for the electrons pertaining to the i^{th} subband of the v^{th} valley. Eq. (5.5) shows that, in a stationary state, Bohm's potential represents the electron kinetic energy associated with its vibrational motion normal to the Si/SiO₂ interface. When a voltage is applied to the drain of the FET, the energy eigenvalues $E_{i,v}$, which also represent the subband edges, are not uniform anymore, i.e. $E_{i,v} = E_{i,v}(x)$. Thus, the electrons pertaining to the i^{th} subband will be subject to a driving force given by $\nabla_x E_{i,v} = \nabla_{\mathbf{r}}[U(\mathbf{r}) + Q_{i,v}(\mathbf{r})]$.

The transport equation in the i^{th} subband, suitably corrected to account for Bohm's potential in the drift-diffusion model reads

$$\mathbf{J}_{i,v} = \mu_{i,v} N_{i,v} \nabla_x E_{i,v} + q D_{i,v} \nabla_x N_{i,v} = \mu_{i,v} N_{i,v} \nabla_x E_{\mathrm{Fn},i,v} \quad (5.6)$$

where $\mathbf{J}_{i,v}$ is the current density per unit width in the i^{th} subband of the v^{th} valley; $N_{i,v} = N_{i,v}(x)$ is the electron concentration of the 2DEG per unit area in the i^{th} subband given in Eq. (1.11) and $E_{\text{Fn},i,v}$ is the electron quasi-Fermi level. Also, $\mu_{i,v}$ and $D_{i,v}$ are the electron mobility and diffusivity, respectively, in the same subband. It should be noticed that, under degenerate conditions, the Einstein relationship is modified in order to ensure that the current vanishes in equilibrium. The generalized Einstein relationship for a 2DEG becomes

$$\frac{D_{i,v}}{\mu_{i,v}} = \frac{k_B T}{q} \ln(1 + \exp(-(E_{i,v} - E_{\mathrm{Fn},i,v})/k_B T))(1 + \exp((E_{i,v} - E_{\mathrm{Fn},i,v})/k_B T)).$$
(5.7)

Finally, the continuity equations to be solved are

$$\nabla_x \cdot \mathbf{J}_{i,v} - C_{i,v} = 0 , \qquad (5.8)$$

where $C_{i,v} = C_{\text{in},i,v} - C_{\text{out},i,v}$ represents the difference between the number of electrons scattered in and out of the i^{th} subband per unit area and unit time. The total current density **J** is simply obtained by summing up the contributions $\mathbf{J}_{i,v}$ of the individual subbands, i.e.

$$\mathbf{J} = \sum_{v} \sum_{i} \mu_{i,v} N_{i,v} \nabla_x E_{i,v} + q D_{i,v} \nabla_x N_{i,v} .$$
(5.9)

Here, a QDD solver for UTB FETs on different crystal orientations is presented. The Schrödinger-Poisson tool introduced in section 1.3 is coupled with the 1D drift-diffusion solver in the longitudinal direction of the device. The solution of the Schrödinger-Poisson problem, indeed, depends on the quasi-Fermi potential profile along the x-direction, which is used for the calculation of the charge density. It is provided by the solution of the drift-diffusion equations. The procedure is iterated until an overall convergence is reached.

The effective mobility models described in the previous chapters as a function of the effective field E_{eff} and the inversion charge concentration N_{inv} have been directly incorporated in the 1D transport model, provided E_{eff} and N_{inv} are calculated in each cross-section, without transforming them into local mobility models, as required in classical simulation codes. It is assumed that the mobility is the same for each subband belonging to the same valley ($\mu_{i,v} \to \mu_v$).

The QDD solver allows to easily include the effects of mechanical stress on device performance. To this purpose, besides the mobility change, accounted for by means of the strain mobility model (see Chapter 4), the strain induced valley edge splittings in Eqs. (4.3) and (4.12), have been addressed in the Schrödinger problem, thus allowing to consider in the simulations the threshold voltage variation with strain.

In order to account for the velocity saturation effect at high longitudinal electric fields, the QDD model assumes the high-field saturation of mobility as given by the Caughey-Thomas formula [97]:

$$\mu(E_{\parallel}) = \frac{\mu_{\text{low}}}{\left[1 + \left(\frac{\mu_{\text{low}}E_{\parallel}}{v_{\text{sat}}}\right)^{\beta}\right]^{1/\beta}}$$
(5.10)

where μ_{low} denotes the low-field mobility, E_{\parallel} the electric field component along the transport direction, $v_{\text{sat}} = 10^7 \text{ cm/s}$ the saturation velocity and β a parameter ($\beta = 1.109$ and 1.213 for electrons and holes, respectively). Such approach is expected to underestimate the on-currents of short-channel devices, when the transport becomes quasi-ballistic [98]. No attempt has been made to change the mobility parameters in the high-field conditions in order to empirically correct


Figure 5.2: Inversion charge density and capacitance vs. gate voltage for two different EOTs given by the Schrödinger-Poisson solver.

the saturation velocity. Anyway, the calculated current characteristics have been analyzed so to find out the relative importance of the low-field mobility.

5.2 Analysis of SOI FETs with different surface orientations

The inversion charge and gate capacitance of MOS capacitors with different orientations are reported in Fig. 5.2 as a function of the gate voltage, showing the orientation influence for device with equivalent oxide thicknesses (EOTs) less than 2 nm. A relative difference of the capacitance for the (110) and (111)orientations with respect to the (100) one at $V_{\rm GS} = 1$ V of about 5% and 10% is found for the EOT = 1.5 and 0.5 nm, respectively.

Fig. 5.3 show the turn-on characteristics at low and high drain voltage for two different SG-SOI MOSFETs, with two gate lengths (L) and silicon thicknesses $t_{\rm Si}$, on different wafer orientations. Also indicated are the relative current variations with respect to the (100) case at $V_{\rm GS} = 1$ V. Such values should be compared with the corresponding relative variations of the low-field mobility indicated in Fig. 5.4 for an effective field corresponding to the average along the channel.

At low V_{DS} , current and mobility variations almost match, in particular for the FETs with the larger thickness. At high V_{DS} , the channel velocity partly saturates, making the current variation smaller. This is more evident in the shorter MOSFETs.

Furthermore, the effect of different surface/channel orientations on the maximum transconductance $g_{m_{MAX}} = (\partial I_{DS} / \partial V_{GS})_{MAX}$ calculated at $V_{DS} = 0.05$ V



Figure 5.3: Calculated turn-on characteristics of SG-SOI FETs on different substrates and channel directions with: (left) 14.9 nm silicon thickness, 60 nm channel length and 1 nm SiO_2 oxide thickness; (right) 7.4 nm silicon thickness, 30 nm channel length and 1 nm SiO_2 thickness.



Figure 5.4: Calculated electron mobility in SOI-MOSFETs on different substrate and channel directions versus the effective field for two silicon thicknesses.

and on the on current $I_{\rm ON}$ for two SG-SOI FETs and one DG-SOI FET with different channel lengths and EOT = 1.4 nm has been investigated. Geometrical data and biases are given in the caption of Fig. 5.5, which reports the relative variations with respect to the best performing (100) wafer. Three considerations are useful to describe the results in Fig. 5.5: i) in general, shorter gate lengths induce a smaller orientation effect, due to the velocity saturation at high longitudinal fields; ii) the series source/drain resistances tend to mask the orientation effects for the shorter gate lengths; iii) the mobility degradation with orientation is stronger for thinner $t_{\rm Si}$, as shown in Fig. 5.6, where the mobility curves as a



Figure 5.5: Relative variation of $g_{\rm mMAX}$ and $I_{\rm ON}$ with respect to the (100) case calculated for two SG-SOI FETs and one DG-SOI FET with different channel orientations. SOI60: L = 60 nm, $t_{\rm Si} = 15 \text{ nm}$. SOI30: L = 30 nm, $t_{\rm Si} = 7 \text{ nm}$. DG22: L = 22 nm, $t_{\rm Si} = 10 \text{ nm}$. All devices have 10 nm source/drain regions doped with $5.2 \times 10^{19} \text{ cm}^{-3}$. $g_{\rm mMAX}$ is calculated with $V_{\rm DS} = 0.05 \text{ V}$, $I_{\rm ON}$ with $V_{\rm GS} = 0.5 \text{ V}$ and $V_{\rm DS} = V_{\rm DD} = 0.85 \text{ V}$.



Figure 5.6: Mobility vs. effective field for different channel orientations and $t_{Si} = 15$, 10 and 7 nm. Top: (100) and (110)/(110) cases. Bottom: (100) and (111) cases.

function of the effective field for $t_{\rm Si} = 15, 10$ and 7 nm and different orientations are reported. This consideration applies mainly to the "SOI30" device.



Figure 5.7: Turn-on characteristics of a DG-FET with $L_G = 22 \text{ nm}$, $t_{\text{IL}} = 0.7 \text{ nm}$ and $t_{\text{HfO2}} = 2.9 \text{ nm}$ stack with TiN gate for different silicon thicknesses.

Table 5.1:	DIBL,	SS, I_{ON}	and I _{OFF}	of the	DG- FET	with	different	$t_{\rm Si}$
------------	-------	--------------	----------------------	--------	-----------	------	-----------	--------------

$t_{\rm Si} \ ({\rm nm})$	12	10	8
DIBL (mV/V)	91	63	41
SS (mV/dec)	78	72	67
$I_{\rm ON}~(\mu {\rm A}/\mu {\rm m})$	473	406	342
$I_{\rm OFF} ({\rm nA}/\mu{\rm m})$	0.033	0.0019	1.04×10^{-4}

5.3 Analysis of DG-FETs with different silicon thicknesses

The effect of different $t_{\rm Si}$ on the 22nm DG-FET by using a SiO₂/HfO₂ stack with $t_{\rm IL} = 0.7$ nm, $t_{\rm HfO_2} = 2.9$ nm ($\kappa = 16$) and TiN gate (workfunction is 4.8 eV) has been analyzed. The use of TiN instead of Polysilicon allows a low channel doping of 1.2×10^{15} cm⁻³ (the threshold voltage is set by the TiN workfunction and high channel doping is not needed), with an implicit advantage in the low-field mobility. The turn-on characteristics with different $t_{\rm Si}$ are shown in Fig. 5.7. The corresponding significant figures of merit to evaluate the device performance are also collected in table 5.1, i.e. the drain-induced barrier lowering (DIBL), the inverse subthreshold slope (SS), the on-current ($I_{\rm ON}$) and the off-current $I_{\rm OFF}$. As $V_{\rm DS}$ increases, a decrease of the source(or drain)-to-channel energy barrier is observed in short-channel MOSFETs, which is responsible of the large $I_{\rm OFF}$ at large $V_{\rm DS}$. This short-channel effect (SCE) can be evaluated with the DIBL, which is the difference of the threshold voltage at low $V_{\rm DS}$ and the one at large



Figure 5.8: Mobility (top) and mobility enhancement (bottom) as a function of the effective field as predicted by the model for uniaxial tensile $\langle 110 \rangle$ strain of 0.1, 0.2, 0.35 and 0.5%. The predicted mobility enhancement nicely reproduces the experimental findings in [6] by using the high-strain wafer bending technique.

 $V_{\rm DS}$. The inverse subthreshold slope is extracted as $SS = \partial V_{\rm GS} / \partial \log(I_{\rm DS})$.

When reducing $t_{\rm Si}$, on one side a better electrostatic control of the channel allows to reduce SCEs, i.e. DIBL, SS and $I_{\rm OFF}$, on the other side electron mobility is degraded and therefore the I_{ON} decreases as well. The device with $t_{\rm Si} = 10$ nm represents a reasonable compromise between $I_{\rm ON}$ and $I_{\rm OFF}$.

5.4 Analysis of DG-FETs under (110) uniaxial stress

Mobility curves and mobility enhancement factors are shown in Fig. 5.8 for different strain conditions, where the parameters of the "DG22" device with HfO₂ and TiN are considered. Fig. 5.9 shows the $g_{\rm mMAX}$ and $I_{\rm ON}$ enhancements of "DG22" versus the mobility enhancement extracted from Fig. 5.8 at an effective field of 10⁵ V/cm. The transconductance is calculated for $V_{\rm DS} = 0.05$ V, while $I_{\rm ON}$ is calculated for $V_{\rm GS} = V_{\rm DS} = V_{\rm DD} = 0.85$ V. A linear fit gives a correlation factor of 0.5 and 0.2 for the $g_{\rm mMAX}$ and $I_{\rm ON}$ enhancements, respectively. It is worth noting, however, that the QDD model intrinsically tends to underestimate the strain-induced enhancement, especially at high $V_{\rm DS}$ and $(I_{\rm ON})$, because of the velocity saturation in short devices (see, e.g., [103]). Some data deduced from published experimental results are also reported in Fig. 5.9 for comparison: the qualitative agreement is satisfactory also taking into account that the experimental data correspond to devices with geometrical parameters different from the "DG22" (i.e., partially- and fully-depleted strained-SOI (PD- FD-sSOI) and strained-Si on SiGe (sSi/SiGe)).



Figure 5.9: Relative variation of $g_{\rm mMAX}$ and $I_{\rm ON}$ with respect to the mobility enhancement for the "DG22" ($t_{\rm IL} = 0.7 \text{ nm}, t_{\rm HfO2} = 2.9 \text{ nm}, \text{ TiN gate}$) (circles). Dashed lines: linear regression curves with slopes 0.5 (top) and 0.2 (bottom). Other symbols: data extracted from experiments on devices with gate lengths ranging from 25 to 75 nm available in literature ([99], [100], [101] and [102]).

5.5 Summary

The carrier mobility models presented in the previous chapters have been incorporated into a 1D quantum drift-diffusion (QDD) simulation tool featuring quantum-confinement effects in SG- and DG-SOI FETs in order to investigate the device performance of future technology nodes. Short (L = 22 nm) DG-FETs, where mobility is modified by quantum-confinement effects, ultrathin SiO₂/HfO₂ gate stacks and metal gate, have been investigated. Finally, the correlations between the mobility enhancement induced by uniaxial stress in a 22nm DG-FET and the transconductance and on-current have been calculated.

Part II

SILICON NANOWIRES FOR GAS DETECTION

Chapter 6

TCAD analysis of gas sensors based on SiNWs

Nanowire (NW) structure has been demonstrated to be an excellent candidate for chemical sensors because of the enhanced sensitivity that derives mainly from small size and very high surface-to-volume ratio [104]. The presence of a few chemically-active charged molecules on the NW surface can modulate the carrier distribution over their entire cross-sectional conductive pathway through the surface charging induced by chemical reaction, thus operating an electrical transduction of the chemical reaction itself. This can result in an increased sensitivity as compared to traditional CMOS-based sensors. Although such structures can be made with a broad array of materials (metal oxides, such as ZnO [105], In_2O_3 [106], conducting polymers [107] and carbon nanotubes [108]) and a significant progress has been recently made in the use of non-traditional fabrication approaches and new materials [104], silicon-based nanowires (SiNWs) have still received special attention given the sophisticated state of silicon fabrication technology for large scale integration. Silicon-based sensors can be readily integrated with CMOS circuits that can be used for signal processing and analysis, thus providing fast and low-cost detection. Moreover, SiNWs can be manufactured using top-down approach [109] [110] [111], which makes the prospect of sensor arrays realistic and the exposed surface can be also modified to act as both electron transfer mediator and immobilizing matrix for biological or chemical molecules [112]. A challenging issue is the processing of the chemical functionalization which is commonly used to enhance the sensitivity, but is always critical for the sensor performance [113] [114].

In addition, despite the important technological advances in silicon-based nanosensors, their design principles are not still well elaborated, and the path to further optimization has to be defined. Although SiNWs have been already experimentally demonstrated for high-sensitive detection of gases [115] [116] [117], DNA [118] [119], pH levels and proteins [120], a detailed analysis of the physical mechanisms playing a role in the detection phase is still missing.

In this thesis, the optimization of a SiNW gas sensor for the detection of chemical warfare agents has been investigated. The organo-phosphorus compounds (OPs) represent one of the most important and lethal classes of chemical warfare agents. The ease of manufacturing OPs based on inexpensive starting materials makes these agents a potential weapon for terrorist attacks [121]. Thus, the rapid sensing of these nerve agents has become an increasingly important research goal. The use of chemically-fuctionalized silicon nanoribbons (SiNRs) as OPs detectors has been already demonstrated [114]. In particular, the silicon surface has been grafted with 3-(4-ethynylbenzyl)-1, 5, 7-Trimethyl-3-AzaBIcyclo [3.3.1] Nonane-7-methanOL (TABINOL) and exposed to diphenylchlorophosphate (DPCP), used as a simulant of nerve agents due to its similar structure and chemical reactivity, but much lower toxicity. Highly sensitive and selective detection of OPs has been experimentally obtained. The SiNR sensors operated at sub-ppm level of gas pressure and showed a very fast and marked response.

In a research activity carried out in strict collaboration with the Institute of Information and Communication Technologies, Electronics and Applied Mathematics (ICTEAM), Université Chatolique de Louvain (UCL), Louvain-la-Neuve, Belgium, the Institute of Electronics Microelectronics and Nanotechnology (IEMN), Villeneuve d'Ascq, France, and the CEA-Grenoble, LITEN/DTNM/LCRE, Grenoble, France, arrays of p-doped parallel NWs fabricated on -insulator (SOI) substrate and chemically-functionalized with TABINOL are shown to improve NRbased sensors performance, providing a higher sensitivity to vapors of DPCP. Although it is generally accepted from experimental results that SiNWs with reduced width provide better sensitivity, a detailed physical analysis of the transduction mechanism, and the study of the semiconductor surface and interface properties is still lacking. In this thesis, an extensive TCAD simulation study of the nanosensors has been performed, aimed at a thorough analysis of the physical behavior and at a systematic optimization of the device. An insight on the physical mechanisms involved in the gas detection has been gained and the impact of surface-to-volume ratio on sensor sensitivity has been quantitatively determined showing a strong increase in sensitivity for widths below 50 nm.

In section 6 the details of the fabrication process of the NR/NW sensors are reported, together with the description of the measurement procedures adopted for their electrical characterization. Moreover, the experimental results for NRand NW-based sensors are described in detail. The analysis of the interface trap formation at the Si/SiO₂ and Si/molecular-layer interfaces of the nanosensors before gas exposure is addressed in sections 6.2 and 6.3, respectively. The chemical



Figure 6.1: Process steps for top-down fabrication of the SiNWs and SiNRs.

and physical mechanisms involved in the gas detection phase are pointed out in section 6.4, together with the TCAD analysis of the nanosensors after gas exposure. Finally, the TCAD analysis of the sensor sensitivity and the role played by the NW dimensions are reported in section 6.5. Device fabrication process and sensor characterization The complete process steps are illustrated in Fig. 6.1. The starting material is a p-type silicon-on-insulator (SOI) substrate with top silicon thickness $(t_{\rm Si})$ of 50 nm, buried oxide thickness $(t_{\rm BOX})$ of 145 nm and bulk Si substrate thickness of approximately 700 μ m. The processing is started by a cleaning step in piranha mixture (sulphuric acid and hydrogen peroxide 1:3) for 10 mins followed by a deionized (DI) water rinse for 10 mins. Then, the substrate is dipped in hydrofluoric acid (HF) of 1% concentration for 20 s to remove the chemical oxide followed by DI water rinse. First, alignment marks are exposed with EL-13% positive tone electron beam resist. After exposure with VISTEC EBPG 5000+ beam writer at 50 keV, the wafer is developed in mixture of propanol (IPA 60 ml) and methylisobutylketone (MIBK 30 ml) for 1 min and then rinsed in IPA for 30 s. After development, the marks are etched with SF6/N2 chemistry to etch the top silicon, CF4/Ar chemistry to etch the buried oxide, followed by SF6/N2 chemistry to etch in the bulk Si substrate. Resist is



Figure 6.2: Top-view SEM image of SiNWs (a) after resist development (b) after silicon etching and resist removal (c) after exposure of PMMA to make openings for metal deposition (d) after lift-off of platinum on the source and drain pads.

removed by exposure under UV light followed by acetone, propanol cleaning and drying by nitrogen. Negative tone electron beam resist hydrogen silsesquioxane (FOx-16/HSQ) is spin coated and exposed. Implantation of source/drain region is done with boron at 20 keV, with a dose of 5×10^{15} cm⁻². The regions where HSQ is exposed are protected from implantation thanks to the physical barrier provided by the resist. After implantation, resist is removed using HF-1% dip for 2 mins followed by DI water rinse and nitrogen blow drying. Activation of impurities is done by subjecting the wafer to 950° C for 2 mins in nitrogen atmosphere. After standard cleaning using piranha mixture for 10 mins followed by DI rinse, FOx-12 is spin coated on the substrate to give a thickness of 50 nm. The resist is exposed (with proximity corrections included) with the pattern consisting of NWs and contact pads. After development in tetramethylammonium hydroxide (TMAH) 25% for 1 min the wafer is rinsed in DI water (Fig. 6.2a). Then the top-silicon is defined with chlorine based chemistry (Fig. 6.2b). An over etch time of 5 s is considered to assure the complete removal of the top-silicon layer. Ellipsometer measurement is done to measure the buried oxide. The resist is removed by immersing the wafer in HF 1% for 1 min followed by DI water rinse. Then, PMMA 2% 950k is spin coated on the wafer to give a thickness of 200 nm. Exposure of patterns to make openings for the contacts is done (Fig. 6.2c), followed by the development and HF1% to remove native oxide. Platinum is deposited to a thickness of 20 nm. Lift-off is performed by immersing the wafer in acetone together with the use of ultrasonic agitation to avoid sidewall flakes of metal. Back side metal deposition is performed by protecting the front side using a resist. After inspecting the wafer with scanning electron microscope (SEM)



Figure 6.3: Measured $I_D - V_{BG}$ characteristics of SiNR before and after DPCP exposure for 1 hour.

(Fig. 6.2d), the front-side resist is removed and the wafer is annealed at 400° C for 2 mins in nitrogen atmosphere. The contacts are of platinum silicide.

One NR resistor (shown as the inset of Fig. 6.2d) has been fabricated with a width W = 1 μ m and a length L = 5 μ m, while a set of array of 10 parallel NW resistors has been realized with a NW spacing of 0.3 μ m, different W (25, 50, 75, and 100 nm) and L (0.3, 0.6, 1.2 and 2.4 μ m). Among the available architectures, the NR and the array of NWs with W = 25 nm and L = 0.3 μ m have been processed to realize the sensor. The devices have been functionalized via covalent grafting through thermal hydrosilylation of TABINOL onto HFpenetrated substrate in refluxing mesitylene for 2 hours.

The electrical characterization has been carried out with an Agilent 4155C semiconductor parameter analyzer. DC drain current versus back-gate voltage $(I_D - V_{BG})$ characteristics have been measured to check the sensor functionality. The functionalized nanosensors were fully characterized before and after 1 h exposure to DPCP vapors. In a typical experiment, two drops of DPCP were deposited in a 50 mL closed Petri dish to generate a stabilized vapor pressure of DPCP, the chip was then introduced carefully so that it was not in direct contact with the DPCP but only exposed to the vapor for one hour. DPCP vapor pressure was estimated to be in the 500-800 ppb range.

 I_D versus time measurement has been also performed. In order to avoid the issues of charge-trapping effects or Joule effect, I_D is acquired in pulsed mode, with $V_{DS} = -2$ V and $V_{BG} = 2$ V pulses maintained for 50 ms to allow the acquisition of I_D , and then followed by a recovery time of 10 s with $V_{DS} = V_{BG} = 0$ V.

Fig. 6.3 shows the measured $I_D - V_{BG}$ characteristics of the NR resistor



Figure 6.4: Measured $I_D - V_{BG}$ characteristics of an array of 10 parallel NWs with W = 25 nm before and after DPCP exposure for 1 hour.

before and after DPCP exposure. The device response to vapors of DPCP is very marked. Both an enhancement of the current level and a shift of the transcharacteristics towards more positive V_{BG} values are observed. They can be quantitatively analyzed by extracting I_0/I_0 and ΔV_0 , where I_0 and V_0 are the drain current and the back-gate voltage at the minimum current level before exposure (see Fig. 6.3). The NR sensor showed $\Delta I_0/I_0 \approx 1.6 \times 10^6$ and $\Delta V_0 \approx 7$ V, which are in agreement with [114]. The experimental transfer curves before and after DPCP exposure of the NW-based sensors are presented in Fig. 6.4. The turn-on characteristics of the NR-based sensor described above are also shown in dashed lines for comparison. The NW-based device clearly shows both an higher $\Delta I_0/I_0$ and a wider ΔV_0 with respect to the NR after DPCP exposure. In particular, it is $\Delta I_0/I_0 \approx 4 \times 10^6$ and $\Delta V_0 \approx 13$ V.

In Fig. 6.5 the time response of I_D for NWs when vapors of DPCP in the 500-800 ppb concentration range are injected in the measurement chamber. The detection of the gas rapidly changes the current level with a very high amplification factor of more than five orders of magnitude. In about 100 s, a plateau is reached corresponding to the saturation of the chemical reactions.

6.1 Definition of the simulation decks

Numerical simulations of the NW/NR-based devices have been carried out within the framework of the Synopsys simulation tools [26]. Poisson's equation coupled with the drift-diffusion transport model has been solved to obtain the $I_D - V_{BG}$ characteristics. The longitudinal dimensions of the devices are indeed large enough to consider the classical drift-diffusion model sufficiently accurate in de-



Figure 6.5: I_D measured in pulsed mode as a function of time. DPCP vapors were introduced at t = 400 s.

scribing the carrier transport. In addition, the transversal dimensions considered here (greater than 25 nm), are large enough to neglect quantum confinement effects.

The resistors under investigation are intentionally p-doped and with ohmic contacts (source and drain regions are p-doped with a concentration of 6×10^{19} cm^{-3}). The channel doping concentration is a critical parameter in these simulations, since the impact of interface properties (fixed and trapped charges at the interface) on carrier transport is strongly influenced by the doping concentration (especially in depletion and low accumulation regions). Unfortunately, slight variations with respect to the nominal value are expected. In order to provide a doping concentration as realistic as possible to the simulation deck, it has been extracted from the measured resistance of the wire at $V_{DS} \approx 0$ (where the $I_D - V_{DS}$ characteristic is quite linear), and a value of about 4×10^{16} cm⁻³ has been extracted and adopted in the simulation. In Fig. 6.6, the simulation decks for the NR and NW sensors are shown. For the sake of simplicity, it has been assumed that the NR has a translational symmetry along the width, due to its large lateral dimensions. So that a 2D structure can be considered (see Fig. 6.6, left). On the other hand a 3D deck is necessary for the NW sensor, due to its reduced lateral dimensions (the minimum width considered here is 25 nm), to take into account the effects of sidewalls in the simulations. Since the NWs are organized in arrays of 10 parallel wires with spacing of 0.3 μ m, a basic cell composed of half wire and half spacing region has been defined and adopted in the simulations (see Fig. 6.6, right). The resulting I_D is then multiplied by 20. In the simulation analysis carried out, the Si/SiO_2 and the Si/molecular layer interfaces are treated separately.



Figure 6.6: (Left) Two-dimensional and three-dimensional simulation decks for the NR (left) and NW (right) sensors, respectively. The brown region represents the buried oxides and the green one the functionalized molecular layer; Cyan lines delimit the electrodes; the p-type doping concentration is plotted in the silicon regions.

6.2 Analysis of the Si/SiO₂ trap states in top-down SiNWs

Since the gas detection occurs via modifications of the physical properties of the interface between Si and the molecular layer, it is very important to accurately model the device interfaces in order to perform numerical simulations of the sensor behavior. Consequently, a combined characterization and simulation study is presented to investigate the impact of interface trap states on I_D and to separate the drain current modulation due to the presence of trap states at the Si/BOX interface from the one due to trap states at the Si/molecular layer interface.

Two types of defects dominate the electronic properties of the Si/SiO₂ interface, namely i) the stretched bonds and ii) the silicon dangling bonds with different residual (back) bond configurations. Differently from the sharp energetic levels of defects in bulk silicon, the interface defects show broad energetic distributions. The broadening can be interpreted as due to the statistical disorder in the bond angle and in the distance to the neighbour atoms of the defects themselves [122]. These two kinds of interfacial defects are called intrinsic because they can be treated as eigen-defects of the silicon lattice. Stretched silicon-silicon bonds represent small distortions of the silicon lattice. The related states split from the bands into the gap and show Boltzmann-like distributions (U_T) in the neighbourhood of the band edges (Fig. 6.7). If the distortion becomes larger a bond breaking may be favorable, creating a dangling bond center. The silicon dangling bond center with three silicon back bonds shows a broad double peaked



Figure 6.7: Schematic representation of trap states in the Si bandgap due to defects at the Si/SiO₂ interface. Black lines denote intrinsic defects (U_T and U_M), red lines indicate extrinsic defects (P_L and P_H). $E_i = E_g/2$.

distribution (U_M) modulated over the whole gap with some distance from the band edges, as shown in Fig. 6.7. Two relative peaks can be seen, the first peak lies at $E_V + 0.25$ -0.35 eV and the second at $E_V + 0.7$ -0.85 eV [123], which can be related to the nature of the intrinsic defect itself as explained below. Such defects correspond, to a large extent, to the so-called P_b centers. More specifically, they are denoted as P_{b0} and P_b at the Si(100)/SiO₂ and Si(111)/SiO₂ interface, respectively. A typical feature of the intrinsic defects is the symmetry of the related distributions in the silicon bandgap caused by their amphoteric character. Each defect gives rise to two corresponding levels: one bonding (donor-like) state in the lower half and one antibonding (acceptor-like) state in the upper half of the bandgap [122]. Acceptor traps are filled with one electron, thus negatively charged, if fully occupied, i.e. if the Fermi level is above the trap level, while they are neutral if unoccupied. On the contrary, donor traps are neutral if occupied, i.e. if the Fermi level is above the trap level, while positively charged (one electron is released) when unoccupied. Unlike doping impurities indeed, donor energy levels are typically located in the lower part of the bandgap and acceptor energy levels are in the upper part of the bandgap [124] [125]. The U-shaped trap distribution, typical for the Si/SiO_2 system, is given by the superposition of both intrinsic defect types U_T and U_M .

The extrinsic traps are dangling bond defects with oxygen back bonds and show only donor-like states inside the silicon bandgap. The distributions are broad peaks in the lower (P_L) and upper (P_H) half of the gap, respectively, as shown in Fig. 6.7. The corresponding acceptor-like states are assumed to be inside the silicon conduction band [122].

Due to the complexity of possible defect configurations in a Si/SiO_2 interface, a thorough methodology based on theoretical assumptions and experimental results, which allows for the extraction of the trap configuration and energetic distribution, is needed. To this purpose, the trap distribution at the Si/BOX interface has been extracted by using the Terman apprach [126]. It allows for the extraction of the energy density of the interface trap concentration at the Si/SiO_2 interfaces as illustrated in [127] [128]. The Terman method relies on a high-frequency capacitance-voltage (C - V) measurement at a frequency sufficiently high so that the interface traps are assumed not to respond. By this way, they should not contribute to the C-V characteristics. Although interface traps do not respond to the ac probe frequency, they respond to the slowly varying DC gate voltage and, in this case, cause the high-frequency C - V curve to stretch out along the gate voltage axis as the interface trap occupancy changes with gate bias. In other words, for a MOS capacitance in depletion or inversion additional charge placed on the gate induces additional semiconductor charge Q_{it} , so that the total semiconductor charge results $Q_G = -(Q_b + Q_n + Q_{it})$, where Q_b is the space-charge region bulk charge density and Q_n the electron charge density. The gate voltage results

$$V_G = V_{FB} + \phi_s + V_{\text{ox}} = V_{FB} + \phi_s + Q_G / C_{\text{ox}}, \qquad (6.1)$$

where V_{FB} is the flat-band voltage and $V_{\rm ox}$ and $C_{\rm ox}$ are the oxide voltage and capacitance, respectively. Therefore it is obvious that for a given surface potential ϕ_s , V_G varies when interface traps are present, leading to the C - V stretchout. The stretch-out produces a non parallel shift of the C - V curve. Interface traps distributed uniformly through the semiconductor band gap produce a fairly smoothly varying but distorted C - V curve. Interface traps with distinct structure, for example peaked distributions, produce more abrupt distortions in the C - V curve. The energy distribution of trap states can be calculated as

$$D_{it}(E) = \frac{C_{\text{ox}}}{q} \left(\frac{dV_G^{\text{exp}}}{d\phi_s} - \frac{dV_G^{\text{ideal}}}{d\phi_s} \right) , \qquad (6.2)$$

where V_G^{\exp} and V_G^{ideal} refer to the gate voltage obtained by experiments and the ideal one in the absence of traps, respectively. Similarly, $D_{it}(E)$, can be extracted from the difference in derivative in the I-V turn-on characteristics of the device by using the so-called "current-Terman" method [127] as

$$D_{it}(E) = \frac{C_{\text{ox}}}{q} \left(\frac{dV_G^{\text{exp}}}{d\phi_s} - \frac{dV_G^{\text{sim}}}{d\phi_s} \right) = \frac{C_{\text{ox}}}{q} \frac{dI_D}{d\phi_s} \left(\frac{dV_G^{\text{exp}}}{dI_D} - \frac{dV_G^{\text{sim}}}{dI_D} \right)$$
(6.3)

where the first term is obtained from the experimental I - V characteristics and the second one is obtained by TCAD device simulations without any trap state. The derivative of I_D with respect to ϕ_s is obtained by the numerical results as well. Moreover, ϕ_s can be directly related to the quasi-Fermi energy in the band gap to obtain $D_{it}(E)$.



Figure 6.8: Energy distribution in the bandgap of trap state density at the Si/BOX interface extracted with the current-Terman method (symbols) and used as input of the simulations (red solid line). The signature of intrinsic defects is highlighted. $E_i = E_g/2$.

The Terman method was one of the first methods for determining the interface trap density and it has been widely criticized in the literature, especially when applied to thin oxides [125]. Moreover, in order to compare experimental with TCAD curves, one needs to accurately describe the doping density local distribution and the device geometry. Despite its questionable accuracy, the method could be useful in our case to extract the trap density at Si/BOX interface of the NR architecture. In fact, the BOX oxide is thick enough and the device structure is simple enough to allow the application of the method with reasonable precision. In addition, the current-Terman method allows to separate the contribution of interface traps at the Si/BOX with that at the Si/molecular layer. Since the surface potential at the Si/molecular layer only slightly varies with V_{BG} , the trapped charge at that interface is almost fixed as well. Therefore, unlike the traps at the Si/BOX interface, the ones at the Si/molecules interface mostly contribute in shifting instead of stretching out the $I_D - V_{BG}$ characteristics and, thus, they do not contribute appreciably to the extracted $D_{it}(E)$.

In Fig. 6.8, $D_{it}(E)$ at the Si/BOX interface is shown, together with the approximated curve adopted in the TCAD simulations (red line). The extracted $D_{it}(E)$ clearly shows the signature of the intrinsic defects: i) a Boltzmann-like distributions (U_T) in the neighbourhood of $E_{\rm V}$ and ii) a broad peak at $E_{\rm V}$ + 0.25-0.35 eV (U_M). By applying $V_{BG} < 0$ to p-type devices, the associated band bending makes the Fermi level to scan the lower part of the bandgap. Accordingly, trap density can be extracted from this bandgap region only. Moreover, this implies that acceptor states, which always lie above the mid gap, are not filled



Figure 6.9: Drain current versus back-gate voltage. Symbols: measurements; lines: simulation results of the ideal case without any trap states (dashed line), considering only traps at the Si/BOX interface (dot-dashed line), with traps both at the Si/BOX and Si/molecular layer interfaces (dotted line) and including the Coulomb scattering contribution due to the interaction with trapped charges (solid line).

and do not contribute to the current modulation.

The donor-like trap state distribution extracted with the current-Terman method has been modeled with the exponential function

$$D_{it}(E) = 3.3 \times 10^{12} \exp\left(-\left|\frac{(E - E_g/2) + E_1}{E_2}\right|\right)$$
(6.4)

with $E_q/2$ the mid gap energy, $E_1 = 0.465$ eV and $E_2 = 0.16$ eV. The model has been used at the Si/BOX interface to simulate the NR device. The TCAD results are reported in Fig. 6.9, showing that the role played by the Si/BOX interface traps is limited to a small modulation of the I - V curve which shows an overall degradation of the current. It is worth pointing out that the mobility degradation due to the coulombic interactions of the interface trapped charges with the carriers in the channel is accounted for consistently. The formulation of the mobility limited by interface charges available in the D-2010.03 release of Sentaurus Device [26] has been used in the simulations. In p-type silicon indeed, the Fermi level lies below the mid gap, thus at $V_{BG} = 0$ some positive trapped charges are present at the interface, given by unoccupied donor states. When V_{BG} moves towards negative values the Fermi level shifts down in the band gap near E_V and more donor states are unfilled, resulting in more net positive charge (Fig. 6.10). Therefore, both threshold shift and change in slope of the trans-characteristic are observed. However, the impact of the traps at the Si/BOX interface is not sufficient to reproduce the experimental data. Hence,



Figure 6.10: Fermi level at the Si/BOX interface and interface trapped holes as a function of back-gate voltage.

the remaining gap between the simulated and the experimental curves can be ascribed to the trap states at the Si/molecular layer interface.

6.3 Analysis of the molecular layer interface before exposure

The chemically functionalized organic monolayer can be non uniform and leave some uncovered patches that can react with the air. Also the steric volume of the considered molecule can lead to a partial coverage of the silicon surface. Therefore, it makes sense to consider that the Si/monolayer interface is affected by traps originating from the $Si/native SiO_x$ system. Angermann et al. [129], monitored the evolution of $D_{it}(E)$ curves during the oxidation in air at room temperature on a previously H-terminated surface, by means of a pulsed fieldmodulated surface photovoltage (SPV) measurements. In order to determine the interface state density $D_{it}(E)$, a varying electric field perpendicular to the surface was applied, which changes the surface potential continuously as a function of the field voltage. Initially, the prepared H-terminated surface is dominated by intrinsic defects, with the characteristic U-shaped distribution, as shown in Fig. 6.11, curve A. An increasing density of these states and additional appearance of extrinsic states P_L was observed after 48 h when the hydrogen coverage decreases (curve B). The occurrence of oxygen backbonded dangling bond defects indicates that the surface is no longer hydrogen-terminated. During further native oxide growth, the P_H group additionally appears (curve C). The superposition of intrinsic and extrinsic states can be approximated with a uniform donor-type D_{it}



Figure 6.11: Interface state distribution $D_{it}(E)$. Symbols: measurements from [129] for a H-terminated surface (curve A), after 48 h (curve B) and 21 days (curve C) storage time in air. Red solid line: model adopted for the Si/molecular layer interface.

level in the bandgap (red solid line) ranging from E_i -0.22 eV to E_i +0.15 eV. On the contrary trap states close to the conduction band edge are acceptors. Fig. 6.9 shows that considering the uniform trap distribution above with $D_{it} = 3 \times 10^{12}$ cm⁻²eV⁻¹, a very good fitting between simulations and experiments is obtained.

6.4 Cooperative effect of molecules on surfaces upon gas exposure

In order to carry out an accurate numerical investigation on gas sensors, the surface transduction mechanism has to be studied in detail and accurately modeled. When TABINOL molecules, grafted on the silicon surface, are exposed to DPCP, aza-adamantane quanternary ammonium salt is formed [114], as shown in Fig. 6.12. It is worth noting that, upon reaction with DPCP, a positive charge is present on the hydrogen atom at a certain distance from the silicon surface, estimated to be of about 0.7-0.8 nm (see blue dots in Fig. 6.13). The modification of the drain current in a molecularly functionalized device can thus be mainly ascribed to many effects, namely: i) a change in the density of silicon surface states [130] [131]; ii) a charge transfer between silicon and molecules, leading to interface dipoles [132]. Since the reaction of DPCP with TABINOL does not change the nature and density of grafting links between TABINOL and silicon, surface states should not be modified during the gas exposure. Therefore, a charge transfer is considered the main effect playing a role [114]. In fact, if 6.4. COOPERATIVE EFFECT OF MOLECULES ON SURFACES UPON GAS EXPOSURE



Figure 6.12: Chemical detection scheme. TABINOL (left) evolves to azaadamantane quanternary ammonium salt (right) upon exposure to OPs simulant (DPCP). Courtesy of CEA-Grenoble, LITEN/DTNM/LCRE, Grenoble, France.



Figure 6.13: Schematic view of the gas nanosensor with source (S), drain (D) and back-gate (G) contacts. Courtesy of CEA-Grenoble, LITEN/DTNM/LCRE, Grenoble, France.

the surface molecules are well-oriented and close-packed, cooperative effects can induce a charge redistribution [132]. By assuming the molecular layer as a dipole, with molecules organized parallel to each other and oriented perpendicular to the substrate, the electrostatic potential difference across the dipole layer reads

$$\Delta \Phi = \left(\mu_{\perp}\rho\right)/\epsilon\,,\tag{6.5}$$

where μ_{\perp} is the component of the molecular dipole perpendicular to the surface, ρ the surface density of the absorbed molecules and ϵ the dielectric constant. For molecular monolayers $\Delta \Phi$ is typically high enough to induce a very high electric field. In our case, the average molecular density $\rho \approx 2.8 \times 10^{14} \text{ cm}^{-2}$, the dielectric constant $\epsilon \approx 2 - 2.5$ and the thickness of the dipole layer $d \approx 0.7 - 0.8$ nm [114]. By considering a typical dipole moment $\mu_{\perp} \approx 1$ Debye (1 Debye = $3.34 \times 10^{-30} \text{ C m}$) [132], the electric field within the molecular layer is $E \approx 7 \times 10^6$ V/cm. Hence, even with small values of μ_{\perp} , the potential across a close-packed molecular dipole layer leads to electric fields well exceeding the breakdown value. In such conditions, the molecular layer/substrate system will be induced by the

107



Figure 6.14: Schematic of the molecularly functionalized silicon device. The molecules are represented as dipoles A) before charge rearrangement and B) after charge rearrangement, which reduces the electric field within the molecular layer.

high field to a charge redistribution so as to decrease the field, leading to a reduction of $\Delta \Phi$ in equation (6.5). The dipolar property of the molecules can lead to intramolecular charge reorganization, which induces a dipole moment reduction. This mechanism is, however, negligible in organized organic layers made up of molecules with low polarizzability, i.e., low ϵ , as in our case. If such molecules are close-packed in a well-ordered layer, then the energetically most favorable way for reducing the field across the layer is a charge transfer to or from the silicon substrate. This amount of charge usually results in more than 10^{12} electrons or holes per cm² [132] [114]. A schematic view of the charge redistribution phenomenon which takes place after gas exposure is shown in Fig. 6.14. The positive charges resulting from the reaction of TABINOL with DPCP cause the formation of negative charges at the interface between the silicon and the molecular layer, which attract holes and modify the silicon surface potential.

In Fig. 6.15, the simulation result is compared with the experimental curve after gas exposure, showing a good agreement. A concentration of 1.93×10^{12} cm⁻² of transferred electrons has been considered in order to reproduce experiments, in line with values in literature [132] [114]. The coulombic interaction of charge carriers with the fixed negative charge at the interface has been accounted for in the mobility model as well.

6.5 Effect of surface-to-volume ratio on sensitivity

A further investigation has been carried out in order to study the effect of surfaceto-volume ratio on sensitivity. To this purpose 3D simulations of NWs with different widths have been performed. The distribution of trap states at the Si/BOX and Si/molecular-layer interfaces and the negative fixed-charge concentration after gas exposure have been fixed to the values extracted against experiments as explained in the previous section. In Fig. 6.16, the sensitivity in terms of



Figure 6.15: Drain current versus back-gate voltage of the SiNR sensor before and after exposure to DPCP. Simulation results (solid lines) are compared with experimental curves (symbols).



Figure 6.16: Maximum drain current variation and back-gate voltage shift after DPCP exposure as a function of NW width.

both the shift of the current curves towards positive V_{BG} (ΔV_0) and the maximum drain current enhancement (($\Delta I_0/I_0$)_{max}) is shown as a function of the NW width. While in [114] V₀ is defined as the value of V_{BG} corresponding to the minimum current in curves with a marked ambipolarity, here the minumum value of I_D is not always well defined, apart from the experimental curves relative to the array of NWs reported in Fig. 6.4. Therefore V₀ is extracted by imposing $I_{\rm ON}/I_{\rm OFF} = 6 \times 10^6$, where $I_{\rm ON}$ is taken at $V_{BG} = -20$ V. Although the slight



Figure 6.17: Hole density distribution along with the 3D simulated structure and the device cross-section of the NWs with W = 50 nm (top) and W = 25 nm (bottom) at $V_D = -2$ V and $V_{BG} = 9$ V. The while line delimits the depletion region. In the 3D pictures the molecular layer which covers the SiNW is not shown.

mismatch between simulation and experimental results, likely due to variations of the silicon surface and interface characteristics, Fig. 6.16 clearly shows that the sensor performance can be improved by reducing the NW width, especially when W is shrunk below 50 nm. While $(\Delta I_0/I_0)_{max}$ increases almost linearly when W decreases from 1 μ m to 25 nm, ΔV_0 is constant at about 7 V from 1 μ m to about 50 nm and exponentially increases when W is below that value. This effect can be ascribed to the impact of negative fixed charges at the NW sidewalls on device electrostatics. In Fig. 6.17 left, the hole density distribution within two NWs of W = 25 and 50 nm is plotted at $V_{BG} = 9$ V. While wider NW is completely depleted, thus giving no contribution to the transport, the narrower one is still conductive, thanks to the negative charges at the interfaces, which induce a conductive path in the upper half of the wire. The different transport regime is clearly visible in the 2D views of the device cross-sections in Fig. 6.17, right.

6.6 Summary

Extensive TCAD simulation analysis combined with device interface study and characterization have been performed to better understand the physical mechanisms playing a role in the gas detection of SiNW/SiNR sensors. The interaction of carriers with interface trap states and fixed charges is addressed to correctly reproduce experimental responses of recently fabricated gas nanosensors. The impact of surface-to-volume ratio on sensor sensitivity is investigated showing 6.6. SUMMARY

enhanced performance for narrower wires.

Chapter 7

Conclusions

In this thesis, the development of physically-based mobility models for electrons and holes in ultrathin body SOI FETs have been presented. The resulting models are analytical function of the effective (or local) electric field, substrate doping concentration, silicon thickness, surface orientation and channel direction. The model has been extended to include the effects induced by gate stacks composed of high- κ oxide and metal gate, as well as the impact of mechanical stress on valley splitting, valley repopulation, band warping and scattering rates on (100) and (110) crystallographic orientations. The developed analytical formulations for every single modeled effect have been combined in unified mobility models. They have been demonstrated to correctly account for the simultaneous and complex combination of several effects by correctly predicting experimental mobility in vertical multi-gate FETs with either SiO₂/polysilicon and high- κ /metal gate stacks and different stress configurations, thus proving to be a useful TCAD tool for device optimization of future technology nodes.

Even if based on strong approximations and very far from a rigorous description of the physical phenomena, the developed models are attractive for their capability of keeping tracks of a large number of different physical effects, allowing quantitatively accurate predictions and very low computational costs. However, the development process of such models is non trivial, despite of the resulting simple analytical formulations. Numerical solvers and a wide set of experiments are compulsory to support the analytical model calibration, and not always available. Moreover, it is worth noting that the success in the development of physically-based analytical models is based on a deep study and comprehension of the physical effects and on an accurate choice of the parameters. Despite the importance of these aspects, analytical models are always preferred to add new physical aspects in commercial tools.

Finally, with the growing interest in alternative channel materials, such meth-

ods, mainly applied to the analysis of carrier mobility, could find new important applications.

In the second part of this thesis a TCAD simulation study has been presented, aimed at gaining an insight of the physical mechanisms playing a role in the gas detection of silicon-nanowire sensors. The accurate study and modeling of the effects of trap states and fixed charges generated upon gas exposure at the device interface has allowed to reproduce the experimental trans-characteristics of chemical nanowire sensors fabricated with a tow-down approach. A TCAD analysis has been also carried out to provide guidelines for the optimization of the sensor sensitivity. The sensor performance are shown to monotonically increases with the reduction of device width, especially below 50 nm.

Despite the above important results, further experimental and simulation analysis are needed for the optimization of different aspects. The development of ad-hoc simulation tools is essential to the purpose of investigating nanoscale dimensions of the nanowire cross-section (below 25 nm) with strong geometrical quantum confinement. In addition, an important issue would be the study of the minimum distance between parallel wires in the array of nanowire sensors from a process point of view (limitations given by the functionalization, limitations given by the exposition to gas, and the electrostatic interactions between neighbors). Advanced characterization techniques could be adopted to study the interface properties in bared, functionalized, and nanowires with oxide on top. Finally, in a multidisciplinary framework, the analysis of further chemical grafted molecules could contribute to optimize the chemical-electrical transduction mechanism, providing faster sensing. Appendices

Appendix A

Mobility model coefficients

In the following pages all the parameters of the analytical mobility models presented in this thesis are reported.

Relative occupancy [Eqs. (1.27) and (1.45)]								
Parameter	(100)	(110)	(111)	Units				
$E_{\rm CE01,2}$	0.2	0.237	0.236	eV				
	0.243	0.221	0.236	eV				
$\chi_{1,2}$	0.622	0.66	0.667	-				
χ_3	0.661	0.65	0.667	-				
	Phonon scattering [Eqs. (1.33) and (1.35)]							
$C_{1,2}$	1.645×10^{7}	2.04×10^7	1.53×10^7	$\mathrm{cm}^{3}\mathrm{eV}^{-1}\mathrm{s}^{-3}$				
C_3	2.942×10^7	1.177×10^{7}	1.53×10^7	$\mathrm{cm}^{3}\mathrm{eV}^{-1}\mathrm{s}^{-3}$				
W _{T01,2}	5×10^{-11}	1.8×10^{-10}	1.5×10^{-10}	cm				
W _{T03}	2×10^{-10}	7×10^{-11}	1.5×10^{-10}	cm				
W _{E01,2}	4.9×10^{-7}	3×10^{-7}	3.4×10^{-7}	cm				
$W_{\rm E03}$	2.1×10^{-7}	4.3×10^{-7}	3.4×10^{-7}	cm				
$\gamma_{1,2}$	0.17	0.29	0.29	-				
γ_3	0.29	0.2	0.29	-				
	Coulomb scatte	ering [Eqs. (1.3	(39) and (1.40)]					
$C_{\rm CS0}$	3.2×10^{-5}	9.6×10^{-5}	1.92×10^{-4}	s^{-1}				
σ	1.18	0.4	0.45	-				
η	0.5	0.33	0.33	_				
	Surface-	roughness [Eq.	(1.42)]					
$C_{\rm SR0}$	5.542×10^{13}	1.724×10^{14}	8.005×10^{13}	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$				
δ	2.7	1.5	1	-				
	Thickness-fluctuation [Eqs. (1.51) and (1.52)]							
$ au_{\delta t_{Sil}}(25 \text{ K})$	8×10^{-3}	-	-	S				
$ au_{\delta t_{Si}h}(25 \text{ K})$	0.33	-	-	s				
$ au_{\delta t_{Si}l}(300 \text{ K})$	$2.6 imes 10^{-2}$	2.2×10^{-3}	-	s				
$\tau_{\delta t_{Si}h}(300 \text{ K})$	0.5	0.025	-	s				
$C_{\delta t_{\rm Si}}$	160	160	-	-				
Suppression of inter-valley phonon scattering [Eq. (1.53)]								
f_0	0.278	-	-	-				
$\Delta E_{\rm C0_1}$	125	-	-	meV				
$\Delta E_{\rm C0_h}$	105	-	-	meV				
C_l	14	-	-	meV				
C_h	50	-	-	meV				
Surface-phonons [Eq. (1.56)]								
$C_{\rm SP0}$	3.617×10^8	4.167×10^{7}	-	s^{-1}				
γ	10	8	-	-				
Volume inversion correction [Eq. (1.57)]								
C'_{vi}	-	1.18	-	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$				
$C_{vi}^{\prime\prime}$	-	5.8	-	-				
$C_{aa}^{\prime\prime\prime\prime}$	-	0.4	-	-				

 Table A.1: Parameters of the electron effective mobility model (Chapter 1)

Relative occupancy [Eqs. (1.27) and (1.45)]							
Parameter	(100)	(110)	(111)	Units			
$E_{\rm CE01,2}$	0.2	0.237	0.236	eV			
	0.243	0.221	0.236	eV			
$\chi_{1,2}$	0.622	0.66	0.667	-			
χ_3	0.661	0.65	0.667	-			
	Phonon scatte	ering [Eqs. (1.3)	(3) and (1.35)]				
$C_{1,2}$	1.556×10^{7}	2.176×10^{7}	1.53×10^7	$cm^{3}eV^{-1}s^{-3}$			
C_3	2.942×10^{7}	1.177×10^{7}	1.53×10^7	$cm^{3}eV^{-1}s^{-3}$			
W _{T01,2}	5×10^{-11}	1.8×10^{-10}	1.5×10^{-10}	cm			
W _{T03}	2×10^{-10}	7×10^{-11}	1.5×10^{-10}	cm			
W _{E01,2}	4.9×10^{-7}	3×10^{-7}	3.4×10^{-7}	cm			
$W_{\rm E03}$	2.1×10^{-7}	4.3×10^{-7}	3.4×10^{-7}	cm			
$\gamma_{1,2}$	0.17	0.29	0.29	-			
γ_3	0.29	0.2	0.29	-			
	Coulomb scatt	ering [Eqs. (1.3	(39) and (1.40)]				
$C_{\rm CS0}$	3.2×10^{-5}	1.34×10^{-4}	1.92×10^{-4}	s ⁻¹			
σ	1.18	0.4	0.45	-			
η	0.5	0.33	0.33	-			
	Surface-	roughness [Eq.	(1.42)]	·			
$C_{\rm SR0}$	6.773×10^{13}	3.386×10^{14}	8.005×10^{13}	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$			
δ	2.7	2	1	-			
Thickness-fluctuation [Eqs. (1.51) and (1.52)]							
$ au_{\delta t_{Sil}}(25 \text{ K})$	8×10^{-3}	-	-	S			
$ au_{\delta t_{Si}h}(25 \text{ K})$	0.33	-	-	s			
$ au_{\delta t_{Si}l}(300 \text{ K})$	3.4×10^{-2}	2.37×10^{-3}	-	s			
$ au_{\delta t_{Si}h}(300 \text{ K})$	1.053	0.054	-	s			
$C_{\delta t_{\rm Si}}$	160	160	-	-			
Suppre	Suppression of inter-valley phonon scattering [Eq. (1.53)]						
f_0	0.278	-	-	-			
$\Delta E_{\rm C0_l}$	125	-	-	meV			
$\Delta E_{\rm C0_h}$	105	_	_	meV			
C_l	14	_	_	meV			
C_h	50	-	-	meV			
Surface-phonons [Eq. (1.56)]							
$C_{\rm SP0}$	5.16×10^{8}	4.167×10^{7}	-	s ⁻¹			
γ	10	8	-	-			
Volume inversion correction [Eq. (1.57)]							
C'_{vi}	-	1.18	-	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$			
$C_{vi}^{\prime\prime}$	-	5.8	-	-			
$C_{vi}^{\prime\prime\prime}$	-	0.4	-	-			

 Table A.2: Parameters of the electron local mobility model (Chapter 1)

119

Acoustic phonon scattering $[Eq. (2.8)]$						
Parameter	(100)	(110)	(111)	Units		
C_1	2.962×10^7	3.416×10^{7}	3.03×10^7	${\rm cm}^{3}{\rm eV}^{-1}{\rm s}^{-3}$		
C_2	1.584×10^7	2.321×10^7	3.03×10^7	$\mathrm{cm}^{3}\mathrm{eV}^{-1}\mathrm{s}^{-3}$		
$W_{\rm T01}$	1.7×10^{-10}	1×10^{-11}	2×10^{-10}	cm		
$W_{\rm T02}$	2.8×10^{-10}	1.8×10^{-10}	5×10^{-10}	cm		
$W_{\rm E01}$	3.3×10^{-7}	4.8×10^{-7}	3.1×10^{-7}	cm		
$W_{\rm E02}$	3×10^{-7}	1.55×10^{-7}	2.25×10^{-7}	cm		
γ_1	0.29	0.17	0.24	-		
γ_2	0.29	0.33	0.29	-		
	Optical pho	onon scattering	[Eq. (2.9)]			
$C_{\rm OP}$	-	1.242×10^{14}	-	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$		
	Сс	oulomb scatteri	ng			
$C_{\rm CS0}$	2.73×10^{-5}	1.64×10^{-4}	4.64×10^{-4}	s^{-1}		
σ	0.844	0.4	0.844	-		
η	0.33	0.33	0.33	-		
Sca	attering induce	ed by interface	states [Eq. 2.1	0]		
C_{it_0}	1.1×10^{-4}	$6.6 imes 10^{-4}$	-	s^{-1}		
	S	urface-roughne	SS			
$C_{\rm SR0}$	4.43×10^{13}	3.92×10^{14}	6.823×10^{12}	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$		
δ	1.3	1.5	2	-		
	Thi	ckness-fluctuat	tion			
$ au_{\delta t_{Sil}}(25 \text{ K})$	1×10^{-5}	-	-	S		
$ au_{\delta t_{Si}h}(25 \text{ K})$	0.02		-	S		
$ au_{\delta t_{Si}l}(40 \text{ K})$	-	3.3×10^{-4}	-	S		
$ au_{\delta t_{Si}h}(40 \text{ K})$	-	0.526	-	S		
$ au_{\delta t_{Si}l}(300 \text{ K})$	1.3×10^{-4}	4.3×10^{-3}	-	S		
$ au_{\delta t_{Si}h}(300 \text{ K})$	0.05	0.132	-	S		
η_1	7.6	6	-	S		
η_2	0.65	1.5	-	S		
$C_{\delta t_{\rm Si}}$	4×10^3	4×10^3	-	-		
Suppression of inter-valley phonon scattering [Eq. (2.13)]						
f_0	-	0.38	-	-		
$\Delta E_{\rm V0}$	-	97	-	meV		
$C_{ m iv}$	-	15	-	meV		
Surface-phonons						
C _{SP0}	2.94×10^{16}	1.631×10^{3}	-	s^{-1}		
\sim	15.6	4.5	-	_		

Table A.3: Parameters of the hole effective mobility model (Chapter 2)
Acoustic phonon scattering [Eq. (2.8)]						
Parameter	(100)	(110)	(111)	Units		
C_1	3.789×10^7	3.416×10^7	3.03×10^7	$cm^{3}eV^{-1}s^{-3}$		
C_2	1.235×10^7	1.934×10^7	3.03×10^7	$cm^{3}eV^{-1}s^{-3}$		
W _{T01}	1.7×10^{-10}	1×10^{-11}	2×10^{-10}	cm		
W _{T02}	2.8×10^{-10}	1.8×10^{-10}	5×10^{-10}	cm		
$W_{\rm E01}$	3.3×10^{-7}	4.8×10^{-7}	3.1×10^{-7}	cm		
$W_{\rm E02}$	3×10^{-7}	1.55×10^{-7}	2.25×10^{-7}	cm		
γ_1	0.29	0.17	0.24	-		
γ_2	0.29	0.33	0.29	-		
	Optical pho	onon scattering	[Eq. (2.9)]			
COP	-	1.242×10^{14}	_	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$		
	Сс	oulomb scatteri	ng			
$C_{\rm CS0}$	2.73×10^{-5}	1.64×10^{-4}	4.64×10^{-4}	s^{-1}		
σ	0.844	0.4	0.844	-		
η	0.33	0.33	0.33	-		
Sca	attering induce	ed by interface	states [Eq. 2.1	0]		
C_{it_0}	1.1×10^{-4}	$6.6 imes 10^{-4}$	-	s^{-1}		
Surface-roughness						
$C_{\rm SR0}$	6.81×10^{13}	4.26×10^{14}	6.823×10^{12}	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$		
δ	1.5	1.5	2	-		
Thickness-fluctuation						
$ au_{\delta t_{Sil}}(25 \text{ K})$	1×10^{-5}	-	-	S		
$ au_{\delta t_{Si}h}(25 \text{ K})$	0.02	-	-	S		
$ au_{\delta t_{Si}l}(40 \text{ K})$	-	3.3×10^{-4}	-	S		
$ au_{\delta t_{Si}h}(40 \text{ K})$	-	0.526	-	S		
$ au_{\delta t_{Sil}}(300 \text{ K})$	1.9×10^{-4}	3.73×10^{-3}	-	S		
$ au_{\delta t_{Si}h}(300 \text{ K})$	0.15	0.111	-	S		
η_1	7.6	6	-	S		
η_2	0.65	1.5	-	S		
$C_{\delta t_{\rm Si}}$	4×10^3	4×10^3	-	-		
Suppression of inter-valley phonon scattering $[Eq. (2.13)]$						
f_0	_	0.494	-	-		
$\Delta E_{\rm V0}$	-	97	-	meV		
$C_{\rm iv}$	-	15	-	meV		
Surface-phonons						
C _{SP0}	2.94×10^{16}	1.435×10^{3}	-	s ⁻¹		
γ	15.6	4.5	-	-		

 Table A.4: Parameters of the hole local mobility model (Chapter 2)

ELECTRONS							
Remote phonon scattering $[Eq. (3.1)]$							
Parameter	$\rm SiO_2/PolySi$	metal gate (TiN)	high- κ (HfO ₂)	Units			
$C_{\rm RPS}$	2.686×10^7	4.626×10^6	4.865×10^6	$\mathrm{cm}^{3}\mathrm{eV}^{-1}\mathrm{s}^{-3}$			
$k_{\rm RPS}$	1.94×10^7	0.68×10^{7}	0.69×10^{7}	$\rm cm^{-1}$			
Remote surface-roughness scattering [Eq. (3.2)]							
$C_{\rm RSS}$	6.658×10^{13}	-	-	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$			
$k_{\rm RSS}$	1.2×10^7	-	-	$\rm cm^{-1}$			
Remote Coulomb scattering [Eq. (3.3)]							
$C_{\rm RCS}$	5.654×10^{-4}	5.56×10^{-3}	1.645×10^{-2}	s^{-1}			
$k_{\rm RCS}$	2.7×10^6	1.88×10^{7}	1.9×10^{7}	$\rm cm^{-1}$			
HOLES							
Remote phonon scattering $[Eq. (3.1)]$							
Parameter	$\rm SiO_2/PolySi$	metal gate (TiN)	high- κ (HfO ₂)	Units			
$C_{\rm RPS}$	-	-	5.661×10^{7}	$cm^{3}eV^{-1}s^{-3}$			
$k_{ m RPS}$	-	-	2.37×10^7	$\rm cm^{-1}$			
Remote surface-roughness scattering [Eq. (3.2)]							
$C_{\rm RSS}$	-	-	-	$\mathrm{cm}^{2}\mathrm{eV}^{-1}\mathrm{s}^{-3}$			
$k_{ m RSS}$	-	-	-	$\rm cm^{-1}$			
Remote Coulomb scattering [Eq. (3.3)]							
$C_{\rm RCS}$	-	-	$3.\overline{159 \times 10^{-4}}$	s^{-1}			

 Table A.5: Parameters of the remote scattering terms (Chapter 3)

 Table A.6: Parameters of the strain model (Chapter 4)

Parameter	LH $(i=1)$	HH (i=2)		
$b_{v,1}$	-36.65	-78.53		
$b_{v,2}$	1.816×10^{3}	2.056×10^{3}		
$b_{v,3}$	-2.153×10^4	0		
Parameters of the hole principal effective masses Eq. $[(4.15)]$				
Parameter	$\varepsilon_{xy} > 0$	$\varepsilon_{xy} < 0$		
$c_{2,1}$	88.26	48.54		
$c_{2,2}$	1948	1363.6		
$d_{2,1}$	-17.652	52.956		
$d_{2,2}$	194.8	1363.6		

Bibliography

- "International Technology Roadmap for Semiconductors 2010 Update Edition," http://public.itrs.net. [cited at p. xiv, 3, 6, 7]
- S. Reggiani, E. Gnani, A. Gnudi, M. Rudan, and G. Baccarani, "Low-Field Electron Mobility Model for Ultra-Thin-Body SOI and Double-Gate MOSFETs with Estremely Small Silicon Thicknesses," *IEEE Trans. on Electr. Dev.*, vol. 54, no. 9, pp. 2204–2212, 2007. [cited at p. 3, 4, 26, 28, 29, 31, 47]
- [3] C. Kampen, A. Burenkov, J. Lorenz, H. Ryssel, V. Aubry-Fortuna, and A. Bournel, "An Application-Driven Improvement of the Drift-Diffusion Model for Carrier Transport in Decanano-Scaled CMOS Devices," *IEEE Trans. on Electr. Dev.*, vol. 55, no. 11, pp. 3227–3235, 2008. [cited at p. 3]
- [4] R. Ohba and T. Mizuno, "Nonstationary electron/hole transport in sub-0.1 μm MOS devices: Correlation with mobility and low-power CMOS application," *IEEE Trans. on Electr. Dev.*, vol. 48, no. 2, pp. 338–343, 2001. [cited at p. 3]
- [5] M. Lundstrom, "On the mobility versus drain current relation for a nanoscale MOSFET," *IEEE Electr. Dev. Lett.*, vol. 22, no. 6, pp. 293–295, 2001. [cited at p. 3]
- [6] S. E. Thompson, S. Suthram, Y. Sun, G. Sun, S. Parthasarathy, M. Chu, and T. Nishida, "Future of Strained Si/Semiconductors in Nanoscale MOS-FETs," in *Technical Digest of the International Electron Devices Meeting*, 2006, pp. 1–4. [cited at p. 3, 60, 68, 72, 73, 89]
- [7] M. Yang, V. W. C. Chan, K. K. Chan, L. Shi, D. M. Fried, J. H. Stathis, A. I. Chou, E. Gusev, J. A. Ott, L. E. Burns, M. V. Fischetti, and M. Ieong, "Hybrid-orientation technology (HOT): Opportunities and challenges," *IEEE Trans. on Electr. Dev.*, vol. 53, no. 5, pp. 965–978, 2006. [cited at p. 3, 25, 43]

- [8] S. Mehrotra, A. Paul, M. Luisier, and G. Klimeck, "Surface and Orientation Dependence on Performance of Trigated Silicon Nanowire pMOSFETs," in *IEEE Workshop on Microelectronics and Electron Devices*, 2009, pp. 1–4. [cited at p. 3]
- [9] B. Yu, L. Chang, S. Ahmed, H. Wang, S. Bell, C. Yang, C. Tabery, C. Ho, Q. Xiang, T. King, J. Bokor, C. Hu, M. Lin, and D. Kyser, "FINFET scaling to 10 nm gate length," in *Technical Digest of the International Electron Devices Meeting*, 2002, pp. 251–254. [cited at p. 3]
- [10] J. Kavalieros, B. Doyle, S. Datta, G. Dewey, M. Doczy, B. Jin, D. Lionberger, M. Metz, W. Rachmady, M. Radosavljevic, U. Shah, N. Zelick, and R. Chau, "Tri-Gate Transistor Architecture with High-κ Gate Dielectrics, Metal Gates and Strain Engineering," in Symposium on VLSI Technology, Dig. of Technical Papers, 2006, pp. 50–51. [cited at p. 3]
- [11] A. Rahman and M. Lundstrom and A. Ghosh, "Generalized effectivemass approach for n-type metal-oxide-semiconductor field-effect transistors on arbitrarily oriented wafers," J. Appl. Phys., vol. 97, p. 053702, 2005. [cited at p. 3, 11, 13]
- [12] K. Uchida, H. Watanabe, A. Kinoshita, J. Koga, T. Numata, and S. Takagi, "Experimental study on carrier transport mechanisms in ultrathin-body SOI n- and p-MOSFETs with SOI thicknesses less than 5 nm," in *Techni*cal Digest of the International Electron Devices Meeting, 2002, pp. 47–50. [cited at p. 3, 27, 28, 35, 46, 47, 48, 50, 51]
- [13] K. Uchida, J. Koga, and S. Takagi, "Experimental study on carrier transport mechanisms in double- and single-gate ultrathin-body MOSFETs Coulomb scattering, volume inversion, and δT_{SOI} -induced scattering -," in *Technical Digest of the International Electron Devices Meeting*, 2003, pp. 805–808. [cited at p. 3, 23, 24, 32]
- [14] D. Esseni, M. Mastrapasqua, G. K. Celler, C. Fiegna, L. Selmi, and E. Sangiorgi, "An experimental study of mobility enhancement in ultrathin SOI transistors operated in double-gate mode," *IEEE Trans. on Electr. Dev.*, vol. 50, no. 3, pp. 802–808, 2003. [cited at p. 3, 32]
- [15] G. Tsutsui, M. Saitoh, T. Saraya, T. Nagumo, and T. Hiramoto, "Mobility Enhancement due to Volume Inversion in (110)-oriented Ultra-thin Body Double-gate nMOSFETs with Body Thickness less than 5 nm," in *Technical Digest of the International Electron Devices Meeting*, 2005, pp. 729–732. [cited at p. 3, 25, 26, 30, 31, 32, 33]

- [16] G. Tsutsui and T. Hiramoto, "Mobility and Threshold-Voltage Comparison Between (110)- and (110)-Oriented Ultrathin-Body Silicon MOS-FETs," *IEEE Trans. on Electr. Dev.*, vol. 53, no. 10, pp. 2582–2588, 2006. [cited at p. 3, 23, 24, 29, 31, 33, 35, 36]
- [17] D. Esseni, A. Abramo, L. Selmi, and E. Sangiorgi, "Physically Based Modeling of Low Field Electron Mobility in Ultrathin Single- and Double-Gate SOI n-MOSFETs," *IEEE Trans. on Electr. Dev.*, vol. 50, no. 12, pp. 2445– 2455, 2003. [cited at p. 3, 20, 23, 24, 25, 26, 29, 46]
- [18] L. Silvestri, S. Reggiani, E. Gnani, A. Gnudi, and G. Baccarani, "Unified model for low-field electron mobility in bulk and SOI-MOSFETs with different substrate orientations and its application to quantum drift-diffusion simulations," in *Proc. of the International Conference on Ultimate Integration of Silicon (ULIS) 2008*, pp. 129–132. [cited at p. 4]
- [19] N. W. Ashcroft and N. D. Mermin, Solid State Physics. W. B. Saunders, Philadelphia, 1976. [cited at p. 4]
- [20] S. M. Sze, Physics of Semiconductor Devices. New York: Wiley-Interscience, 1981. [cited at p. 5, 6, 19]
- [21] T. Ando, A. B. Fowler, and F. Stern, "Electronic Properties of Two-Dimensional Systems," *Rev. Mod. Phys.*, vol. 54, p. 437, 1982. [cited at p. 7, 9, 17, 42]
- [22] F. Stern and E. Howard, "Properties of Semiconductor Surface Inversion Layers in the Electric Quantum Limit," *Phys. Rev.*, vol. 163, no. 3, pp. 816–835, 1967. [cited at p. 8, 19, 38]
- [23] E. Gnani, S. Reggiani, M. Rudan, and G. Baccarani, "On the Electrostatics of Double-Gate and Cylindrical Nanowire MOSFETs," J. Comput. Electron., vol. 4, pp. 71–74, 2005. [cited at p. 10]
- [24] S. Takagi, A. Turiumi, M. Iwase, and H. Tango, "On the universality of inversion layer mobility in Si MOSFETs: Part II - Effects of surface orientation," *IEEE Trans. on Electr. Dev.*, vol. 41, no. 12, pp. 2363–2368, 1994. [cited at p. 11, 18, 19, 25, 45, 54, 57]
- [25] M. L. Lee, E. A. Fitzgerald, M. T. Bulsara, M. T. Currie, and A. Lochtefeld, "Strained Si, SiGe, and Ge channels for high-mobility metal-oxidesemiconductor field-effect transistors," J. Appl. Phys., vol. 97, p. 11101, 2005. [cited at p. 12, 63, 64]
- [26] S. Inc., Sentaurus Device Users' Manual, release D-2010.03 ed. [cited at p. 14, 98, 104]

- [27] A. Nathan and H. Baltes, Microtransducer CAD Physical and Computational Aspects, ser. Computational Microelectronics. Springer-Verlag Wien New York, 1999. [cited at p. 14, 40, 61]
- S. Dhar, H. Kosina, V. Palankovsky, E. Ungersboeck, and S. Selberherr, "Electron Mobility Model for Strained-Si Devices," *IEEE Trans. on Electr. Dev.*, vol. 52, no. 4, pp. 527–533, 2005. [cited at p. 15]
- [29] M. Ishizaka, T. Iizuka, S. Ohi, M. Fukuma, and H. Mikoshiba, "Advanced Electron Mobility Model of MOS Inversion Layer Considering 2Ddegenerated Electron Gas Physics," in *Technical Digest of the International Electron Devices Meeting*, 1990, pp. 763–766. [cited at p. 15]
- [30] S. Takagi, A. Turiumi, M. Iwase, and H. Tango, "On the universality of inversion layer mobility in Si MOSFETs: Part I - Effects of substrate impurity concentration," *IEEE Trans. on Electr. Dev.*, vol. 41, no. 12, pp. 2357–2362, 1994. [cited at p. 18]
- [31] L. Lucci, P. Palestri, D. Esseni, L. Bergagnini, and L. Selmi, "Multisubband Monte Carlo Study of Transport, Quantization, and Electron-Gas Degeneration in Ultrathin SOI n-MOSFETs," *IEEE Trans. on Electr. Dev.*, vol. 54, no. 5, pp. 1156–1164, 2007. [cited at p. 19]
- [32] B. Goebel, D. Schumann, and E. Bertagnolli, "Vertical N-Channel MOS-FETs for Extremely High Density Memories: The Impact of Interface Orientation on Device Performance," *IEEE Trans. on Electr. Dev.*, vol. 48, no. 5, pp. 897–906, 2001. [cited at p. 19, 44]
- [33] S. Takagi, J. Koga, and A. Toriumi, "Mobility Enhancement of SOI MOS-FETs due to Subband Modulation in Ultrathin SOI Films," Jpn. J. Appl. Phys., vol. 37, pp. 1289–1294, 1998. [cited at p. 20, 27, 28]
- [34] K. Uchida, A. Kinoshita, and M. Saitoh, "Carrier Transport in (110) nMOS-FETs: Subband Structure, Non-Parabolicity, Mobility Characteristics, and Uniaxial Stress Engineering," in *Technical Digest of the International Electron Devices Meeting*, 2006, pp. 1–3. [cited at p. 20, 21, 22, 23, 68, 69, 70, 71, 79]
- [35] C. Jacoboni and L. Reggiani, "The Monte Carlo method for the solution of charge transport in semiconductors with applications to covalent materials," *Rev. Mod. Phys.*, vol. 55, no. 3, pp. 645–705, 1983. [cited at p. 21]
- [36] J. Koga, S. Takagi, and A. Toriumi, "A Comprehensive Study of MOS-FET Electron Mobility in Both Weak and Strong Inversion Regimes," in *Technical Digest of the International Electron Devices Meeting*, 1994, pp. 475–478. [cited at p. 24]

- [37] O. Weber and S. Takagi, "Experimental Examination and Physical Understanding of the Coulomb Scattering Mobility in Strained-Si nMOS-FETs," *IEEE Trans. on Electr. Dev.*, vol. 55, no. 9, pp. 2386–2396, 2008. [cited at p. 24]
- [38] Y. Nakabayashi, T. Ishihara, J. Koga, M. Takayanagi, and S. Takagi, "New findings on inversion-layer mobility in highly doped channel Si MOSFETs," in *Technical Digest of the International Electron Devices Meeting*, 2005, pp. 133–136. [cited at p. 24]
- [39] L. Trojman, L. Pantisano, I. Ferain, S. Severi, H. Maes, and G. Goeseneken, "Mobility and Dielectric Quality of 1-nm EOT HfSiON on Si(110) and (100)," *IEEE Trans. on Electr. Dev.*, vol. 55, no. 12, pp. 3414–3420, 2008. [cited at p. 25, 43]
- [40] K. Uchida and S. Takagi, "Carrier scattering induced by thickness fluctuation of silicon-on-insulator film in ultrathin-body metal-oxidesemiconductor field-effect transitors," *Appl. Phys. Lett.*, vol. 82, no. 17, pp. 2916–2918, 2003. [cited at p. 26, 27]
- [41] M. T. Currie, C. V. Leitz, T. A. Langdo, G. Taraschi, E. A. Fitzgerald, and D. A. Antoniadis, "Carrier mobilities and process stability of strained Si n- and p- MOSFETs on SiGe virtual substrates," J. Vac. Sci. Technol. B, vol. 19, no. 6, pp. 2268–2279, 2001. [cited at p. 28, 68, 73]
- [42] K. Rim, K. Chan, L. Shi, D. Boyd, J. Ott, N. Klymko, F. Cardone, L. Tai, S. Koester, M. Cobb, D. Canaperi, B. To, E. Duch, I. Babich, R. Carruthers, P. Saunders, G. Walker, Y. Zhang, M. Steen, and M. Ieong, "Fabrication and Mobility Characteristics of Ultra-thin Strained Si Directly on Insulator (SSDOI) MOSFETs," in *Technical Digest of the International Electron Devices Meeting*, 2003, pp. 3.1.1–3.1.4. [cited at p. 28, 68, 73]
- [43] J. Welser, J. L. Hoyt, S. Takagi, and J. F. Gibbons, "Strain Dependence of the Performance Enhancement in Strained-Si n-MOSFETs," in *Technical Digest of the International Electron Devices Meeting*, 1994, pp. 15.2.1– 15.2.4. [cited at p. 28]
- [44] I. Balslev, "Influence of Uniaxial Stress on the Indirect Absorption Edge in Silicon and Germanium," *Phys. Rev.*, vol. 143, no. 2, pp. 636–647, 1966. [cited at p. 28]
- [45] V. Sverdlov, E. Ungersboeck, H. Kosina, and S. Selberherr, "Volume inversion mobility in SOI MOSFETs for different thin body orientations," *Solid-State Electr.*, vol. 51, pp. 299–305, 2007. [cited at p. 32]

- [46] A. Schenk, "A local mobility model for ultra-thin DGSOI nMOSFETs," in Proc. of SISPAD 2004, pp. 113–116. [cited at p. 35]
- [47] G. Tsutsui, M. Saitoh, and T. Hiramoto, "Experimental Study on Superior Mobility in (110)-Oriented UTB SOI pMOSFETs," *IEEE Electr. Dev. Lett.*, vol. 26, no. 11, pp. 836–838, 2005. [cited at p. 37, 42, 43, 44, 47, 48, 49, 50, 51]
- [48] M. V. Fischetti, Z. Ren, P. M. Solomon, M. Yang, and K. Rim, "Six-band k·p calculation of the hole mobility in silicon inversion layers: Dependence on surface orientation, strain, and silicon thickness." J. Appl. Phys., vol. 94, pp. 1079–1095, 2003. [cited at p. 37, 38, 39, 40, 42, 43, 44, 45, 46, 47, 61, 62, 63, 64]
- [49] A. Pham, C. Jungemann, and B. Meinerzhagen, "Physics-Based Modeling of Hole Inversion-Layer Mobility in Strained-SiGe-on-Insulator," *IEEE Trans. on Electr. Dev.*, vol. 54, no. 9, pp. 2174–2182, 2007. [cited at p. 37]
- [50] L. Donetti, F. Gamiz, and N. Rodriguez, "Simulation of hole mobility in two-dimensional systems," *Semiconductor Science and Technology*, vol. 24, p. 035016, 2009. [cited at p. 37]
- [51] M. D. Michielis, D. Esseni, Y. L. Tsang, P. Palestri, L. Selmi, A. G. O'Neill, and S. Chattopadhyay, "A Semianalitical Description of the Hole Band Structure in Inversion Layers for the Physically Based Modeling of pMOS Transistors," *IEEE Trans. on Electr. Dev.*, vol. 54, no. 9, pp. 2164–2173, 2007. [cited at p. 39, 41]
- [52] M. Saitoh, S. Kobayashi, and K. Uchida, "Physical Understanding of Fundamental Properties of Si (110) pMOSFETs Inversion Layer Capacitance, Mobility Universality, and Uniaxial Stress Effects," in *Technical Digest of the International Electron Devices Meeting*, 2007, pp. 711–714. [cited at p. 45]
- [53] H. Irie, K. Kita, K. Kyuno, and A. Toriumi, "In-Plane Mobility Anisotropy and Universality Under Uni-axial Strains in n- and p- MOS Inversion Layers on (100), (110), and (111) Si," in *Technical Digest of the International Electron Devices Meeting*, 2004, pp. 9.5.1–9.5.4. [cited at p. 45, 74]
- [54] B. Mereu, C. Rossel, E. P. Gusev, and M. Yang, "The role of Si orientation and temperature on the carrier mobility in metal oxide semiconductor fieldeffect transistors with ultrathin HfO2 gate dielectrics," J. Appl. Phys., vol. 100, p. 014504, 2006. [cited at p. 45, 54, 56, 58]
- [55] J. Koga, S. Takagi, and A. Toriumi, "Influences of Buried-Oxide Interface on Inversion-Layer Mobility in Ultra-Thin SOI MOSFETs," *IEEE Trans.* on Electr. Dev., vol. 49, no. 6, pp. 1042–1048, 2002. [cited at p. 44]

- [56] S. Kobayashi, M. Saitoh, and K. Uchida, "More-than-Universal Mobility in Double-Gate SOI p-FETs with Sub-10-nm Body Thickness-Role of Light-Hole Band and Compatibility with Uniaxial Stress Engineering," in *Technical Digest of the International Electron Devices Meeting*, 2007, pp. 707–710. [cited at p. 50, 51]
- [57] R. M. Wallace and G. D. Wilk, Materials Issues for High-κ Gate Dielectric Selection and Integration, in High Dielectric Constant Materials-VLSI MOSFET Applications. H. R. Huff and D. C. Gilmer, Eds. Berlin, Germany: Springer-Verlag, 2005. [cited at p. 53]
- [58] V. Misra, Issues in metal gate electrode selection for bulk CMOS devices, in High Dielectric Constant Materials-VLSI MOSFET Applications. H. R. Huff and D. C. Gilmer, Eds. Berlin, Germany: Springer-Verlag, 2005. [cited at p. 53]
- [59] D. Esseni and A. Abramo, "Modeling of Electron Mobility Degradation by Remote Coulomb Scattering in Ultrathin Oxide MOSFETs," *IEEE Trans.* on Electr. Dev., vol. 50, no. 7, pp. 1665–1674, 2003. [cited at p. 53]
- [60] F. Gamiz and J. B. Roldan, "Scattering of Electrons in Silicon Inversion Layers by Remote Surface Roughness," J. of Appl. Phys., vol. 94, no. 1, pp. 392–399, 2003. [cited at p. 53]
- [61] M. V. Fischetti, D. A. Neumayer, and E. Cartier, "Effective Electron Mobility in Si Inversion Layers in Metal-Oxide-Semiconductor Systems with a High-κ Insulator: The Role of Remote Phonon Scattering," J. Appl. Phys., vol. 90, no. 9, pp. 4587–4608, 2001. [cited at p. 53]
- [62] M. Cassé, L. Thevenod, B. Guillaumot, L. Tosti, F. Martin, J. Mitard, O. Weber, F. Andrieu, T. Ernst, G. Reimbold, T. Billon, M. Mouis, and F. Boulanger, "Carrier Transport in HfO2/Metal Gate MOSFETs: Physical Insight Into Critical Parameters," *IEEE Trans. on Electr. Dev.*, vol. 53, no. 4, pp. 759–768, 2006. [cited at p. 54, 55, 56, 57]
- [63] M. Yang, E. P. Gusev, M. Ieong, O. Gluschenkov, D. C. Boyd, K. K. Chan, P. M. Kozlowski, C. P. D'Emic, R. M. Sicina, P. C. Jamison, and A. I. Chou, "Performance Dependence of CMOS on Silicon Substrate Orientation for Ultrathin Oxynitride and HfO2 Gate Dielectrics," *IEEE Electr. Dev. Lett.*, vol. 24, no. 5, pp. 339–341, 2003. [cited at p. 54, 58]
- [64] S. Takagi and M. Takayanagi, "Experimental Evidence of Inversion-Layer Mobility Lowering in Ultrathin Gate Oxide Metal-Oxide-Semiconductor Field-Effect-Transistors with Direct Tunneling Current," Jpn. J. Appl. Phys., vol. 41, pp. 2348–2352, 2002. [cited at p. 54, 55]

- [65] L. Thevenod, M. Cassé, M. Muis, G. Reimbold, F. Fillot, B. Guillaumot, and F. Boulanger, "Influence of TiN metal gate on Si/SiO2 surface roughness in N and PMOSFETs," *Microelectronic Engineering*, vol. 80, pp. 11– 14, 2005. [cited at p. 56]
- [66] R. People, J. C. Bean, D. V. Lang, A. M. Sergent, H. L. Stormer, K. W. Wecht, R. T. Lynch, and K. Baldwin, "Modulation doping in Ge_xSi_(1-x)/Si strained layer heterostructures," *Appl. Phys. Lett.*, vol. 45, pp. 1231–1233, 1984. [cited at p. 59]
- [67] S. Ito, H. Namba, K. Yamaguchi, T. Hirata, K. Ando, S. Koyama, S. Kuroki, N. Ikezawa, T. Suzuki, T. Saitoh, and T. Horiuchi, "Mechanical stress effect of etch-stop nitride and its impact on deep submicron transistor design," in *Technical Digest of the International Electron Devices Meeting*, 2000, pp. 247–250. [cited at p. 59]
- [68] S. Gannavaram, N. Pesovic, and C. Ozturk, "Low temperature (800° C) recessed junction selective silicon-germanium source/drain technology for sub-70 nm CMOS," in *Technical Digest of the International Electron De*vices Meeting, 2000, pp. 437–440. [cited at p. 59]
- [69] S. E. Thompson, G. Sun, K. Wu, J. Lim, and T. Nishida, "Key differences for process-induced uniaxial vs. substrate-induced biaxial stressed Si and Ge channel MOSFETs," in *Technical Digest of the International Electron Devices Meeting*, 2004, pp. 221–224. [cited at p. 59]
- [70] N. Serra, F. Conzatti, D. Esseni, M. D. Michielis, P. Palestri, L. Selmi, S. Thomas, T. Whall, E. Parker, D. Leadley, L. Witters, A. Hikavyy, M. Hÿtch, F. Houdellier, E. Snoeck, T. Wang, W. Lee, G. Vellianitis, M. van Dal, B. Duriez, G. Doornbos, and R. Lander, "Experimental and physicsbased modeling assessment of strain induced mobility enhancement in Fin-FETs," in *Technical Digest of the International Electron Devices Meeting*, 2009, pp. 71–74. [cited at p. 60, 77, 78, 79]
- [71] T. Rudenko, V. Kilchytska, N. Collaert, M. Jurczak, A. Nazarov, and D. Flandre, "Carrier Mobility in Undoped Triple-Gate FinFET Structures and Limitations of Its Description in Terms of Top and Sidewall Channel Mobilities," *IEEE Trans. on Electr. Dev.*, vol. 55, pp. 3532–3541, 2008. [cited at p. 60, 76, 78]
- [72] E. Ungersboeck, S. Dhar, G. Karlowatz, V. Sverdlov, H. Kosina, and S. Selberherr, "The Effect of General Strain on the Band Structure and Electron Mobility of Silicon," *IEEE Trans. on Electr. Dev.*, vol. 54, no. 9, pp. 2183–2190, 2007. [cited at p. 60, 65]

- [73] E. Wang, P. Matagne, L. Shifren, B. Obradovic, R. Kotlyar, S. Cea, M. Stettler, and M. D. Giles, "Physics of Hole Transport in Strained Silicon MOS-FET Inversion Layers," *IEEE Trans. on Electr. Dev.*, vol. 53, pp. 1840– 1851, 2006. [cited at p. 60, 62, 63]
- [74] C. G. Van de Walle and R. M. Martin, "Theoretical Calculations of Heterojunction Discontinuities in Si/Ge System," *Phys. Rev. B*, vol. 34, pp. 5621–5634, 1986. [cited at p. 62]
- [75] S. E. Thompson, G. Sun, Y. S. Choi, and T. Nishida, "Uniaxial-Process-Induced Strained-Si: Extending the CMOS Roadmap," *IEEE Trans. on Electr. Dev.*, vol. 53, pp. 1010–1020, 2006. [cited at p. 62, 63, 72, 73, 74]
- [76] F. Driussi, D. Esseni, L. Selmi, P.-E. Hellström, G. Malm, J. Hallstedt, M. Östling, T. J. Grasby, D. R. Leadley, and X. Mescot, "Experimental and Simulation Study of the Biaxial Strain and Temperature Dependence of the Electron Mobility Enhancement in Si MOSFETs," in *Proc. of the International Conference on Ultimate Integration of Silicon (ULIS) 2007*, pp. 21–24. [cited at p. 63, 64, 68]
- [77] H. M. Nayfeh, C. W. Leitz, A. J. Pitera, E. A. Fitzgerald, J. L. Hoyt, and D. A. Antoniadis, "Influence of High Channel Doping on the Inversion Layer Electron Mobility in Strained Silicon n-MOSFETs," *IEEE Electr. Dev. Lett.*, vol. 24, p. 248, 2003. [cited at p. 63, 64, 68]
- [78] K. Rim, J. Chu, H. Chen, K. Jenkins, T. Kanarsky, K. Lee, A. Mocuta, H. Zhu, R. Roy, J. Newbury, J. Ott., K. Petrarca, P. Mooney, D. Lacey, S. Koester, K. Chan., D. Boyd, M. Ieong, and H.-S. Wong, "Characteristics and Device Design of Sub-100 nm Strained Si N- and PMOSFETs," in *Symposium on VLSI Technology, Dig. of Technical Papers, 2002*, p. 98. [cited at p. 63, 64, 65]
- [79] T. Mizuno, N. Sugiyama, T. Tezuka, T. Numata, and S.Takagi, "High Performance CMOS Operation of Strained-SOI MOSFETs using Thin Film SiGe-on-Insulator Substrate," in *Symposium on VLSI Technology, Dig. of Technical Papers, 2002*, p. 106. [cited at p. 65]
- [80] C. W. Leitz, M. T. Currie, M. L. Lee, Z.-Y. Cheng, D. A. Antoniadis, and E. A. Fitzgerald, "Hole mobility enhancements and alloy scatteringlimited mobility in tensile strained Si/SiGe surface channel metaloxidesemiconductor field-effect transistors," J. Appl. Phys., vol. 92, p. 3745, 2002. [cited at p. 65, 73]
- [81] K. Uchida, T. Krishnamohan, K. C. Saraswat, and Y. Nishi, "Physical Mechanisms of Electron Mobility Enhancement in Uniaxial Stressed MOS-FETs and Impact of Uniaxial Stress Engineering in Ballistic Regime," in

Technical Digest of the International Electron Devices Meeting, 2005, pp. 129–132. [cited at p. 65, 66, 67, 68]

- [82] K. Uchida, R. Zednik, C.-H. Lu, H. Jagannathan, J. McVittie, P. C. McIntyre, and Y. Nishil, "Experimental Study of Biaxial and Uniaxial Strain Effects on Carrier Mobility in Bulk and Ultrathin-body SO1 MOSFETs," in *Technical Digest of the International Electron Devices Meeting*, 2004, pp. 229–232. [cited at p. 67, 69, 73]
- [83] B. Obradovic, P. Matagne, L. Shifren, E. Wang, M. Stettler, J. He, and M. D. Giles, "A Physically-Based Analytic Model for Stress-Induced Hole Mobility Enhancement," in *Proc. of the IWCE 2004*, pp. 26–29. [cited at p. 72]
- [84] L. Washington, F. Nouri, S. Thirupapuliyur, G. Eneman, P. Verheyen, V. Moroz, L. Smith, X. Xu, M. Kawaguchi, T. Huang, K. Ahmed, M. Balseanu, L.-Q. Xia, M. Shen, Y. Kim, R. Rooyackers, K. D. Meyer, and R. Schreutelkamp, "pMOSFET With 200% Mobility Enhancement Induced by Multiple Stressors," *IEEE Electr. Dev. Lett.*, vol. 27, p. 511, 2006. [cited at p. 72, 73]
- [85] K. S. K., T. Saraya, and T. Hiramoto, "Physical understandings of Si (110) hole mobility in ultra-thin body pFETs by <110> and <111> uniaxial compressive strain," in *Technical Digest of the International Electron De*vices Meeting, 2009, pp. 1–4. [cited at p. 74]
- [86] M. Saitoh, A. Kaneko, K. Okano, T. Kinoshita, S. Inaba, Y. Toyoshima, and K. Uchida, "Three-Dimensional Stress Engineering in FinFETs for Mobility/On-Current Enhancement and Gate Current Reduction," in Symposium on VLSI Technology, Dig. of Technical Papers, 2008, pp. 18–19. [cited at p. 74]
- [87] J. Chen, T. Saraya, K. Miyaji, K. Shimizu, and T. Hiramoto, "Electron Mobility in Silicon Gate-All-Around [100]- and [110]-Directed Nanowire Metal-Oxide-Semiconductor Field-Effect Transistor on (100)-Oriented Siliconon-Insulator Substrate Extracted by Improved Split Capacitance-Voltage Method," Jpn. J. Appl. Phys., vol. 48, p. 011205, 2009. [cited at p. 76]
- [88] T. Irisawa, K. Okano, T. Horiuchi, H. Itokawa, I. Mizushima, K. Usuda, T. Tezuka, N. Sugiyama, and S. Takagi, "Electron Mobility and Short-Channel Device Characteristics of SOI FinFETs With Uniaxially Strained (110) Channels," *IEEE Trans. on Electr. Dev.*, vol. 56, pp. 1651–1658, 2009. [cited at p. 76, 77, 78, 79]
- [89] T. Tezuka, E. Toyoda, S. Nakaharai, T. Irisawa, N. Hirashita, Y. Moriyama, N. Sugiyama, N. Taoka, Y. Yamashita, . Kiso, M. Harada, T. Yamamoto,

and S. Takagi, "Observation of Mobility Enhancement in Strained Si and SiGe Tri-Gate MOSFETs with Multi-Nanowire Channels Trimmed by Hydrogen Thermal Etching," in *Technical Digest of the International Electron Devices Meeting*, 2007, pp. 887–890. [cited at p. 77]

- [90] N. Collaert, R. Rooyackers, F. Clemente, P. Zimmerman, I. Cayrefourcq, B. Ghyselen, K. Sand, B. Eyckens, M. Jurczak, and S. Biesemans, "Performance enhancement of MUGFET devices using Super Critical StrainedSOI (SC-SSOI) and CESL," in Symposium on VLSI Technology, Dig. of Technical Papers, 2006. [cited at p. 78]
- [91] W. Xiong, K. Shin, C. R. Cleavelin, T. Schulz, K. Schruefer, I. Cayrefourcq, M. Kennard, C. Mazure, P. Patruno, and T.-J. K. Liu, "FinFET Performance Enhancement with Tensile Metal Gates and Strained Silicon on Insulator (sSOI) Substrate," in *Dev. Res. Conf.*, 2006, p. 39. [cited at p. 78, 79]
- [92] J. Houska, J. E. Klemberg-Sapieha, and L. Martinu, "Relationships between composition and properties of (Cr/Ti)SiN and (Cr/Ti)CN alloys: an ab initio study," J. Phys.: Condens. Matter., vol. 21, p. 285302, 2009. [cited at p. 78, 79]
- [93] K. Shimizu and T. Hiramoto, "Mobility Enhancement in Uniaxially Strained (110) Oriented Ultra-Thin Body Single- and Double-Gate MOS-FETs with SOI Thickness of Less Than 4 nm," in *Technical Digest of the International Electron Devices Meeting*, 2007, pp. 715–718. [cited at p. 79]
- [94] E. Gnani, A. Gnudi, S. Reggiani, and G. Baccarani, "Effective Mobility in Nanowire FETs Under Quasi-Ballistic Conditions," *IEEE Trans. on Electr. Dev.*, vol. 57, no. 1, pp. 336–344, 2010. [cited at p. 81]
- [95] G. Baccarani, E. Gnani, A. Gnudi, S. Reggiani, and M. Rudan, "Theoretical Foundations of the Quantum Drift-Diffusion and Density-Gradient Models," *Solid-State Electr.*, vol. 52, no. 4, pp. 526–532, 2008. [cited at p. 81]
- [96] D. Bohm, "A suggested interpretation of the quantum theory in terms of hidden variables I," *Phys. Rev.*, vol. 85, pp. 166–179, 1952. [cited at p. 82]
- [97] D. M. Caughey and R. E. Thomas, "Carrier mobilities in silicon empirically related to doping and field," *Proc. IEEE*, vol. 55, pp. 2192–2193, 1967. [cited at p. 84]
- [98] P. Palestri, C. Alexander, A. Asenov, V. Aubry-Fortuna, G. Baccarani, A. Bournel, M. Braccioli, B. Cheng, P. Dollfus, A. Esposito, D. Esseni, C. Fenouillet, C. Fiegna, G. Fiori, A. Ghetti, G. Iannaccone, A. Martinez,

B. Majkusiak, S. Monfray, V. Peikert, S. Reggiani, C. Riddet, J. Saint-Martin, E. Sangiorgi, A. Schenk, L. Selmi, L. Silvestri, P. Toniutti, and J. Walczak, "A Comparison of Advanced Transport Models for the Computation of the Drain Current in Nanoscale nMOSFETs," *Solid-State Electr.*, vol. 53, pp. 1293–1302, 2009. [cited at p. 84]

- [99] A. Thean, T. White, M. Sadaka, L. McCormick, M. Ramon, R. Mora, P. Beckage, M. Canonico, X.-D. Wang, S. Zollner, S. Murphy, V. V. D. Pas, M. Zavala, R. Noble, O. Zia, L.-G. Kang, V. Kolagunta, N. Cave, J. Cheek, M. Mendicino, B.-Y. Nguyen, M. Orlowski, S. Venkatesan, and J. Mogab, "Performance of Super-Critical Strained-Si Directly On Insulator (SC-SSOI) CMOS Based on High-Performance PD-SOI Technology," in Symposium on VLSI Technology, Dig. of Technical Papers, 2005, pp. 134– 135. [cited at p. 90]
- [100] F. Andrieu, T. Ernst, O. Faynot, Y. Bogumilowicz, J.-M. Hartmann, J. Eymery, D. Lafond, Y.-M. Levaillant, C. Dupré, R. Powers, F. Fournel, C. Fenouillet-Beranger, A. Vandooren, B. Ghyselen, C. Mazure, N. Kernevez, G. Ghibaudo, and S. Deleonibus, "Co-integrated Dual Strained Channels on Fully Depleted sSDOI CMOSFETs with HfO2/TiN Gate Stack down to 15nm Gate Length," in *IEEE Int. SOI Conf.*, 2005, pp. 223–225. [cited at p. 90]
- [101] F. Andrieu, C. Dupre, F. Rochette, O. Faynot, L. Tosti, C. Buj, E. Rouchouze, M. Casse, B. Ghyselen, I. Cayrefoureq, L. Brevard, F. Allain, J. C. Barbe, J. Cluzel, A. Vandooren, S. Denorme, T. Ernst, C. Fenouillet-Beranger, C. Jahan, D. Lafond, H. Dansas, B. Previtali, J. P. Colonna, H. Grampeix, P. Gaud, C. Mazure, and S. Deleonibus, "25nm Short and Narrow Strained FDSOI with TiN/HfO2 Gate Stack," in Symposium on VLSI Technology, Dig. of Technical Papers, 2006, pp. 134–135. [cited at p. 90]
- [102] Q. Xiang, J.-S. Goo, J. Pan, B. Yu, S. Ahmed, J. Zhang, and M.-R. Lin, "Strained Silicon NMOS with Nickel-Silicide Metal Gate," in *Symposium on VLSI Technology, Dig. of Technical Papers, 2003*, pp. 101–102. [cited at p. 90]
- [103] A. Lochtefeld and D. A. Antoniadis, "On Experimental Determination of Carrier Velocity in Deeply Scaled NMOS: How Close to the Thermal Limit?" *IEEE Electr. Dev. Lett*, vol. 22, pp. 96–97, 2001. [cited at p. 89]
- [104] Y. Dan, S. Evoy, and A. T. Johnson, *Chemical Gas Sensors Based on Nanowires*. Nanowire Research Progress, Chapter 3, Nova Science Publisher, 2008. [cited at p. 93]

- [105] Z. Fan and J. G. Lu, "Chemical Sensing With ZnO Nanowire Field-Effect Transistor," *IEEE Trans. on Nanotech.*, vol. 5, no. 4, pp. 393–396, 2006. [cited at p. 93]
- [106] D. Zhang, Z. Liu, C. Li, T. Tang, X. Liu, S. Han, B. Lei, and C. Zhou, "Detection of NO₂ down to ppb Levels Using Individual and Multiple In₂O₃ Nanowire Devices," *Nano Lett.*, vol. 4, p. 1919, 2004. [cited at p. 93]
- [107] H. Liu, J. Kameoka, D. A. Czaplewski, and H. G. Craighead, "Polymeric Nanowire Chemical Sensor," *Nano Lett.*, vol. 4, p. 671, 2004. [cited at p. 93]
- [108] Y. Wang and J. T. W. Yeow, "A Review of Carbon Nanotubes-Based Gas Sensors," *Journal of Sensors*, p. ID 493904, 2009. [cited at p. 93]
- [109] A. Agarwal, K. Buddharaju, I. Lao, N. Singh, N. Balasubramanian, and D. Kwong, "Silicon nanowire sensor array using topdown CMOS technology," Sensors and Actuators A, vol. 145146, pp. 207–213, 2008. [cited at p. 93]
- [110] J. H. Chua, R.-E. Chee, A. Agarwal, S. M. Wong, and G.-J. Zhang, "Label-Free Electrical Detection of Cardiac Biomarker with Complementary Metal-Oxide Semiconductor-Compatible Silicon Nanowire Sensor Arrays," Anal. Chem., vol. 81, p. 62666271, 2009. [cited at p. 93]
- [111] I. Park, Z. Li, A. P. Pisano, and R. S. Williams, "Top-down fabricated silicon nanowire sensors for real-time chemical detection," *Nanotechnology*, vol. 21, p. 015501, 2010. [cited at p. 93]
- [112] M.-W. Shao, H. Wang, Y. Fu, J. Hua, and D. D. D. Ma, "Surface Functionalization of HF-treated Silicon Nanowires," J. Chem. Sci., vol. 121, no. 3, pp. 323–327, 2009. [cited at p. 93]
- [113] I. Park, Z. Li, A. P. Pisano, and R. S. Williams, "Silicon nanowires for high-sensitivity glucose detection," *Appl. Phys. Lett.*, vol. 88, p. 213104, 2006. [cited at p. 93]
- [114] S. Clavaguera, A. Cella, L. Caillier, C. Celle, J. Pecaut, S. Lenfant, D. Vuillaume, and J.-P. Simonato, "Sub-ppm Detection of Nerve Agents Using Chemically Functionalized Silicon Nanoribbon Field-Effect Transistors," *Angewandte Chemie International Edition*, vol. 49, no. 24, pp. 4063–4066, 2010. [cited at p. 93, 94, 98, 106, 107, 108, 109]
- [115] X. T. Zhou, J. Q. Hu, C. P. Li, D. D. D. Ma, C. S. Lee, and S. T. Lee, "Silicon Nanowires as Chemical Sensors," *Chem. Phys. Lett.*, vol. 369, pp. 220–224, 2003. [cited at p. 94]

- [116] A. A. Talin, L. L. Hunter, F. Leonard, and B. Rokad, "Large Area Dense Silicon Nanowire Array Chemical Sensors," *Appl. Phys. Lett.*, vol. 89, p. 153102, 2006. [cited at p. 94]
- [117] M. C. McAlpine, H. Ahmed, D. Wang, and J. R. Heath, "Highly Ordered Nanowire Arrays on Plastic Substrates for Ultrasensitive Flexible Chemical Sensors," *Nature materials*, vol. 6, pp. 379–384, 2007. [cited at p. 94]
- [118] M.-W. Shao, Y.-Y. Shan, N.-B. Wong, and S.-T. Lee, "Silicon Nanowire Sensors for Bioanalytical Applications: Glucose and Hydrogen Peroxide Detection," Adv. Funct. Mat., vol. 15, pp. 1478–1482, 2005. [cited at p. 94]
- [119] G.-J. Zhang, G. Zhang, J. H. Chua, R.-E. Chee, E. H. Wong, A. Agarwal, K. D. Buddharaju, N. Singh, Z. Gao, and N. .Balasubramanian, "DNA Sensing by Silicon Nanowire: Charge Layer Distance Dependence," *Nano Lett.*, vol. 8, pp. 1066–1070, 2008. [cited at p. 94]
- [120] E. Stern, J. F. Klemic, D. A. Routemberg, P. N. Wyrembak, D. B. Turner-Evans, A. D. Hamilton, D. A. LaVan, T. N. Fahmy, and M. A. Reed, "Label-free Immunodetection with CMOS-compatible Semiconducting Nanowires," *Nature*, vol. 445, no. 7127, pp. 519–523, 2007. [cited at p. 94]
- [121] R. J. Brennan, J. F. Waeckerle, T. W. Sharp, and S. R. Lillibridge, "Chemical Warfare Agents: Emergency Medical and Emergency Public Health Issues," Ann. Emerg. Med., vol. 34, pp. 191–204, 1999. [cited at p. 94]
- [122] W. Füssel, M. Schmidt, H. Angermann, G. Mende, and H. Flietner, "Defects at the Si/SiO₂ Interface: Their Nature and Behaviour in Technological Processes and Stress," *Nuclear Instruments and Methods in Physics Research A*, vol. 377, pp. 177–183, 1996. [cited at p. 100, 101]
- [123] H. U. Kim and S. W. Rhee, "Electrical Properties of Bulk Silicon Dioxide and SiO₂/Si Interface Formed by Tetraethylorthosilicate-Ozone Chemical Vapor deposition," J. Electrochem. Society, vol. 147, pp. 1473–1476, 2000. [cited at p. 101]
- [124] D. K. Schroder and J. A. Babcock, "Negative Bias Temperature Instability: Road to Cross in Deep Submicron Silicon Semiconductor Manufacturing," *Appl. Phys. Rev.*, vol. 94, no. 1, pp. 1–18, 2003. [cited at p. 101]
- [125] D. K. Schroder, Semiconductor Material and Device Characterization, 3rd Edition. Wiley-IEEE Press, 2006. [cited at p. 101, 103]
- [126] L. M. Terman, "An Investigation of Surface States at a Silicon-Silicon Oxide Interface Emloying Metal-Oxide-Silicon Diodes," *Solid-State Electr.*, vol. 5, pp. 285–299, 1962. [cited at p. 102]

- [127] Y. Nakajima, H. Tomita, K. Aoto, N. Ito, T. Hanajiri, T. Toyabe, T. Morikawa, and T. Sugano, "Characterization of Trap States at Silicon-On-Insulator (SOI)/Buried Oxide (BOX) Interface by Back Gate Transconductance Characteristic in SOI MOSFETs," Jpn. J. Appl. Phys., vol. 42, pp. 2004–2008, 2003. [cited at p. 102]
- [128] K. Kajiwara, Y. Nakajima, T. Hanajiri, T. Toyabe, and T. Sugano, "Characterization of Distribution of Trap States in Silicon-On-Insulator Layers by Front-Gate Characteristics in n-Channel SOI MOSFETs," *IEEE Trans.* on Electr. Dev., vol. 55, no. 7, pp. 1702–1707, 2008. [cited at p. 102]
- [129] H. Angermann, W. Henriona, M. Rebiena, D. Fischer, J.-T. Zettler, and A. Röselerc, "H-Terminated Silicon: Spectroscopic Ellipsometry Measurements Correlated to the Surface Electronic Properties," *Thin Solid Films*, vol. 313–314, pp. 552–556, 1998. [cited at p. 105, 106]
- [130] N. Elfström, R. Juhasz, I. Sychugov, T. Engfeldt, A. E. Karlström, and J. Linnros, "Surface Charge Sensitivity of Silicon Nanowires: Size Dependence," *Nano Lett.*, vol. 7, pp. 2608–2612, 2007. [cited at p. 106]
- [131] R. Cohen, L. Kronik, A. Shanzer, D. Cahen, A. Liu, Y. Rosenwaks, J. K. Lorenz, and A. B. Ellis, "Molecular Control over Semiconductor Surface Electronic Properties: Dicarboxylic Acids on CdTe, CdSe, GaAs, and InP," J. Am. Chem. Soc., vol. 121, pp. 10545–10553, 1999. [cited at p. 106]
- [132] D. Cahen, R. Naaman, and Z. Vager, "The Cooperative Molecular Field Effect," Adv. Func. Mater., vol. 15, pp. 1571–1578, 2005. [cited at p. 106, 107, 108]

Curriculum Vitae

Luca Silvestri was born in Pescara, Italy, on December 15, 1981. He received his Bachelor and Master Degree (with honors) in Electronic Engineering in 2004 and 2007, respectively. Since May 2007, he has been with the ARCES research center, University of Bologna, Bologna, starting the European Ph.D. program in Information Technology in January 2008. From October 2009 to July 2010, he has been a Visiting Researcher with the Institute of Information and Communication Technologies, Electronics and Applied Mathematics (ICTEAM), Université Catholique de Louvain (UCL), Louvain-la-Neuve, Belgium. He is currently working with the ARCES device modeling and simulation group.

List of Publications

Journal Publications

- V. Passi, F. Ravaux, E. Dubois, S. Clavaguera, A. Carella, C. Celle, J.-P. Simonato, L. Silvestri, S. Reggiani, J.-P. Raskin, D. Vuillaume, "High Gain and Fast Detection of Warfare Agent using Back-Gated Silicon Nanowires MOSFETs", submitted to IEEE Electron Device Letters.
- 2) L. Silvestri, S. Reggiani, E. Gnani, A. Gnudi, G. Baccarani, "A Low-Field Mobility Model for Bulk and Ultrathin Body SOI p-MOSFETs with Different Surface and Channel Orientations", IEEE Trans. Electron Devices, Vol. 57, No. 12, pp. 3287-3294, 2010.
- 3) L. Silvestri, S. Reggiani, E. Gnani, A. Gnudi, G. Baccarani, "A Low-Field Mobility Model for Bulk, Ultrathin Body SOI and Double-Gate n-MOSFETs with Different Surface and Channel Orientations. Part I: Fundamental Principles", IEEE Trans. Electron Devices, Vol. 57, No. 7, pp. 1567-1574, 2010.
- 4) L. Silvestri, S. Reggiani, E. Gnani, A. Gnudi, G. Baccarani, "A Low-Field Mobility Model for Bulk, Ultrathin Body SOI and Double-Gate n-

MOSFETs with Different Surface and Channel Orientations. Part II: Ultrathin Silicon Films", IEEE Trans. Electron Devices, Vol. 57, No. 7, pp. 1575-1582, 2010.

5) P. Palestri, C. Alexander, A. Asenov, V. Aubry-Fortuna, G. Baccarani, A. Bournel, M. Braccioli, B. Cheng, P. Dollfus, A. Esposito, D. Esseni, C. Fenouillet, C. Fiegna, G. Fiori, A. Ghetti, G. Iannaccone, A. Martinez, B. Majkusiak, S. Monfray, V. Peikert, S. Reggiani, C. Riddet, J. Saint-Martin, E. Sangiorgi, A. Schenk, L. Selmi, L. Silvestri, P. Toniutti, J. Walczak, "A Comparison of Advanced Transport Models for the Computation of the Drain Current in Nanoscale nMOSFETs", Solid-State Electronics, Vol. 53, pp. 1293-1302, 2009.

Conference Proceedings

- L. Silvestri, S. Reggiani, A. Gnudi, E. Gnani, G. Baccarani, "Mobility Model for Electrons and Holes in FinFETs with High-κ Stacks, Metal Gate and Stress", in Proc. of the 11th International Conference on Ultimate Integration of Silicon (ULIS), pp. 73-76, March 2010, Glasgow, UK.
- 2) P. Palestri, C. Alexander, A. Asenov, V. Aubry-Fortuna, G. Baccarani, A. Bournel, M. Braccioli, B. Cheng, P. Dollfus, A. Esposito, D. Esseni, C. Fenouillet, C. Fiegna, G. Fiori, A. Ghetti, G. Iannaccone, A. Martinez, B. Majkusiak, S. Monfray, V. Peikert, S. Reggiani, C. Riddet, J. Saint-Martin, E. Sangiorgi, A. Schenk, L. Selmi, L. Silvestri, P. Toniutti, J. Walczak, "Comparison of Advanced Transport Models for Nanoscale nMOSFETs", in Proc. of the 10th International Conference on Ultimate Integration of Silicon (ULIS), pp. 125-128, March 2009, Aachen, DE.
- 3) L. Silvestri, S. Reggiani, E. Gnani, A. Gnudi, M. Rudan, G. Baccarani, "Effects of Channel Orientations, High-κ Gate Stacks and Stress on UTB-FETs: A QDD Simulation Study", in Proc. of the IEEE International Workshop on Design and Test of Nano Devices, Circuit and Systems (NDCS), pp. 7-10, October 2008, Boston MA.
- 4) L. Silvestri, S. Reggiani, E. Gnani, A. Gnudi, G. Baccarani, "Unified model for low-field electron mobility in bulk and SOI-MOSFETs with different substrate orientations and its application to quantum drift-diffusion simulation", in Proc. of the 9th International Conference on Ultimate Integration of Silicon (ULIS), pp. 129-132, March 2008, Udine, IT.
- 5) S. Reggiani, L. Silvestri, A. Cacciatori, E. Gnani, A. Gnudi, G. Baccarani, "Physically-based unified compact model for low-field carrier mobility in

MOSFETs with different gate stacks and biaxial/uniaxial stress conditions", in Proc. of the IEEE International Electron Devices Meeting (IEDM), pp. 557-560, December 2007, Washington DC.